A STUDY ON THE FEASIBILITY OF ESTABLISHING A STATE EQUIVALENT OF THE FEDERAL VISTA PROGRAM

JENSEN UCHIDA Researcher

Report No. 2, 1990

Legislative Reference Bureau State Capitol Honolulu, Hawaii 96813

TABLE OF CONTENTS

		Page
4		1
2	VOLUNTARISM AND NATIONAL SERVICE	3
	The Perspectives of Young Adults Toward Voluntarism and National Service	4
	Endnotes	6
3	STATE VOLUNTEER PROGRAMS	7
	The State's Policy Concerning the Utilization of Volunteers Statewide Volunteer Services Endnotes	7 8 9
4	THE STATUS OF FEDERAL SUPPORT TOWARD VOLUNTARISM	11
	Volunteers in Service to America (VISTA) The Status of Federal Support for National Service	12 15
	S. 1430, The National and Community Service Act of 1989 Public Law 101-204, The Domestic Volunteer Service Act Amendments of	18
	1989. Endnotes	20 21
5	EXAMPLES OF STATE PROGRAMS THAT DUPLICATE THE FUNCTION OF THE FEDERAL VISTA PROGRAM	24
	The Volunteers in Service to Iowa (VISTI) Program Ohana O Hawai'i: A Proposal To Put In The Forefront Of National	24
	Service Endnotes	25 27

Page

6	CONCLUSION	28
	Summary Discussion Recommendation	28 29 32

Exhibit

4-1 Proposals for Nationa	al or Community Service	17
---------------------------	-------------------------	----

Appendices

Α.	Senate Concurrent Resolution No. 155, S.D. 1, Fifteenth Legislature Regular Session of 1989, State of Hawaii	33
В.	Annual Assessment Results on Use of Volunteers	35
C.	Directory of Major State Government Volunteer Programs	39
D.	National and Community Service Act of 1989, 101st Congress, 1st Session of 1989	48

Chapter 1

INTRODUCTION

Senate Concurrent Resolution No. 155, S.D. 1 (Appendix A), was adopted by the Legislature during the Regular Session of 1989. S.C.R. No. 155, S.D. 1, requests the Legislative Reference Bureau to study the feasibility of establishing a pilot program for volunteer youth services similar to the California Conservation Corps or the Volunteers in Service to America (VISTA) program.

Also adopted by the Legislature during the 1989 session was House Concurrent Resolution No. 286, H.D. 1. H.C.R. No. 286, H.D. 1, requests the Department of Labor and Industrial Relations to study the feasibility of establishing a pilot program for volunteer youth services similar to the California Conservation Corps or the VISTA program.

Apart from the differences in the parties they designate to carry out the assignments, the concurrent resolutions, as adopted, are entirely identical. Apparently due to oversights in the House and Senate committee hearings held on the measures, the 1989 Legislature inadvertently requested two state agencies to undertake the same study. To avoid a situation wherein the efforts of each agency would be duplicated unnecessarily, the Department and the Bureau agreed to divide the assignment into two distinct reports. Under this agreement, the Department of Labor and Industrial Relations is examining the feasibility of conducting a pilot project modeled after the California Conservation Corps while the Bureau is determining the feasibility of conducting a pilot project patterned after the VISTA program. Accordingly, this report should be reviewed in conjunction with the report issued by the Department pursuant to H.C.R. No. 286, H.D. 1.

S.C.R. No. 155, S.D. 1, notes that the potential of Hawaii's youth to contribute to the solution of important issues is generally underestimated. The concurrent resolution points out that programs such as the California Conservation Corps and the federal VISTA and Peace Corps programs have demonstrated the ability of young men and women to focus their drive and determination in a positive manner. Involvement in these programs provides the participant with basic skills and an understanding of the value of a day's work. The concurrent resolution further notes that the VISTA program, in particular, is a program that is mutually beneficial to both the volunteer and the community.

Interpreted literally, the functional responsibilities of a state VISTA project would include the recruitment, training, placement, coordination, and payment of compensation to young adult volunteers involved in anti-poverty programs in Hawaii. The program would require a voluntary full-time commitment on the part of the volunteer for a period of one year. Volunteers would be allowed to renew their commitments for up to a maximum of five years.

Volunteers would be required to live in the areas they serve, and would be paid a minimum living allowance to cover basic expenses. A stipend would be paid to the volunteer upon the volunteer's completion of the program. State and county agencies and private non-profit programs interested in sponsoring volunteers would be permitted to request the assistance of the project in locating and training young adult volunteers. For purposes of consistency with the most common use of the term, "young adult" means individuals between 18 to 24 years of age.

Among the major topics reviewed in this report is the campaign currently underway in Congress to reaffirm the federal government's commitment toward programs promoting voluntarism and national service. Federal legislation reauthorizing the federal ACTION program has already been approved. Other bills relating to the establishment of new programs at the federal, state, and local level are currently pending before Congress.

The urgency behind the effort by Congress to address the issue of duty to the country is indicative of the belief held by some of the nation's top lawmakers that the very foundation of the American community is in the process of disintegration. The growing self-centeredness of American society and, in particular, its youth, has led many lawmakers and social scientists to believe that the only alternative available to reverse the direction of this trend is to hinge certain privileges of society upon the service an individual gives back to the community. Youth volunteer programs, most of which focus on conservation or environmental problems, have been developed in many jurisdictions throughout the country to address the problems of this age group in making the transition from youth to adulthood.

While it is too early to predict the outcome of the proposals pending before Congress, several measures--some of which are aimed directly at assisting programs identical to those that have been suggested in the concurrent resolution--if successful, will undoubtedly have an effect on the overall "feasibility" of establishing such programs in Hawaii.

This report has been divided into six chapters. Chapter 2 provides a brief analysis of voluntarism in the United States and looks at the opinions of the nation's youth toward voluntarism as it applies to their generation. Chapter 3 reviews the federal VISTA program and the federal government's present effort to revitalize VISTA and establish a national youth service program. Chapter 4 examines the State's law and program structure for the administration of volunteers and voluntary programs with special emphasis on those programs that deal with the problem of poverty in the State. Chapter 5 reviews the proposal currently under consideration in the State of Iowa to establish a "Volunteers in Service to Iowa" or VISTI program within that state and some of the proposals being developed in Hawaii that may be presented to the Legislature during the 1990 session. In conclusion, Chapter 6 presents the findings and recommendations of the report.

Chapter 2

VOLUNTARISM AND NATIONAL SERVICE

Youth accurately reflect one of the larger values of American society: making money is more important than spending one's discretionary time in service to others. (The Constitutional Rights Foundation)

The tradition of civic duty involves many streams of service--both civilian and military--and government has traditionally been a devoted supporter of these institutions and their causes. During the last decade, however, federal assistance earmarked toward volunteer organizations declined dramatically. In 1981, funding for the VISTA program was cut in half; and the Young Adult Conservation Corps was eliminated.¹

The new outlook toward programs promoting the involvement of the voluntary sector on the part of the Bush Administration has resulted in a flurry of proposals by Congress to climb aboard the bandwagon to develop some variation of a national service program. The Bush plan to establish a "Points of Light Foundation" to administer the "YES (Youth Engaged in Service) to America" program has resulted in the development of nearly 25 proposals by key members of Congress to promote new ideas on their own behalf.

Testimony presented before Congress on behalf of the President's plan to establish a national services foundation expressed an urgency to "tap and channel the energy and enthusiasm of America's young people" in response to what it perceives as the "disintegration" of society.² In defense of S. 3 (a bill to establish a citizens corps to administer a program of voluntary national service) Senator Sam Nunn testified before the Senate that:³

The citizens corps is based on a very simple but profound principle: that those who take from the common good should give something back. It rests on the premise that citizenship in a free society entails rights as well as responsibilities, and that a careful balance must be maintained between them if America's experiment in self government is to endure.

The rhetoric of the federal government's top political leaders is indicative of the widespread concern over the "growing self-centeredness of America's youth" and their understanding of the commitment they bear toward the communities they live in. In an article entitled; "The Power of Community Service", the Constitutional Rights Foundation noted that

"Young people have never been more self-centered, more concerned with money, power, and status and less concerned about helping others. Youth have never been more violent, out of control and beyond the influence of their parents and the community."⁴

The purpose of this chapter is to discuss the issue of youth and their attitudes toward voluntarism.

The Perspectives of Young Adults Toward Voluntarism and National Service

By their own admission, young Americans today are less involved in the world around them than previous generations were, according to a recent survey published by the "People for the American Way".⁵ According to the survey, when asked to rate the goals most important to them in life, a good job, family success, and having a good time ranked well, while involvement in making the community a better place to live ranked last.⁶ While the issue of assisting the community fared poorly in the context of their personal goals, many young people expressed concern for the poor and homeless in the community. But by more than a three to one margin, those questioned rejected the notion of a national law requiring all young people to spend one year in either a community service program or the military by the time they reached the age of 21.⁷ According to a time-series study performed by the Gallup Organization beginning in 1981 and ending in 1985, voluntarism among individuals 18 to 24 years of age fell from 54 percent to 43 percent.⁸

Lawmakers as well as experts in the field of social science have debated and deliberated on the reasons and causes of this phenomenon. A recent report published by the W.T. Grant Foundation suggested that, traditionally in the past, "young people were initiated into lifelong commitments to others and the community through family, church, and school"--influences that have diminished in recent years. As a result, young people feel isolated from society, a condition that helps to account for high levels of drug and alcohol abuse, teen-age pregnancy, and suicide.⁹

A study by the Carnegie Foundation for the Advancement of Teaching reached a similar conclusion. The study noted that:¹⁰

Time and again, students complained that they felt isolated, unconnected to the larger world. We were struck during our study that students can go through twelve years of formal education without becoming socially engaged, without spending time with older people who may be lonely, helping a child who has not learned to read, cleaning up the litter on the street, or even rendering some useful service to the school itself. And this detachment is occurring at the very time students are deciding who they are and where they fit.

Evidence supporting this view was presented by the American Council on Education. A survey conducted by the Council indicated that while 83 percent of a sampling of incoming college freshmen in 1967 felt that "developing a meaningful philosophy in life" was important, only 39 percent of their counterparts in 1987 indicated that they held the same philosophy.¹¹ The study further indicates that the goal of being "well off financially" rose from 29 percent in 1970 to 76 percent in 1987.¹²

Theories on the causes, studies on the cures, and surveys on the status of the youth condition in America abound in literature and journals focused on the problems of youth in contemporary society. Although extremely controversial, several researchers involved with the problems of youth as well as several key members of Congress have begun to explore the possibility of establishing mandatory requirements for service to the country as a possible solution to the problem of youth alienation. Several bills before Congress, although they have been defeated, proposed to hinge certain benefits of government on the service an individual performs for the country. Among the most frequently quoted theoreticians on the topic of national service is author Charles C. Moskos. In his book entitled: A Call to Civic Service, National Service for Country and Community, Moskos argues that:¹³

Citizenship is a matter of rights--and obligations. The advent of the all volunteer army during a period in which Americans were focusing on ensuring individual rights and entitlements while neglecting the teaching of civic responsibility resulted in a loss of a sense of civic obligation. National service, the term used to describe short-term participation in public service jobs by young citizens for little pay, offers a way to instill a sense of civic responsibility in today's youth.

The issue of mandatory service versus voluntary service sparked vigorous debate in Congress (see Chapter 4). While researchers agree that a problem exists, many researchers in the field suggest that the source of the problem is the changing values of society itself. Some suggest that it is plausible that the attitudes of youth are merely a reflection of the dominant values of the larger society today.¹⁴ The Constitutional Rights Foundation notes that "more than any other time in the last 20 years, youth hold values quite similar to adults on achievement and success."¹⁵ The Foundation continues that, "In fact, in the late 1960s, when young people did not hold those values, many adults pressured young people to abandon anti-materialistic and politically-progressive beliefs."¹⁶ The Foundation claims that blaming young people for not being more altruistic than their adult models is unfair and is equivalent to blaming the victim.¹⁷

ENDNOTES

- 1. U.S., Congress, Senate Committee on Labor and Human Resources, <u>Report Together with Additional and</u> <u>Minority Views to Accompany S. 1430</u>, 101st Cong., 1st Sess., 1989, Report 101-176, p. 33.
- 2. <u>Ibid.</u>
- 3. U.S., Congressional Record, 101st Cong. 1st Sess., January 25, 1989, vol. 135, no. 6-Part II.
- 4. Constitutional Rights Foundation, "The Power of Community Service", <u>School Youth Service Network</u>, Vol. 1, Issue 2, Spring 1989, p. 1 (hereinafter referred to as Constitutional Rights Foundation).
- 5. The Honolulu Advertiser, November 24, 1989, p. D-5
- 6. Ibid.
- 7. <u>Ibid.</u>
- 8. Constitutional Rights Foundation, p. 1.
- 9. U.S., Congress, Senate Committee on Labor and Human Resources, <u>Report Together with Additional and</u> Minority Views to Accompany S. 1430, 101st Cong., 1st Sess., 1989, Report 101-176, p. 33.
- 10. Ibid., p. 34.
- 11. Constitutional Rights Foundation, p. 1.
- 12. Ibid.
- 13. Charles C. Moskos, <u>A Call to Civic Service</u>, <u>National Service for Country and Community</u>, (The Twentieth Century Fund, 1988), p. ix (hereinafter referred to as Moskos).
- 14. Constitutional Rights Foundation, p. 1.
- 15. <u>Ibid.</u>
- 16. <u>Ibid.</u>
- 17. <u>Ibid.</u>

Chapter 3

STATE VOLUNTEER PROGRAMS

The overall theme of the Hawaii State Plan encourages the development of a "sense of social responsibility, of caring for others and for the well-being of the community and of participating in social and political life."¹ Indeed, voluntarism is strongly encouraged at all levels of government and in the private sector as well. During fiscal year 1987-1988, it is estimated that volunteers expended a total of 1,346,552 hours in assistance to state agencies alone.² Volunteers are utilized to assist the State in virtually every area of its responsibilities. Together with other existing opportunities at other levels of government and the private non-profit sector, the opportunities in Hawaii available to those who wish to volunteer are numerous.

This chapter reviews the State's law regarding the use of volunteers. In addition, the functions and the duties of the State's principal agency in charge of coordinating volunteer activities will be discussed.

The State's Policy Concerning the Utilization of Volunteers

Study and research on the State's law regarding the utilization of volunteers began in 1976 with the involvement of the volunteer coordinators of the Department of Social Services and Housing, the Judiciary, the Waimano Training School and Hospital, and the Hawaii State Hospital System.³ Act 10, Session Laws of Hawaii 1978, codified as chapter 90, Hawaii Revised Statutes, established the State's policy on the use of volunteers by state agencies. The Legislature observed in 1978, that there was a growing movement among state agencies to utilize the services of volunteers to supplement, strengthen, and support their ability to accomplish their missions.⁴ The Legislature noted that "vast numbers" of volunteers participated in programs under the Departments of Health, Social Services and Housing, Education, and the Judiciary.⁵ The Legislature found that volunteers played a vital role in assisting and augmenting the services of the State.⁶

The preamble to the Act declares that the use of volunteers by state government should be encouraged based on the following premises:⁷

(1) That every citizen regardless of his present economic condition, race, color, ancestry, political affiliation, religious affiliation, sex, age, physical or mental handicap, or marital status has the right to volunteer;

- (2) That volunteers supplement but do not compete with nor supplant paid jobs;
- (3) That volunteers provide an extra source of caring that cannot be evaluated in monetary or material terms;
- (4) That volunteering provides citizens with an opportunity to be responsive to and supportive of state government.

The State's law relating to the use of volunteers authorizes any agency of the State to recruit, train, and accept the services of volunteers.⁸ The law further authorizes state agencies to reimburse volunteers for expenses as provided in section 90-4, Hawaii Revised Statutes. Some of the benefits receivable by volunteers of the State include: meals, temporary lodging, transportation reimbursement, parking reimbursement, mileage reimbursement, training reimbursement, personal liability insurance coverage, reimbursement for reasonable expenses, and recognition awards for service.⁹ The law also authorizes state agencies to designate or establish a position to coordinate and administer the volunteer activities of the agency.¹⁰ The law further provides that agencies wishing to utilize the help of volunteers may do so to extend their services provided that volunteers shall not be used to displace paid employees. ¹¹

Section 90-3, Hawaii Revised Statutes, outlines the rights and responsibilities of volunteers involved in state programs. Volunteers of the State may expect to receive proper training, orientation, supervision, and recognition, and a job that is worthwhile and challenging. The law also requires that volunteers attempt to fulfill their commitment to the agency and follow the policies established by the agency.¹² Agencies utilizing volunteers are required to report the total number of volunteers utilized and their hours of service in their annual reports to the Governor or the Legislature.¹³

Statewide Volunteer Services

Central coordination of information relating to volunteers is performed by Statewide Volunteer Services (SVS). SVS was established in May 1976 to advocate and coordinate voluntarism in Hawaii.¹⁴ Initial funding of the program was provided through a three year grant from the ACTION program. Subsequent funding was also received from the Progressive Neighborhoods Program.¹⁵ SVS is currently a program within the Office of the Governor.

SVS serves as a central agency responsible for the coordination of volunteer programs in the public as well as the private sector. The program also serves as the liaison with national programs and agencies.

The responsibilities of SVS include serving as a clearinghouse to disseminate information, publishing a statewide newsletter, initiating research on volunteer issues, and conducting an annual assessment on the use of volunteers by state agencies.¹⁶ SVS also offers assistance on the management of volunteer programs and co-sponsors training sessions for volunteer program directors.¹⁷

SVS also coordinates the "First Lady's Outstanding Volunteer Awards Program", to honor the efforts of outstanding volunteers. Awards are presented to recipients in the following categories: youth, adult, senior, innovative project, organization and business. As of February 1989, over 2,000 awards had been presented by SVS to outstanding volunteers in the community.

Appendix B presents the results of the SVS annual assessment on the use of volunteers by state agencies. Data are provided for fiscal years 1980-1981 through 1988-1987. The data provide the number of volunteers utilized by the agency during the fiscal year, the hours of work expended by the volunteers, and the monetary contributions collected by agencies.

Appendix C, also compiled by SVS, is the State's directory of volunteer programs. The directory exhibits the wide range of opportunities available to individuals who wish to volunteer, and identifies the programs that were established to facilitate voluntarism in state government.

ENDNOTES

- 1. Hawaii Rev. Stat. sec. 226-3.
- 2. <u>Annual Assessment Results on the Use of Volunteers in the State, Fiscal Year 1987-1988.</u> Statewide Volunteer Services, Office of the Governor, State of Hawaii.
- 3. Hawaii, Statewide Volunteer Services, Office of the Governor, <u>Hawaii Revised Statutes</u>, <u>Chapter 90</u>, <u>State</u> Policy <u>Concerning</u> the Utilization of Volunteer Services, September 1984, p. 1.
- 4. 1978 Haw. Sess. Laws, Act 10.
- 5. <u>Ibid.</u>
- 6. Ibid.
- 7. <u>Ibid.</u>
- 8. Hawaii Rev. Stat., sec. 90-2(a).
- 9. Hawaii Rev. Stat., sec. 90-4.

- 10. Hawaii Rev. Stat., sec. 90-2(e).
- 11. Hawaii Rev. Stat., sec. 90-3(e).
- 12. Hawaii Rev. Stat., sec. 90-3(c).
- 13. Hawaii Rev. Stat., sec. 90-5
- 14. "Fact Sheet" (Statewide Volunteer Services, Office of the Governor, State of Hawaii, February 17, 1989).
- 15. <u>Ibid.</u>
- 16. <u>Ibid.</u>
- 17. <u>Ibid.</u>

Chapter 4

THE STATUS OF FEDERAL SUPPORT TOWARD VOLUNTARISM

Despite the outward appearance that the idea of establishing a national service program may be a new item on the federal agenda, the current wave of enthusiasm on the part of Congress could be viewed as a new campaign to support an old idea. The idea of civic duty is as old as the nation, and examples of successful federal programs in support of this notion are numerous. The Civilian Conservation Corps of the 1930s is one example of a massive federal undertaking that successfully mobilized the efforts of the citizenry to address the social and economic problems of the country during that troubled period.

Perhaps the most ambitious federal campaign to call upon the people of the country began in 1961, when President Kennedy issued his call to action: "Ask not what your country can do for you--ask what you can do for your country". Thus began the development of numerous innovative programs including the Peace Corps, the Volunteers in Service to America (VISTA) program, the National Health Service Corps, and a handful of other programs now situated under the umbrella of the federal ACTION program.

Voluntarism is once again on the federal agenda, and it seems that for the first time in nearly a decade, some form of federal support for new or existing federal, state, and local volunteer programs may be on the horizon. While it may be unwise to speculate on the type or the amount of federal support that may be in the offing for the states, an analysis of some of the measures currently under consideration by Congress may provide an insight into the direction Congress may be headed toward in the development of its final package. It is apparent that any serious action on the part of Congress to support volunteer programs at the national, state, and local levels will more than likely elevate the feasibility of establishing or expanding such programs at the state and local levels of government.

The purpose of this chapter is to review the status of federal support toward programs and proposals aimed at promoting voluntarism. In keeping with the focus of this study, special emphasis will be placed on reviewing those programs and proposals that focus on the use of volunteers to combat the problem of poverty. The VISTA program is reviewed in this chapter because it is the program identified in the concurrent resolution as a possible model for the pilot project.

A brief review will also be presented of H.R. 1312, the Domestic Volunteer Service Act Amendments of 1989. H.R. 1312 was passed by Congress on November 20, 1989, as Public Law 101-204. H.R. 1312 represents a milestone for the VISTA program in the 1980s. In

addition to appropriating nearly \$144 million to the VISTA program over the next three years, the bill allows for the expansion and continuation of several other programs under ACTION.

Also in this chapter is a review of S. 1430, a bill relating to "national service" that is reputed to be the most promising candidate for passage by Congress in 1990. This bill is significant to this report in that it proposes, among many other things, to offer assistance to states in the development of youth conservation corps programs at the state level.

Volunteers in Service to America (VISTA)

In the aftermath of the highly successful Peace Corps the establishment of a domestic equivalent seemed to follow the natural order of the social reform movement of the 1960s. President John Kennedy convened a study group headed by United States Attorney General Robert Kennedy to explore the idea of a domestic civilian service program.¹ It was not until 1964, however, that President Lyndon Johnson established the Volunteers in Service to America or VISTA program as part of the "Great Society" program.² Initially placed within the federal Office of Economic Opportunity, VISTA's objective was to combat the problem of poverty in the United States.³ The function of the program was to enlist the aid of those individuals who were concerned enough about the issues of domestic poverty, malnutrition, inadequate housing, poor education, and limited opportunities to make a personal commitment toward the elimination of these problems. The primary strategy of VISTA was to place these individuals within existing community and neighborhood organizations established to fight the problem of poverty.

According to the Domestic Volunteer Service Act of 1973,⁴ the mission of VISTA is to:

...strengthen and supplement efforts to eliminate poverty and poverty-related human, social, and environmental problems in the United States by encouraging and enabling persons from all walks of life and age groups, including elderly and retired Americans to perform meaningful and constructive volunteer service in agencies, institutions, and situations where the application of human talent and dedication may assist in the solution of poverty and povertyrelated problems and secure and exploit opportunities of selfadvancement by persons afflicted with such problems.

VISTA volunteers are assigned to local sponsoring agencies, which may be state or local public agencies or private, non-profit organizations in the 50 states, the District of Columbia, Puerto Rico or the Virgin Islands.⁵ The volunteer becomes a part of the organization that requested the individual's assistance. The volunteer's efforts, directed by the sponsoring organization, are aimed at solving problems identified in the community.

VISTA volunteers generally fall under two categories: (1) locally recruited volunteers; and (2) volunteers recruited from the "national pool". The national pool volunteer is a volunteer recruited and trained federally, and assigned to a VISTA project consistent with the individual's interests and skills. Locally recruited volunteers are trained in the community and are important members of VISTA projects in that the individual's knowledge of the community provides valuable skills. These volunteers are usually (but not necessarily) experienced in some local volunteer activities. ⁶

VISTA requires all volunteers to be 18 years of age or older. The VISTA program requires participants to voluntarily commit themselves to a year-long, full-time program.⁷ Volunteers are permitted to renew their commitments up to a maximum of five years. Volunteers are paid a monthly subsistence allowance commensurate with the cost of living in the low-income community in which they are located. The allowance is intended to cover the cost of food, housing and incidentals. Upon completion of their assignments, volunteers receive a stipend of \$75 for each month of service in the program. The VISTA commitment is a 24-hour commitment; volunteers are prohibited from taking on other jobs or attending school. Because VISTA allowances do not reduce the social security or welfare benefits volunteers may be receiving at the time, the VISTA program is particularly popular with individuals receiving such benefits.

The volunteer's role in the process is to mobilize the resources of the community to increase its capacity to solve its own problems. Local self-reliance is a crucial concept promoted by VISTA volunteers in that the ultimate objective of the program is to eventually phase out external assistance to permit communities to address their problems independently.

The early VISTA program was patterned after the Peace Corps. Like their Peace Corps counterparts, the typical VISTA volunteer came from an affluent background and was a recent graduate of a liberal arts college.⁸ In 1969, however, Congress amended the law to facilitate the recruitment of low-income individuals to work in their home communities.⁹ By the late 1970s, VISTA experienced a significant change in the social composition and age of its members. During this period, low-income volunteers accounted for two-thirds of VISTA's membership.¹⁰ In addition to this, the average age of members began to rise. The proportion of those volunteers 26 years of age or older--negligible in the 1960s--constituted more than half of the program's enrollment in the late 1970s.¹¹

Originally situated in the Office of Economic Opportunity, VISTA became a part of the new federal ACTION program in 1971.¹² Developed as the new center for federal volunteer programs, resources, and initiatives, ACTION has evolved into an agency of many programs and departments. Other programs and offices under ACTION include: the Foster Grandparent Program, the Retired Senior Volunteer Program, the Senior Companion

Program, the National Center for Service-Learning, the Peace Corps, the Urban Crime Prevention Program, the Fixed Income Consumer Counseling Program, and the Office of Voluntary Citizen Participation.

During the Administration of President Jimmy Carter, the VISTA program underwent several controversial transformations. Under the leadership of Sam Brown, Director of the ACTION program, VISTA became charged with "empowering the poor".¹³ VISTA volunteers began organizing tenant strikes and advocating welfare rights.¹⁴ In large urban areas, where concentrations of the poor are frequently found, the adage "power in numbers" became the slogan of the volunteers.¹⁵ Often VISTA volunteers successfully organized communities against the local power structure. Specific targets were chosen such as schools, housing, unemployment, and basic city services.¹⁶ Volunteers found that confrontation and public demonstrations brought the quickest responses. Challenges to unfair credit practices, rent strikes, boycotts and voter registration drives soon followed.¹⁷ The VISTA program of the Brown era found many of its objectives in an adversarial relationship with the views of local establishments. While some viewed the actions of VISTA as promoting democratic participation among the poor, others characterized these actions as promoting radicalism.¹⁸

During the first term of the Reagan administration, the membership of VISTA was cut in half. Under the leadership of James Burnley who stated that he was "working hard to be the last director of VISTA", the administration set out to do away with what it viewed as a leftist bastion.¹⁹ However, a strong grass-roots campaign by an organization called the Friends of VISTA, succeeded in keeping the program alive. Although the program that remains is said to be barely recognizable in relation to its former self, a new commitment on the part of the Bush administration may lead to the restoration of the program.

Since the inception of the program, 100,000 citizens have served as VISTA workers on community projects aimed at eliminating poverty. Without the support the program had received prior to the 1980s, however, membership in the program has dropped off considerably. The membership of the program consists almost entirely of low-income volunteers, and fewer than one member in five is below the age of 26.²⁰

In Hawaii, VISTA volunteers are recruited, trained, and coordinated through the locallybased federal ACTION program which falls under Region IX of the national program. Although the number of volunteers serving in Hawaii's VISTA program may vary from time to time, the average number of volunteers participating in the program at any given time may range from 30 to 35 volunteers. According to the ACTION program, the typical VISTA volunteer in Hawaii can be characterized as a single mother of two to three children without a college education receiving welfare assistance from the State. As noted earlier, VISTA benefits do not preclude the receipt of welfare assistance--a factor that makes the program popular with single mothers seeking to supplement the income they receive through public assistance. VISTA assignments often provide these individuals with the opportunity to learn a skill, acquire a permanent position in the organization in which they may be assigned, or return to school to receive a diploma or a college degree.

As noted earlier, living allowances paid to volunteers are based on the amount estimated to be necessary to meet the expenses of living in an area. The allotment paid to VISTA volunteers in Hawaii is thought to be among the highest in the nation. Volunteers receive an allowance of \$246.92 twice a month, and a stipend equivalent to \$75 per month of service (\$900.00 per year), upon completion of their one year terms.

The Status of Federal Support for National Service

Although programs facilitating or encouraging voluntarism on the part of the nation's citizenry have consistently garnered the support of Congress, the intensity with which these programs are currently being promoted seems quite unprecedented. The idea of service to the nation has long endured as a means by which the country may be able to meet its social, environmental, economic and military needs. Since 1973, over 60 bills proposing some kind of national service program have been reviewed by Congress. Past bills have proposed a wide range of approaches and strategies which vary significantly in size and scope, with different eligibility requirements, conditions of service, types of assignments, and post-service benefits.²¹

Sparked by the President's commitment to rekindle the government's support of programs promoting voluntarism and patriotism, Congress is currently reviewing some of the most ambitious proposals to address this issue in many years. For the first time in nearly a decade, it appears that major legislation aimed at supporting or promoting voluntarism has been enacted. Other bills before Congress relating to national service are also reputed to be on a "fast track" to passage.

On August 18, 1988, George Bush accepted the Republican nomination for the office of president. In his speech before the Republican National Convention, Vice President Bush established his agenda for the upcoming presidential campaign. In concluding his speech the Vice President promised that "I will keep America moving forward, always forward, for a better America, for an endless enduring dream and a thousand points of light."²² Although the meaning of his statement was not immediately discernible, the Vice President quickly explained the significance of his reference when queried on the matter. In short, he explained that the "thousand points of light" represented his vision of America's volunteers, and his intent in discussing the issue was to affirm his commitment toward the cause of promoting voluntarism in the United States. Voluntarism became a major theme of the Vice President's presidential campaign.

After months of internal debate and White House consultation with external groups, President Bush unveiled, on June 22, 1989, his proposal to establish a \$25 million-a-year program to spawn a new generation of volunteers. ²³ The four-year program involved the development of a government chartered foundation called "The Points of Light Initiative Foundation" to encourage all citizens--especially youth--to serve their country and their communities for free. The foundation, with a paid staff of about 50 employees, would be run by a volunteer board of directors with President Bush serving as honorary chairperson. The purpose of the program would be to promote voluntarism throughout the country by encouraging corporations and other organizations to donate their employees' time to offer service or provide technical assistance to programs dealing with the disadvantaged. The program would set up a national hot line for information on volunteer organizations and would issue awards to recognize the work of outstanding volunteers.

Inasmuch as the issue of voluntarism was a major theme of the president's campaign, Congress viewed the Bush plan as being rather modest. Critics of the plan saw little direction and purpose in the administration's agenda, and felt that the proposal amounted to nothing more than a program to present recognition awards. Democrats within Congress dismissed the Bush plan as mere rhetoric.²⁴ With the position of the President on record, however, Congress launched its own initiative by introducing a flurry of legislation seeking to support various existing social programs or to propose new programs aimed at encouraging people to serve. Exhibit 4-1 is a summary of some of the national service bills currently pending before Congress. It is estimated that as many as 22 proposals dealing with national service have been placed on the congressional agenda for consideration. Action on these measures, must resume with the convening of Congress in early 1990.

Amid the flurry of bills introduced in 1989 were several measures that drew particular attention to the concept of national service. One proposal in particular, H.R. 660 and S. 3, companion measures offered by Representative Dave McCurdy and Senator Sam Nunn, touched off considerable debate.²⁵ The focus of the debate centered on an aspect of the Nunn-McCurdy proposal that caused many lawmakers to become rather leery of supporting the general approach of the bills. The Nunn-McCurdy plan sought to instill in the nation's youth, a sense of duty to serve their country in a manner they readily conceded was a radical approach: making military or civilian "national service" a prerequisite for receiving federal student aid.²⁶

Under the Nunn-McCurdy bill, volunteers would serve one to two years in a citizen corps doing community service for \$100 a week, or two years' active duty in the military for two-thirds of the regular pay, plus six years in the reserves. In return, civilian volunteers would get \$10,000 for each year served and two-year military volunteers would receive \$24,000. The money, in the form of vouchers, could be used toward training, payment for education, or a down payment on a home.²⁷

EXHIBIT 4-1

Proposals for National or Community Service

	Kennedy (S 650)	Mikulski/ Bonior (S 408, HR 1000)	Nunn/McCurdy (S 3, HR 660)	Dodd/Panetta (S 322, HR 717)	Pell (S 576)	Moynihan (\$ 232)	Graham (S 382)	Bumpers (S 539, S 54 S 541)
SUMMARY	Student com- munity service	National Guard model	Education benefits tied to service	American Con- servation Corps and Youth Service Corps	National serv- ice demonstra- tion program	American Con- servation Corps	Business & cit- zen volunteers in schools	Student-loan forgiveness/ cancellation service
FIRST-YEAR COST & MATCH	\$100 million: 80% 1st year, 70% 2nd year, 60% 3rd year, 50% 4th & after	\$250 million; state pays ad- ministrative costs	Not available; 25% match on some costs	\$152.4 million; 50%	\$30 million plus \$5 million for Peace Corps demo.; no match	\$75 million: 50%	\$5 million: no match	Not available
TARGET POPULATION/ SELECTIVITY	Elementary, secondary, and college stu- dents; out-of- school youth	Selection based on fit- ness for duty: availability of position; date of application	Age 17 or over; high school graduates; se- nior citizens; admission by application	Age 16-25 (fuli- year); 15-21 (summer); high school gradu- ates or drop- outs; special efforts to enroll disadvantaged	Age 16-25 high school gradu- ates (2 years at higher ed inst. for Peace Corps): admis- sion for Peace Corps demo: competitive; ef- fort to enroli underserved groups	Unemployed, Age 16-25 (full- year), 15-21 (summer); spe- cial effons to enroll the dis- advantaged; in- dividuals with special skills	higher educa- tion students; older Ameri- cans receive	College grad ates with loa
TYPE OF SERVICE	Community service such as tutoring, assist- ing the elderly, anti-drug- abuse efforts, work in day- care centers, homeless shel- ters, food banks	Placed by state in: non-profit agencies, youth pro- grams, pro- grams, pro- grams, pro- derly, public safety work (volunteer fire- fighters, emer- gency medical services), con- servation, pov- erty programs	Educational, human, and conservation service, public safety work, or existing pro- grams (Peace Corps, VISTA) (civ.); military service	ACC: improve- ment of public lands; Youth Corps: place- ment with gov. agencies, hos- pitals, schools, nursing homes, law-enforce- ment agencies, and non-profit service orgs.	Determined by state higher ed agency: health care, child care, iteracy, welfare, social services, trans- portation, housing, public safety, recre- ation, rurai development, etc.; Peace Corps	Improvement of public lands	Volunteers in schools; prior- ity for transfer of high-tech skills, anti-drug and alcohol abuse, drop- out prevention, nutrition, health education	Peace Corps VISTA, or set ice with any tax-exempt, non-profit organization
Ength of Bervice	Sustained commitment part-time	3-6 years part- time, 2 week- ends + 2 weeks, or 9 hours week av- erage	1-2 years full- time (civ.) or 2 years active duty + 6 years reserve or 8 years reserve (mil.); unspeci- fied for senior citizens	2 years tuli time	2 years tuli time	2 years fuil time	Unspecified, part time	1 year full tim
	Optional; aca- demic credit	\$3,000/year for education or down payment on home	\$10,000/year (civ.); \$12,000/year (mil.); for edu- cation or down payment on home	In-service job training! education benefits up to 110% of mini- mum wage	\$600/month (Educ. costs for Peace Corps demo.)	95% of pay rate for grade E-1 members of armed forces minus room and board	Unspecified	Cancellation student-loan debts
		6 weeks re- quired	Not specified	Training re- quired to en- hance skills	Training only in Peace Corps demo.	Required to en- hance skills	Optional for managers	N.A.

Source: Congressional Quarterly Weekly Report, Vol. 47, No. 12, p. 64.

Critics of the bill claimed that the measure challenged the very premise of the nation's student-aid system, in which government endeavors to give all high school graduates--regardless of income--an equal opportunity to attend college.²⁸ A wide array of grants and loans are available to students in this regard, with no strings attached.

Education leaders in Congress denounced the bill as being blatantly discriminatory.²⁹ In a column printed in the <u>Chronicle of Higher Education</u>, Representative William Ford stated that the Nunn-McCurdy plan "holds educational aspirations of the poor hostage to public service, while excusing the affluent".³⁰ The chairman of the Senate Labor Committee's education subcommittee called the idea of making poor students work for educational assistance "cruel".³¹ The chairman of the House Education Committee said "I think its dead".³²

Although S. 3 promoted vigorous debate in the Senate, strong consensus existed among the democratic members of Congress to advance some sort of legislation aimed at establishing a national service program. Opponents of the Nunn-McCurdy proposal were quick to offer their own proposals to encourage, but not require, voluntary national or community service, and Nunn and McCurdy were apparently amenable to suggestions and compromise.³³ On July 27, 1989, Senate Majority Leader George Mitchell and Senators Nunn and Kennedy announced the development of S. 1430. A massive, multi-purpose, omnibus bill, S. 1430 was the product of the merger of more than six bills.

S. 1430, The National and Community Service Act of 1989

S. 1430 (Appendix D), offered by Senators Kennedy, Mitchell, Nunn, Pell, Mikulski, Dodd, Robb, Graham, Simon, Matsunaga, and Bumpers is a lengthy multi-purpose measure developed to enhance, establish, and provide assistance to various national and community service programs. In its preamble, S. 1430 declares that "service to the community and the nation is a responsibility of all Americans, regardless of the economic level or age of such citizens", and that involvement in such service at an early age will enable the country's youth to "better understand the responsibilities of citizenship and continue to serve the community into adulthood". The bill finds that conservation corps and human service corps such as VISTA provide "important benefits to participants and to the community" in a "cost effective" manner. The bill further finds that the youth of the United States offer a "powerful and largely untapped resource for community service". The bill recognizes the impediments standing in the way of young adults (cost of higher education, loan indebtedness, cost of housing, etc.) with regard to volunteering their time toward community service programs. The preamble concludes with the finding that "a range of full-time and part-time national and community service opportunities should be made available to all citizens, particularly youth and older Americans".

The stated objective of S. 1430 is to renew the ethic of civic responsibility in the people of the country by calling on all citizens to engage in full-time or part-time service to the nation. To accomplish this, the bill attempts to remove some of the barriers that have been created by the costs of education, loan indebtedness, and housing. The goal of the bill is to "generate 1,000,000,000 additional volunteer service hours" to help meet human, educational, environmental, and public safety needs, particularly those needs relating to poverty.

S. 1430 is subdivided into six titles as follows:

Title I	School Based Community Service
Title II	Youth Service Corps
Title III	National Service Demonstration Program
Title IV	Corporation for National Service
Title V	Expansion of VISTA
Title VI	National Older Americans Volunteer Programs

As originally drafted, S. 1430 proposed to authorize \$100 million in federal grants to school-based volunteer programs. Citizens would be encouraged to work for free in elementary and secondary schools; students would be encouraged to do likewise in the community. Colleges would be urged to use work-study money for community service projects. Federally-backed and subsidized student loans would be forgiven for students who perform a certain amount of full-time community service.³⁴

The bill also appropriates \$100 million for the establishment of a new youth service corps. States would be allowed to request grants from a new national service board to offer youths aged 16 to 25 poverty-wage stipends (less than \$7,000 per year) and up to \$100 a week (\$5,200 a year) in post-service student aid for full-time community service or "conservation corps" type work.³⁵

The bill further authorizes \$100 million for a demonstration project that would give states grants to establish full-time and part-time civilian service programs for anyone 17 years of age or older. Full-time workers would receive poverty-wage stipends. After serving for a period of two years, full-time volunteers would receive \$8,500 for each year of their service to finance their education or a first purchase of a home.³⁶

The bill also calls for the expansion of the ACTION program which includes the VISTA program. Funding for the VISTA program is increased to restore the program to its peak level of the past. The Older American Volunteer Programs are expanded to involve senior citizens in the struggle against AIDS, teen-age pregnancy, drug and alcohol abuse, and other national problems. The bill increases the incentive for higher education institutions to use work study funds for community service placement, and allows VISTA, the Peace Corps, and other full-time volunteers to reduce their student loan indebtedness.³⁷

Although the Bush Administration disagrees with the concept of paying stipends and other benefits to volunteers, the proponents of the bill point out that this concept is not new.³⁸ VISTA and the Peace Corps are two programs that offer stipends to volunteers to assist them in meeting their minimum economic needs. Although S. 1430 is completely unlike the measure originally introduced by Senator Nunn, the senator believes that the demonstration project proposed in the bill will serve as a pilot project to test several concepts he proposed in S. 3.³⁹

S. 1430 was referred to the Senate Committee on Labor and Human Resources. The measure was reported from that committee with amendments on October 27, 1989. No further action was taken on the bill during the remaining period of the session.

Public Law 101-204, The Domestic Volunteer Service Act Amendments of 1989

On November 20, 1989, Congress approved legislation (H.R. 1312) for the reauthorization of the federal government's principal domestic volunteer programs.⁴⁰ Final action came when the House of Representatives accepted by voice vote a conference committee report the Senate had approved the day before.

While Congress experienced moderate difficulty and debate in seeking to pass bills to establish new "national service" programs, agreement on the reauthorization of existing programs presented no problems. P.L. 101-204 affects programs under the auspices of the ACTION agency under the Domestic Volunteer Services Act. Among these programs are VISTA and its Literacy Corps, and the Older American Volunteer Program which includes the Foster Grandparent Program and the Retired Senior Volunteer Program.⁴¹ Unlike S.1430, which encourages states to develop conservation corps programs, the VISTA reauthorization law does not seem to be oriented toward providing grants or assistance to states to establish state VISTA programs. The focus of the law is on expanding existing federal programs.

In addition to renewing the programs under ACTION, a major objective of the law is to stimulate recruitment of more volunteers, especially for VISTA. The law directs ACTION to recruit at least 2,800 VISTA volunteers in fiscal year 1990, rising to 3,400 in 1993. The law calls for a renewed campaign to promote voluntarism through radio, television and print advertising. The law specifies that at least 1.5 percent of VISTA's appropriation must be spent on recruitment and public relations activities.⁴²

To increase its appeal to young adults and adults with college educations, the law requires VISTA to publicize the availability of student loan deferrals or forgiveness of student loans. The law also increases the monthly stipend payable to the volunteer upon the volunteer's completion of the program from \$75 to \$90. The law also specifies that the subsistence allowance should be raised to 105 percent of the poverty threshold.⁴³

H.R. 1312 contains the following appropriations for fiscal years 1990, 1991, 1992, and 1993:

VISTA:	\$174.3 million
VISTA Literacy Corps:	\$33 million
Service-Learning Program:	\$8.2 million
Special Volunteer Programs:	\$4.7 million
Drug-Abuse Prevention (1992, 1993):	\$5.8 million
RSVP:	\$185.2 million
Foster Grandparents:	\$341.6 million
Senior Companion Program:	\$169 million.

ENDNOTES

- 1. Charles C. Moskos, <u>A Call to Civic Service</u>, <u>National Service for Country and Community</u> (The Twentieth Century Fund, 1988), p. 53 (hereinafter referred to as Moskos).
- 2. <u>Ibid.</u>
- 3. John G. Cull. Volunteerism: An Emerging Profession (American lecture series in social and rehabilitation psychology, Publication No. 887, Springfield. III., 1974), p. 23 (hereinafter referred to as Cull).
- 4. 42 U.S.C. 4951
- 5. U.S., VISTA, Volunteers in Service to America, <u>Serve in the USA, Be a VISTA Volunteer</u> (Informational pamphlet VISTA #304) (hereinafter referred to as VISTA pamphlet).
- 6. Cull, p. 24.
- 7, VISTA pamphlet.
- 8. Moskos, p. 54.
- 9. <u>Ibid.</u>
- 10. <u>Ibid.</u>
- 11. <u>(bid.</u>
- 12. Cull, p. 25.
- 13. Moskos, p. 54.
- 14. Ibid.

- 15. Kenneth Oldfield, "The Early Years of VISTA: The Political Alteration of a Successful Policy", <u>The Journal</u> of Volunteer Administration, Vol. VII, No. 1, p. 15.
- 16. Ibid.
- 17. <u>Ibid.</u>
- 18. Moskos, p. 54.
- 19. <u>Ibid.</u>
- 20. <u>Ibid.</u>
- 21. U.S., Senate Committee on Labor and Human Resources, "Report Together with Additional and Minority Views to Accompany S. 1430", 101st Congress, 1st Sess., Calendar No. 331, Report 101-176, October 27, 1989, p. 39.
- 22. "Bush Takes the Lead and Defines His Mission", <u>Congressional Quarterly Weekly Review</u>, August 20, 1988, Vol. 46, No. 34, p. 2356.
- 23. "Bush Outlines His Proposal For Volunteer Service, Congressional committees plan to turn up wattage of president's 'thousand points of light'", Congressional Quarterly Weekly Report, June 24, 1989, p. 1555.
- 24. "Senate Democrats Unveil Bill to Promote Voluntarism", <u>Congressional Quarterly Weekly Reporter</u>, July 29, 1989, Vol. 47, No. 3, p. 1970.
- 25. "Nunn-McCurdy Plan Ignites National Service Debate", <u>Congressional Quarterly Weekly Report</u>, March 25, 1989, Vol. 47, No. 12, p. 645.
- 26. Ibid.
- 27. <u>Ibid.</u>, p. 647.
- 28. <u>Ibid.</u>
- 29. <u>Ibid.</u>
- 30. Ibid.
- 31. <u>Ibid.</u>
- 32. Ibid.
- 33. Ibid.
- 34. "Senate Democrats Unveil Bill to Promote Voluntarism", <u>Congressional Quarterly Weekly Report</u>, July 29. 1989, Vol. 47, No. 30, p. 1970.
- 35. Ibid.
- 36. Ibid.

- 37. Ibid.
- 38. <u>Ibid.</u>
- 39. <u>Ibid.</u>
- 40. "Renewal of Volunteer Programs Cleared for White House", <u>Congressional Quarterly Weekly Report</u>, November 25, 1989, Vol. 47, No. 47, p. 3243.
- 41. <u>Ibid.</u>
- 42. <u>Ibid.</u>
- 43. <u>Ibid.</u>

.

Chapter 5

EXAMPLES OF STATE PROGRAMS THAT DUPLICATE THE FUNCTION OF THE FEDERAL VISTA PROGRAM

State anti-poverty volunteer programs designed to duplicate the functions already performed by the federal VISTA program are uncommon. This is not to say that state programs that deal with the problem of poverty do not utilize volunteers. Indeed, the use of volunteers to assist state and local programs in the fight against poverty undoubtedly occurs in every state. Instead, it is the belief that a need exists for a state program to perform the same function as the federal program that is somewhat unique.

Apparently in most states, including Hawaii, volunteers involved in programs focusing on poverty are trained and supervised in the program most consistent with the volunteer's interests. Although a comprehensive survey of anti-poverty programs that utilize volunteers at the state level throughout the United States was not available, this seems to be the most common system of managing volunteers. While it is difficult to speculate on the reasons most state anti-poverty volunteer programs opt not to organize their programs under state VISTA agencies, the most obvious reason may be that this action would duplicate the function that is already performed by another program.

The purpose of this chapter is to examine other programs that duplicate the function of the VISTA program at the state level. Although funding for the "Volunteers in Service to Iowa" (VISTI) project failed to be authorized during 1989, the VISTI proposal seems entirely consistent with the focus of the pilot project being reviewed in this report.

This chapter will also present a review of a prospectus entitled: "A Proposal To Put Hawaii in the Forefront of National Service". Based on S. 3, the Nunn-McCurdy bill (see Chapter 4), the proposal would establish an independent, state-funded, non-profit corporation called "Ohana O Hawai'i". Although the proposal does not deal with the establishment of a state VISTA program, the proposal is discussed in this chapter because of the likelihood that it will be submitted to the Legislature during the 1990 session in anticipation of the authorization of federal assistance by Congress.

The Volunteers in Service to Iowa (VISTI) Program

In anticipation of possible assistance from the federal government, Iowa Governor Terry E. Branstad, established a committee to explore the barriers that existed toward voluntarism in that state.¹ The committee was composed of individuals from a variety of state agencies including the Department of Education, Human Services, Aging, and

STATE PROGRAMS THAT DUPLICATE THE VISTA PROGRAM

Community Education. Also included in the committee was a single member from the federal ACTION office in Iowa. The committee held a series of six public hearings across the state to ascertain the public's view toward the issue of voluntarism.²

Among the recommendations developed by the committee was a proposal to establish a "Volunteers in Service to Iowa" (VISTI) program. The purpose of the program would be to recruit and train Iowans to serve as volunteers within the State of Iowa in programs designed to meet the needs of Iower income and disadvantaged citizens of that state.³ The project would be placed under the Office of the Statewide Volunteer Coordinator, and the director of the office would serve as the project's leader. The program would allow any state, county, local governmental unit, or private non-profit organization to submit an application for volunteers. Full-time enrollment in the program would be open to all citizens of the state 18 years of age or older. Re-enrollments would be allowed up to five years. The proposal also set up an "associate volunteer" program for participants 13 to 17 years of age.⁴

The VISTI program would issue a subsistence allowance to all full-time volunteers, and health insurance would be provided at no cost to the volunteers. A stipend of \$75 would be paid to the volunteer for every month the volunteer served in the program. The stipend would be issued upon the volunteer's completion of the program.⁵ The program would provide training to all volunteers through orientation sessions scheduled at least four times a year. On-the-job training would be the responsibility of the sponsoring agency. The cost to the State of Iowa per VISTI volunteer per year was estimated at \$7,000.⁶ This estimate includes the living allowance, the stipend, any food, transportation, and lodging reimbursement, and insurance.⁷

Although the committee remains optimistic that the project will be established sometime in the future, funds for the program were not authorized in 1989.⁸ According to the lowa ACTION program director, due to the proximity of the next gubernatorial election, legislators as well as the governor have exercised particular restraint in authorizing new programs.⁹ Instead, the state has opted to provide assistance toward the expansion of existing programs such as the federal RSVP program.¹⁰

Ohana O Hawai'i: A Proposal To Put Hawaii In The Forefront Of National Service

Developed by the "Coalition for Service to Hawaii" (Coalition) the draft proposal is based on the philosophy of the Nunn-McCurdy bill (S. 3) currently before Congress. The stated purpose of the Coalition is to "rekindle the spirit of aloha in the community by reviving a sense of mutual obligation and sharing among all."¹¹ To accomplish this, the coalition will press for the creation of a state-chartered corporation to:¹²

- (1) Mobilize both public and private resources so as to provide the fullest possible opportunities for civic service to the citizens of Hawaii; and
- (2) Coordinate the programs so created with federally sponsored programs of national service.

The proposal notes that several bills before Congress have come to the conclusion that a national service program has become a necessity. The proposal notes that "to take maximum advantage of this emerging federal program, the Legislature should act now."¹³

The proposal states the key concepts of the civic service plan are "aloha", "ohana", and "ho'oponopono".¹⁴ The proposal notes that it is important that program administration "be as decentralized and localized as possible" and that "all programs must possess 'civic content'".¹⁵

With respect to decentralization, the proposal notes that Ohana O Hawai'i will not administer most programs.¹⁶ Instead, the program would coordinate and facilitate volunteer recruitment, training, and administration for the various government and private "sponsoring agencies". In situations where sponsoring agencies did not fill a functional area, however, the proposal states that Ohana O Hawai'i could "directly administer" a program. The proposal states that the program would be run largely by volunteers, maintaining only a small permanent staff for administration and other functions.¹⁷

The proposal estimates the cost per "sponsored program" participant per year to be \$7,200 (stipend, medical insurance, and other expenses).¹⁸ The cost per participant for "directly administered" residential programs would be \$16,000. The cost per participant for directly administered non-residential programs would be \$9,000.¹⁹ The proposal states that the revenues of the program would be derived through new appropriations, fees charged to sponsoring agencies (i.e., \$500 to \$1000 per volunteer per year), private or federal grants, transfer of funds from existing state programs.²⁰

The proposal also outlines several initial projects for the program, these projects include: (1) the expansion of the YMCA and the YWCA; (2) the recruitment of high school graduates to serve as aides to police officers and social workers to allow these professionals to concentrate on their primary duties; (3) the establishment of a conservation corps; (4) the establishment of a requirement that students perform 100 hours of civic service as a requirement of high school graduation; and (5) the establishment of a civic service requirement as a condition of graduation from the University of Hawaii.

STATE PROGRAMS THAT DUPLICATE THE VISTA PROGRAM

ENDNOTES

- Telephone interview with Joel Weinstein, State Action Program Director, State of Iowa, December 14.
 1989 (hereinafter referred to as "Weinstein interview").
- 2. Ibid.
- 3. "Volunteers in Service to Iowa", by Joel Weinstein, ACTION Program Director, Des Moines, Iowa.
- 4. ibid.
- 5. <u>Ibid.</u>
- 6. Weinstein interview.
- 7. <u>Ibid.</u>
- 8. Ibid.
- 9. Ibid.
- 10. <u>Ibid.</u>
- 11. "Coalition for Service to Hawai'i, <u>A Proposal To Put Hawaii In the Forefront of National Service.</u> 1989.
- 12. <u>Ibid.</u>
- 13. <u>Ibid.</u>
- 14. Ibid.
- 15. <u>Ibid.</u>
- 16. <u>Ibid.</u>
- 17. <u>Ibid.</u>
- 18. <u>Ibid.</u>
- 19. Ibid.
- 20. <u>Ibid.</u>

Chapter 6

CONCLUSION

Summary

Senate Concurrent Resolution No. 155, S.D. 1, requests the Legislative Reference Bureau to study the feasibility of establishing a pilot program for volunteer youth services similar to the California Conservation Corps or the Volunteers in Service to America (VISTA) program. House Concurrent Resolution No. 286, H.D. 1, requests the Department of Labor and Industrial Relations to perform the same assignment. In accordance with an agreement established between the Department and the Bureau, the Bureau focused its report on the feasibility of conducting a pilot project patterned after the VISTA program, and the Department focused its study on the feasibility of conducting a pilot project patterned after the California Conservation Corps program. Accordingly, this report should be read in conjunction with the report issued by the Department of Labor and Industrial Relations in response to H.C.R. No. 268, H.D. 1.

The issue of civic duty to the country is once again a topic of discussion on the federal agenda. In response to what some of the nation's top lawmakers equate as the "disintegration" of the fabric of American society, Congress has launched the most ambitious effort to establish a national service program in many years. Federal reauthorization of existing federal programs has already occurred. Funds for the expansion of the VISTA program have been approved by Congress.

For the first time in nearly a decade, the establishment of a national service program for youth may be a distinct possibility. While some lawmakers have labeled the concept of mandatory national service as being rather draconian, the idea of making contingent certain social benefits upon the performance of some kind of service to the community is currently being promoted by several key lawmakers in Congress.

Although the youth of the country readily admit that they are less interested and involved in social affairs than previous generations, the idea of mandatory service to the country was rejected by a three to one margin in a survey recently conducted.

Hawaii's law regarding the use of volunteers at the state level appears as chapter 90, Hawaii Revised Statutes. The law was established to promote the use of volunteers by state agencies, define the rights and responsibilities of volunteers, define the responsibilities of agencies that utilize volunteers, and authorize the payment of reasonable benefits or compensation to volunteers. Although the use of volunteers by state agencies is monitored

CONCLUSION

by the Office of Statewide Volunteer Services, state agencies have traditionally maintained a high level of independence over the administration of their programs. Volunteers are utilized to perform a wide array of functions in virtually every field within the State's responsibility. The opportunities available to those who are interested in volunteering are virtually unlimited. Presently absent, however, is any formal program to utilize the services of volunteers to assist the State in the maintenance of state conservation lands.

While there are numerous examples of youth programs that function somewhat similarly to the California Conservation Corps, state programs that parallel the functions already performed by the federal VISTA program are rare. Although it is difficult to speculate on the reasons most states have opted to establish state youth conservation corps programs instead of programs that parallel the federal VISTA program, several factors may account for their actions. The following section discusses some of the possible factors.

Discussion

By suggesting that the proposed pilot program either focus on conservation or antipoverty issues, the agencies performing the studies have been placed in the difficult position of suggesting that a trade-off be made. It is extremely difficult to suggest which program would be more beneficial to the State. The VISTA program and the California Conservation Corps are programs with widely divergent missions and objectives. Although they both involve the use of volunteers, these programs benefit the community in different areas and in different ways. The mission of the VISTA program is to eradicate poverty in communities throughout the United States while the California Conservation Corps directs its focus at environmental problems in California. Without question, the benefits and merits of one program cannot be contrasted against the benefits of the other. Eradicating poverty and maintaining the quality of the environment are two important policy objectives of the State. It is difficult for the agencies to suggest to the Legislature that one may be more important than the other.

Additionally, the rewards and benefits received by participants and the community as a result of these programs are often intangible and defy measurement through conventional cost/benefit or feasibility analysis. The good that may be brought about to the State as the result of the establishment of either program cannot be translated into in quantitative or economic terms. The benefits of such programs cannot be contrasted with the cost of their implementation.

Faced with the task of developing a recommendation on the possible focus of the proposed pilot program, however, several facts regarding the alternatives suggested can be examined. In its review of the possibility of establishing a state VISTA project, the Legislative Reference Bureau found that:

- (1) The federal VISTA program that operates under the ACTION program in Hawaii already functions in this capacity;
- (2) P.L. 101-204 reauthorizes the federal VISTA program. The law calls for the expansion of the federal program but does not encourage or suggest that states develop state-level VISTA programs;
- (3) Although examples of state programs that duplicate the function of VISTA are difficult to locate, examples of successful youth corps programs that function similar to the California Conservation Corps are numerous (see the Department of Labor and Industrial Relations' report on H.C.R. No. 286); and
- (4) Federal assistance to states for the establishment of state youth conservation corps programs is currently part of a congressional bill to promote voluntarism.

Although most individuals and programs contacted during the course of this study viewed the proposal to establish a state volunteer program to assist the less fortunate as a noble and worthy undertaking, one question that frequently surfaced during meetings on this issue was "Why is such a program being proposed?". Indeed, the notion of promoting voluntarism to assist the less fortunate is a wholesome and noncontroversial idea. However, a program to facilitate this presently exists at the federal level, and personnel involved with programs that utilize volunteers failed to sense any need to duplicate this function at the state level. Traditionally, state programs that utilize volunteers have relied on their own staff to train and supervise their volunteers independently. State law permits the hiring of volunteer coordinators within state agencies to perform this function. In terms of recruitment, the names of interested individuals can readily be obtained through existing referral services such as the Office on Statewide Volunteer Services, VISTA, or the Volunteer Information and Referral Service.

One issue that was frequently brought forward during meetings with individuals familiar with the problem of recruiting volunteers was the situation that has evolved as a result of the ongoing labor shortage in the State. Although low unemployment obviously benefits the economy of the State, the highly competitive market for labor in Hawaii has undoubtedly had a negative effect on the likelihood of certain groups of people to consider the possibility of volunteering. For example, the high salaries currently being offered to young, unskilled, entry-level workers obviously works as a disincentive to these individuals to consider working for free. Although a formal analysis or survey was not performed with regard to this issue, it may be prudent to expect low rates of voluntarism from individuals of this age group to be a factor. To ensure the success of a youth volunteer program, it is apparent that a serious commitment on the part of the Legislature to provide the program with the resources and support to recruit and maintain its membership will be necessary.

CONCLUSION

Another issue that became a major topic of discussion in several meetings with agency personnel was the problem of "high-risk" youth and youth gangs in Hawaii. While the priorities of the Legislature were never questioned or criticized, it was frequently noted that programs of this nature rarely attract that category of youth that seems to be in need of the most assistance at this time. Indeed, programs offering experiential rewards that promise to pay-off during entrance interviews with prospective colleges or employers will more often than not attract those well motivated individuals that need little assistance in finding their direction in life. While it is clear that the objective of the Legislature in considering the possibility of establishing a state VISTA program was not to solve the problem of youth gangs in Hawaii, the Legislature should be cognizant of the concern expressed by state personnel involved with individuals in this age category.

Clearly, one of the basic objectives of the Legislature is to promote voluntarism by offering a program that would appeal to the interests and tastes of young adults. Implicitly, therefore, one measure of such a project's success would be the number of young adults it succeeds in attracting to the program. The greater the project's appeal to the target age group, the higher the project's enrollment--the higher the project's enrollment, the greater its outreach to the environment or the community. Obviously, a program designed to appeal to young adults will more than likely attract a greater number of individuals from this age category than a program that is not.

VISTA programs are not widely known for their appeal to the young adult age group while youth conservation corps programs have proven to be quite popular with individuals in this age category. A basic tenet of most youth conservation corps programs is that the youth of America require some form of guidance during the period of their passage from youth to adulthood. Conservation corps programs are specifically designed to guide individuals of this age group through this period of transition by subjecting them to what some have described as being the experiential equivalent to military "boot camp". According to supporters of these programs, this "rite of passage" into adulthood instills character, teaches responsibility, and assists participants in finding their goals in life.

With the prospect of federal assistance for the development of youth conservation corps on the horizon, the economic considerations may cease to be a limiting factor in the development of the program. Indeed, federal assistance, if and when it arrives, will undoubtedly broaden the options available to the State in developing such a program. The State would be remiss in passing up the opportunity to capitalize on such assistance.

Although several bills before Congress are said to be on a "fast track", it may be unwise to attempt to anticipate the passage or the contents of the final federal package. It is fairly evident, however, that major changes to Hawaii's law to ensure that the State will be eligible to receive grants or assistance under the proposed assistance package is not

31

immediately necessary. The consideration of elaborate proposals to establish central umbrella agencies or other corporations in anticipation of legislation still pending before Congress may be premature. For example, grants to local school districts would probably be arranged through the Department of Education; and, in situations where responsible agencies or jurisdictional lines may be unclear, designation by the Governor will probably suffice. Ostensibly, the agency most appropriate to act as the central clearinghouse of the federal assistance and the coordinator of the pilot project would be the Statewide Volunteer Services. In summary, it seems that the infrastructure to receive federal assistance under the proposed federal package already exists. Indeed, it seems that Congress is structuring its package to facilitate the use of its funds by existing governmental organizations at the federal, state, and local level. It does not appear that Congress is planning to impose unreasonable requirements on the states in order to qualify for grants or assistance under the proposed program.

Recommendation

The Bureau finds that the most logical focus of a program to promote voluntarism among young adults would be one that is consistent with the interests of individuals in this age group. Youth conservation corps programs such as the California Conservation Corps were specifically designed to focus on the interests and needs of young adults. Youth conservation corps programs have experienced widespread success throughout the United States--a fact that Congress apparently recognizes and is presently seeking to promote. Assistance from the federal government to states for the development of youth conservation corps programs is a distinct possibility in the near future. Accordingly, if a youth volunteer program is to be established by the State, the development of a youth volunteer pilot project based on the California Conservation Corps model would appear to be a more prudent alternative than a VISTA type program.

As noted earlier, the focus of this report has been directed toward the examination of establishing a state VISTA pilot project. By agreement, the Department of Labor and Industrial Relations has focused its study on the feasibility of conducting a pilot project patterned after the California Conservation Corps. In this regard, the Bureau defers to the recommendations developed by the Department of Labor and Industrial Relations in its report. Specific recommendations concerning the structure, organization, and placement of the proposed youth conservation corps pilot program are contained in the Department's report.

Appendix A

S.C.R. NO. 155 S.D. 1

THE SENATE FIFTEENTH LEGISLATURE, 1989 STATE OF HAWAII

SENATE CONCURRENT RESOLUTION

REQUESTING A STUDY ON THE FEASIBILITY OF A YOUTH VOLUNTARY SERVICE PROGRAM.

WHEREAS, the ability of Hawaii's youth to contribute to the solution of important issues in the State is generally underestimated; and

WHEREAS, Hawaii's youth can harness their determination, drive, and ability to aid the State in many areas of concern, such as conserving and maintaining our natural resources, or aiding the capabilities of persons of low income to improve the conditions of their lives; and

WHEREAS, these abilities have been demonstrated in the California Conservation Corps, which is operated by that State, and throughout the nation by the work of youths and others ages eighteen or older through the federal VISTA and Peace Corps programs; and

WHEREAS, the California Conservation Corps is a volunteer program in which young men and women engage in projects that preserve, maintain, and enhance environmentally important lands and waters; accomplish useful and needed public works projects in urban and rural areas; assist in emergency operations, including fire prevention and suppression; contribute to the conservation of energy; and contribute to making public facilities accessible to disabled persons; and

WHEREAS, volunteers are selected for participation in the Corps on the basis of their motivation for hard work, personal development, and public service; and

WHEREAS, the California Legislature has found that participation of California's youth in the Corps is beneficial to the youth by providing them with educational and work opportunities, as well as furthering their understanding and appreciation of natural resources, learning basic and fundamental work ethics, and learning the value of a day's work for a day's wages; and

S.C.R. NO. 5.D 1

WHEREAS, the VISTA program, although not just for young men and women, is another volunteer program that seeks to do good in the community, as well as providing the volunteers with experience, skills, and the satisfaction that comes from true achievement; and

WHEREAS, Hawaii's youth could also gain these tangible and intangible benefits from a similar program in Hawaii, while also aiding the State and its residents by improving the environment, helping the poor, and providing other appropriate services; now, therefore,

BE IT RESOLVED by the Senate of the Fifteenth Legislature of the State of Hawaii, Regular Session of 1989, the House of Representatives concurring, that the Legislative Reference Bureau is requested to study the feasibility of establishing a pilot program for volunteer youth services similar to the California Conservation Corps or the Volunteers in Service to America (VISTA) program; and

BE IT FURTHER RESOLVED that the Legislative Reference Bureau make recommendations concerning the appropriate agency to administer the pilot program; and

BE IT FURTHER RESOLVED that the Legislative Reference Bureau report findings and recommendations to the Legislature no later than twenty days before the convening of the Regular Session of 1990; and

BE IT FURTHER RESOLVED that a certified copy of this Concurrent Resolution be transmitted to the Director of the Legislative Reference Bureau.

Page 2

Appendix B

		ANM	NUAL ASSESSMENT RESULTS ON	USE OF VOLUNTEERS				
	FISCAL YEAR 80-81		FISCAL YEAR 81-82			FISCAL YEAR 82	82-83	
	VOI.UNTEERS UTILIZED	HOURS EXPENDED	VOLUNTEERS UTILIZED	HOURS EXPENDED	VOLUNTEERS UTILIZED	HOURS EXPENDED	MONETARY CONTRIBUTION	
GOVERNOR	93	55,461	984	16,267	300	12,016	\$ 7,300.00	
I.T. GOVERNOR	49	3,198	757	6,056	95	809	\$ 4,160.00	
LEGISLATURE	0	0	1	213.5	0	0		
JUDICIARY	171	16,405	241	20,928	201	41,390	\$ 767.70	
ACCOUNTING & GENERAL SERVICES	83	1,424	29	1,622	115	4,072	\$ 3,000.00	
AGRICULTURE	7	10.5	0	0	120	1,337		
ATTORNEY GENERAL	1	120	1	120	2	240		
BUDGET & FINANCE	1	400	1	20	0	0		
COMMERCE & CONSUMER AFFAIRS	362	2,736	170	823	236	662.5		
DEFENSE	24	687	28	876	28	839		
EDUCATION	31,166	347,963	27,982	297,508	26,649	501,634		
HAWAIIAN AFFAIRS	N/A	N/A	8	536	10	1,000		
HAWAIIAN HOME LANDS	102	358	1	1,040	7	3,949		
HEALTH	2,580	122,021	200,254	343,899,75	120,628	388,847.25	\$ 263,972.00	
LABOR & INDUSTRIAL RELATIONS	58	1,975	101	5,024	85	3,829.5		
LAND & NATURAL RESOURCES	441	37,282	441	37,282	507	34,077		
FERSONNEL SERVICES	0	0	0	0	7	715		
PLANNING & ECONOMIC DEVELOPMENT	63	4,152	193	6,084	177	8,252	\$ 400.00	
SOCIAL SERVICES & HOUSING	1,842	34,127	2,378	226,160	3,568	272,956	\$ 37,732.23	
TAXATION	6	195	215	2,880	232	3,245		
TRANSPORTATION	0	0	0	0	0	0		
UNIVERSITY OF HAWAII	1,077	20,339	4,332	463,917	174,116	333,036.78	\$ 450,319.50	
TOTAL	38,126	648,853.5	238,117	1,431,256.25	327,083	1,612,907.03	\$ 767,651.43	

FY 80-81 648,853.5 hours x \$3.35 minimum wage = \$2,173,659.23 Supplemental Government Services

FY 81-82 1,431,256.25 hours x \$3.35 minimum wage = \$4,794,708.44 Supplemental Government Services

FY 82-83 1,612.907.03 hours x \$3.35 minimum wage = \$5,403,238.55 Supplemental Government Services + \$767,651.43 Monetary Contribution + \$6,170,889.93

Revised 11/03/53

			AN	NUAL ASSESSMEN	T RESULTS ON U	SE OF VOLUNIEE	RS					
	F	ISCAL YEAR 83-	84			FISCAL YEAR 84	-85		FI	ISCAL YEAR 85-8	б	
DEPARTMENTS	VOLUNTEERS UTILIZED	HOURS EXPENDED		MONETARY NTRIBUTION	VOLUNTEERS UTILIZED	HOURS EXPENDED		MONETARY INTRIBUTION	VOLUNTEERS UTILIZED	HOURS EXPENDED		IONETARY
GOVERNOR	336	4,375	\$	19,638.01	821	12,114.50	\$	25,230.31	667	7,241	\$	29,720.86
LT. GOVERNOR	800	4,000	\$	460.00	1,006	18,585	\$	92,885.00	101	300	\$	75.00
LEGISLATURE	0	0			0	0			5	480		
JUDICIARY	477	75,764	\$	4,187.84	673	112,104.17	\$	4,087.82	647	82,835	\$	5,700.98
ACCOUNTING & GENERAL SERVICE:	S 220	3,747	\$	400.00	189	7,277	\$	900.00	905	13,893		
AGRICULTURE	113	1,940			129	3,203.25			98	1,692		
ATTORNEY GENERAL	0	0			14	1,170.75			8	1,214.50		
BUDGET & FINANCE	0	0			4	624			13	349		
COMMERCE & CONSUMER AFFAIRS	288	1,157.25	\$	100.00	309	1,460.50			232	1,682		
DEFENSE	28	839			28	700			28	2,083		
EDUCATION	48,291.50	519,387	\$	163,659.50	30,418	448,424	\$	128,562.00	35,027	506,294	\$	58,461.80
HAWALIAN AFFAIRS	5	150	\$	6,850.00	459	16,552	\$	25,00	78	1,845		
HAWAIIAN HOME LANDS	7	3,024			18	3,392			7	4,200		
HEALTH	77,631	358,169.25	\$	258,072.48	60,683	309,048.85	\$	258,315.27	54,822	327,408.12	\$	174,008.58
LABOR & INDUSTRIAL RELATIONS	103	3,026			153	3,450			99	3,513.50		
LAND & NATURAL RESOURCES	867	39,140.72	\$	1,000.00	943	34,015			1,781	36,311		
OAHU METROPOLITAN PLANNING O	RG. 23	1,570			21	348			36	369		
PERSONNEL SERVICES	7	700	\$	138,000.00	7	280			0	0		
PLANNING & ECONOMIC DEVELOPM	ENT 138+	8,212+			131	6,843	\$	1,140.00	296	6,942		
SOCIAL SERVICES & HOUSING	3,954	266,633	\$	16,800.00	2,264	247,192.75	\$	30,363.00	2,685	276,694.70	\$	34,662.65
TAXATION	255	3,635			379	6,035			302	4,407		
TRANSPORTATION	11	528			71	1,477			21	1,330		
UNIVERSITY OF HAWAII	8,127	379,905.50	\$	455,670.00	343,377	403,090,15	\$1	,081,427.84	231,741	348,879.80	\$1.	231,955.00
TOTAL	141,681.50	1,675,902.72	\$1	,064,837.83	442,097	1,637,386.92	\$1	,622,936.24	329,599	1,629,963.62	\$1	,534,584.87
FY 83-84 1.675.902.72 (houne v \$2.35	minimum wara	¢£	614 274 11 Sum	nlownetal Cour	mmont familia	~ 1	\$1 ACK 027 02	Monotony Contas	hution - \$6.57	0 11	1 04
FY 83-84 1,675,902.72 hours x \$3.35 minimum wage - \$5,614,274.11 Supplemental Government Services + \$1,064,837.83 Monetary Contribution = \$6,679,111.94												

ANNUAL ASSESSMENT RESULTS ON USE OF VOLUNTEERS

FY 84-85 1,637,386.92 hours x \$3.35 minimum wage - \$5,485,246.18 Supplemental Government Services + \$1,622,936.24 Monetary Contribution = \$7,108,182.42

FY 85-86 1,629,963.62 hours x \$3.35 minimum wage - \$5,460,378.13 Supplemental Government Services + \$1,534,584.87 Monetary Contribution = \$6,994,963.00

11/03/86

STATEWIDE VOLUNTEER SERVICES OFFICE OF THE GOVERNOR ANNUAL ASSESSMENT RESULTS ON THE USE OF VOLUNTEERS IN STATE GOVERNMENT

FY 85-86

FY 86-87

	Volunteers	Hours	In-Kind/Cash	Volunteers	Hours	In-Kind/Cash
DEPARTMENT	Utilized	Expended	Contributions	Utilized	Expended	Contributions
Governor	667	7,241.00	\$29,720.86	363	8,192.00	\$8,714.11
Lt. Governor	101	300.00	\$75.00	133	1,860.50	\$0.00
Legislature	5	480.00	\$0.00	42	673.50	\$0.00
Judiciary	647	82,835.00	\$5,700.98	637	123,707.00	\$1,820.91
Accounting and General Services	905	13,893.00	\$0.00	195	15,217.00	\$0.00
Agriculture	98	1,692.00	\$0.00	129	1,940.00	\$0.00
Attorney General	8	1,214.50	\$0.00	7	548.75	\$9,900.00
Budget & Finance	13	349.00	\$0.00	22	1,068.00	\$0.00
Business & Economic Devpmnt				319	6,474.00	\$900.00
Commerce & Consumer Affairs	232	1,682.00	\$0.00	23,419	7,465.25	\$871,234.00
Corrections				484	30,533.00	\$4,538.00
Defense	28	2,083.00	\$0.00	22	865.00	\$0.00
Education	35,027	506,294.00	\$58,461.80	36,364	489,245.00	\$86,686.00
Hawaiian Affairs	78	1,845.00	\$0.00	56	1,084.00	\$0.00
Hawaiian Home Lands	7	4,200.00	\$0.00	7	2,080.00	\$0.00
Health	54,822	327,408.12	\$174,008.58	8,204	137,713.25	\$130,539.50
Human Services				1,754	212,854.50	\$37,409.08
Labor & Industrial Relations	99	3,513.50	\$0.00	220	4,349.00	\$1,000.00
Land & Natural Resources	1,781	36,311.00	\$0.00	446	31,134.50	\$0.00
Oahu Metropolitan Planning Org					•	
Personnel Services	0	0	\$0.00	7	950.00	\$0.00
Planning and Economic Devpmnt	296	6,942.00	\$0.00			
Social Services & Housing	2,685	276,694.70	\$34,662.65			
Taxation	302	4,407.00	\$0.00	21	350.00	\$0.00
Transportation	21	1,330.00	\$0.00	51	2,010.00	\$0.00
University of Hawaii	231,741	348,879.80	\$1,231,955.00	8,556	254,676.50	
TOTAL	329,563	1,629,594.62	\$1,534,584.87	81,458		\$1,808,060.59
			TT 05 06			
Wage Value of Supplemental Govern	mont Convins	~	FY 85-86		FY 86-87	
In-Kind/Cash Contributions	ment Service	2	\$5,459,141.98		\$4,472,219.01	
			\$1,534,584.87		\$1,808,060.59	
Dollar Value of Supplemental Gove	ernment Servi	ces	\$6,993,726.85		\$6,280,279.60	

STATEWIDE VOLUNTEER SERVICES OFFICE OF THE GOVERNOR ANNUAL ASSESSMENT RESULTS ON THE USE OF VOLUNTEERS IN STATE GOVERNMENT

DEPARTMENT Governor Lt. Governor Judiciary Accounting and General Services Agriculture Attorney General Budget & Finance Business & Economic Devpmnt	Utilized 386 1,800 766 915 97 942 22 95 378	Expended 2,876.50 7,200.00 118,354.00 12,674.00 1,611.00 4,271.00 1,068.00 7,941.00	Contributions \$17,886.39 \$0.00 \$1,952.00 \$4,750.00 \$0.00 \$108,039.02 \$0.00
Lt. Governor Judiciary Accounting and General Services Agriculture Attorney General Budget & Finance	1,800 766 915 97 942 22 95	7,200.00 118,354.00 12,674.00 1,611.00 4,271.00 1,068.00	\$0.00 \$1,952.00 \$4,750.00 \$0.00 \$108,039.02
Lt. Governor Judiciary Accounting and General Services Agriculture Attorney General Budget & Finance	1,800 766 915 97 942 22 95	7,200.00 118,354.00 12,674.00 1,611.00 4,271.00 1,068.00	\$0.00 \$1,952.00 \$4,750.00 \$0.00 \$108,039.02
Judiciary Accounting and General Services Agriculture Attorney General Budget & Finance	766 915 97 942 22 95	118,354.00 12,674.00 1,611.00 4,271.00 1,068.00	\$1,952.00 \$4,750.00 \$0.00 \$108,039.02
Accounting and General Services Agriculture Attorney General Budget & Finance	915 97 942 22 95	12,674.00 1,611.00 4,271.00 1,068.00	\$4,750.00 \$0.00 \$108,039.02
Agriculture Attorney General Budget & Finance	97 942 22 95	1,611.00 4,271.00 1,068.00	\$0.00 \$108,039.02
Attorney General Budget & Finance	942 22 95	4,271.00 1,068.00	\$108,039.02
Budget & Finance	22 95	1,068.00	
• I	95		\$0.00
Pupinose & Economia Doummet		7.941.00	
Dusiness & Economic Devprint	378	,=	\$900.00
Commerce & Consumer Affairs		1,903.50	\$71.25
Corrections	318	21,220.00	\$6,740.00
Defense	22	865.00	\$0.00
Education	36,701	493,976.75	\$273,665.00
Hawaiian Affairs			
Hawaiian Home Lands	7	3,346.00	\$0.00
Health	135,476	212,284.75	\$463,607.07
Human Services	113	50,403.00	\$2,655.00
Labor & Industrial Relations	140	2,650.00	\$0.00
Land & Natural Resources	1,415	66,851.75	\$5,000.00
Personnel Services			
Taxation	26	416.00	\$0.00
Transportation	1	65.00	\$0.00
University of Hawaii	8,268	377,166.25	\$1,042,575.28
TOTAL	187,888	1,387,144	\$1,927,841.01
Wage Value of Supplemental Government Ser	vices	\$5,340,504.40	
In-Kind/Cash Contributions	910C3	\$1,927,841.01	
Dollar Value of Supplemental Government Ser		\$7,268,345.41	

FY 87-88

Appendix C

DIRECTORY OF MAJOR STATE GOVERNMENT VOLUNTEER PROGRAMS

DEPARTMENT ISLAND CONTACT ADDRESS PHONE

DEPARTMENT OF CORRECTIONS

Volunteers in Col	rrections (V	/olinCor)		
	Oahu	Ms. Jo desMarets	Volunteer Services Administrator Volunteers in Corrections (VolinCor) Institutional Support Services Department of Corrections 677 Ala Moana Boulevard, Suite 700	
			Honolulu, Hawaii 96813	548-3630
	Oahu	Mr. John Shinkawa	Community Services Administrator Hawaii Youth Correctional Facility Department of Corrections 42-477 Kalanianaole Highway Kailua, Hawaii 96734	261-0858
	Hawaii	Mr. Robert J.K. Roy	Correctional Counselor Hawaii Community Correctional Facil Department of Corrections 60 Punahele Street Hilo, Hawaii 96720	ity 961-7515
	Hawaii	Mr. James Bradley	Recreation Specialist Kulani Correctional Facility Department of Corrections Star Route 1, Stainback Highway	935-2280
	Kauai	Mr. Anthony Visaya	Hilo, Hawaii 96720 Corrections Supervisor Kauai Community Correctional Facility Department of Corrections 5350 Kuhio Highway Lihue, Hawaii 96766	
	Maui	Ms. Marye Deming	Social Worker Maui Community Correctional Center Department of Corrections 600 Waiale Drive Walluku, Hawaii 96793	245-9581

DEPARTMENT	ISLAND	CONTACT	ADDRESS	PHONE
DEPARTMENT O	F EDUCAT	ION		
Public School Pro	ograms	Ms. Kathleen Rozmiarek	Personnel Specialist III Employee Benefits Section Personnel and Industrial Relations B Department of Education	ranch
			P. O. Box 2360 Honolulu, Hawaii 96704	548-3884
Library for the Bli	nd and Phy	sically Handicapped		
	Oahu	Ms. Sue Sugimura	Public Services Librarian Library for the Blind and Physically Handicapped Department of Education 402 Kapahulu Avenue Honolulu, Hawaii 96815	732-5407
	Hawaii	Ms. Lucretia Pladera	Hawaii Library District Administrator Department of Education P. O. Box 647 Hilo, Hawaii 96720	935-5407
	Kauai	Ms. Donna Garcia	Kauai Library District Administrator Department of Education 4344 Hardy Street Lihue, Hawaii 96766	245-3617
	Maui	Ms. Joyce Van Zwalenburg	District Administrator Department of Education P. O. Box B 251 High Street Wailuku, Hawaii 96793	
DEPARTMENT OF		am		
	Hawaii	Mr. Ronald Okamura	Deputy Director of County Parks Department of Parks & Recreation County of Hawaii 25 Aupuni Street	
			Hilo, Hawaii 96720	961-8311

			LIDDDFAD	
DEPARTMENT	ISLAND	CONTACT	ADDRESS	PHONE
DEPARTMENT O	F HEALTH	(continued)		
Community Work		• •		
	Kauai	Mr. Beryl Moir	Kauai Beautification Committee P. O. Box 610	
			Koloa, Hawaii 96756	245-3387
	Maui	Ms. Jan Dapitan	Director, Maui Community Work Day 275 Uhu Street	-
			Kahului, Hawaii 96732	244-7855
Litter Control Offi	се			
		Mr. Clyde Morita	Administrator, Litter Control Office	
			Department of Health	
			205 Koula Street Honolulu, Hawaii 96813	548-6444
			HUHUHU, HAWAII 30013	04070444
Hospitals, County				
Hawaii State Ho	ospital	Mr. Calvin Arata	Volunteer Services Coordinator	
			Hawaii State Hospital Department of Health	
			45-710 Keaahala Road	
			Kaneohe, Hawaii 96744	247-2191
				ext. 298
Waimano Traini	ng School	and Hospital		
	u	Mr. James Tateyama	Volunteer Services Coordinator	
			Waimano Training School and Hospit	al
			Department of Health	
			Mental Health Division 2201 Waimano Home Road	
			Pearl City, Hawaii 96782	456-6266
				=
DEPARTMENT OF	-			
Foster Grandpare				
	Oahu	Ms. Rene Nakama	Project Director	
			Foster Grandparent Program	
			Department of Human Services Waimano Training School and Hospit	al
			2201 Waimano Home Road	4.41
			Pearl City, Hawaii 96782	455-5381

DEPARTMENT	ISLAND	CONTACT	ADDRESS	PHONE
DEPARTMENT OF	HUMAN S	SERVICES (continued)		
Foster Grandpare				
	Hawaii	Ms. Ellen Pavao	Supervisor Foster Grandparent Program Department of Human Services 75 Aupuni Street Hilo, Hawaii 96720	961-7451
Senior Companior	n Program			
	Oahu	Mr. Gerald Sumida	Program Director Senior Companion Program Family and Adult Services Division Department of Human Services P. O. Box 339	
			Honolulu, Hawaii 96809	548-4177
	Hawaii	Ms. Lillian Tim Sing	Coordinator Hawaii County Office on Aging 127 Kamana Street Hilo, Hawaii 96720	961-3418
	Hawaii	Ms. Jenny Uemura	Senior Companion Program Kona Coordinated Services P. O. Box 955 Kona, Hawaii 96740	323-2085
	Kauai	Mr. Terry Hill	Senior Companion Program Kauai Office of Elderly Affairs 4396 Rice Street Lihue, Hawaii 96766	245-4737
	Maui	Ms. Jo Reyes	Senior Companion Program Senior Outreach Division Department of Human Concerns 404 Alakapa Place Paia, Hawaii 96779	877-7212
	Molokai	Ms. Rose Siquian	Molokai Senior Outreach Program P. O. box 161 Kualapuu, Hawaii 96757	553-5241

DEPARTMENT	ISLAND	CONTACT	ADDRESS	PHONE
DEPARTMENT	OF HUMAN	SERVICES (continued)		
Services for the	Blind			
		Ms. Jere Masumoto	Community Services Coordinator Services for the Blind Branch Vocational Rehabilitation Division Department of Human Services 1901 Bachelot Street Honolulu, Hawaii 96817	548-7409
			·····	
Volunteers in P			Malumtuna Consisso Conseliantes	
	Oahu	(Currently vacant)	Volunteer Services Coordinator Public Welfare Division Department of Human Services P. O. Box 339 Honolulu, Hawaii 96809	548-5852
	Hawaii	Mr. Andrew Higa	Administrator Department of Human Services P. O. Box 1562 Hilo, Hawaii 96720	961-7251
	Kauai	Ms. Georgia Meyer	Administrator	
	Nadai	MB. GEOIGIA MEYO	P. O. Box 1707	
			Lihue, Hawaii 96766	245-4350
	Maui/ Molokai	Mr. Steve Maenaka	Administrator 54 South High Street, Room 125 Wailuku, Hawaii 96793	244-4256
THE JUDICIARY	,			
Program Servic	es	Mr. Earl Yonehara	Administrator, Program Services	
			The Judiciary P. O. Box 2560	
			F. O. Box 2560 Honolulu, Hawaii 96804	548-5805
Volunteers in Putto the Courts	ublic Service			
	Oahu	Mr. Norman Fujioka	Program Manager Volunteers in Public Service to the C The Judiciary	Courts
			P. O. Box 2560	

DEPARTMENT	ISLAND	CONTACT	ADDRESS	PHONE
DEPARTMENT O	F HUMAN S	SERVICES (continued)		
	Hawaii	Ms. Ellen Yasukara	Program Services Manager Judiciary Program Services Office The Judiciary P. O. Box 1007 Hilo, Hawaii 96721	961-7311
	Kauai	Mr. Dennis Williams	Program Services Coordinator Judiciary Program Services Office The Judiciary P. O. Box 191 Lihue, Hawali 96766	245-4313
	Maui	Ms. Gail Nakamae	Program Services Coordinator Judiciary Program Services Office The Judiciary 2145 Main Street, Suite 226 Wailuku, Hawaii 96793	244-9014
Community Serv	ice Sentenc	ing Program		
		Mr. Calvin Sagara	Program Manager Community Service Sentencing Prog Program Services Branch The Judiciary P. O. Box 2560 Honolulu, Hawaii 96804	jram 548-8912
	_			
Foster Parenting	Program	Ms. Nancy Donnelly	Program Manager Foster Parenting Program The Judiciary P. O. Box 2560 Honolulu, Hawaii 96804	548-2001
Guardian Ad Lite	m Program	Mr. Rodney Hee	Program Manager Guardian Ad Litem Program Program Services Office The Judiciary P. O. Box 2560 Honolulu, Hawaii 96804	548-2137

DEPARTMENT	ISLAND	CONTACT	ADDRESS	PHONE
·····			izen Malder (
DEPARTMENT	OF LAND A	ND NATURAL RESOURCES		
Conservation a	nd Resource	9S		
Enforcement (-			
	Oahu	Mr. Noah Pekelo	Chief of Program Support Services	
			Conservation and Resources Enfor Department of Land and Natural Re	
			1151 Punchbowl Street, Room 311	3001023
			Honolulu, Hawaii 96813	548-5919
	Kauai	Mr. Tatsuo Nakamura	Kauai Branch Chief	
			Conservation and Resources Enfor	cement Prg.
			Department of Land and Natural Re 3060 Eiwa Street	sources
			Lihue, Hawaii 96766	245-4444
	Maui	Mr. Keith Keau	Maui Branch Chief	
			Conservation and Resources Enforce	
			Department of Land and Natural Re	sources
			54 High Street Wailuku, Hawaii 96793	244-4352
				211 1002
Volunteer Suppo				
	Statewide	Mr. Ralston Nagata	State Parks Administrator	
			State Parks Division Department of Land and Natural Re	sources
			P. O. Box 621	3001063
			Honolulu, Hawaii 96809	548-7455
	Oahu	Mr. Richard Yoshimura	District Superintendent	
	Ound		State Parks Division - Outdoor, Rec	reation
			and Historic Sites	
			Department of Land and Natural Res	sources
			1151 Punchbowl Street, Room 310	
			Honolulu, Hawaii 96813	548-7455
	Hawaii	Mr. Charles Supe	District Superintendent	
			State Parks Division	
			Department of Land and Natural Res	sources
			75 Aupuni Street	061 7000
			Hilo, Hawaii 96720	961-7200

DEPARTMENT ISLA	ND CONTACT	ADDRESS	PHONE

DEPARTMENT OF LAN	D AND NATURAL RESOURCE	S (continued)	
Maui		Maui District Superintendent	
		Department of Land and Natural Re 54 High Street	sources
		Wailuku, Hawaii 96793	244-4324
Hunter Education Progr		Uniter Education Coordinator	
	Mr. Wendell W.S. Kam	Hunter Education Coordinator Conservation and Resources Enfor	cement Div.
		Department of Land and Natural Re	esources
		1151 Punchbowl Street, Room 311 Honolulu, Hawaii 96813	548-3966
Special Programs Iolani Palace	Ms. Alice Gill	Iolani Palace	
ioiani raiace	WS. Ance an	P. O. Box 2259	
		Honolulu, Hawaii 96809	536-3552
Queen Emma's Sumn	ner Palace		
and Hulihee Palace		Daughters of Hawaii	
		Department of Land and Natural Re 2913 Pali Highway	esources
		Honoulu, Hawaii 96817	595-3167
DEPARTMENT OF TAX	ATION		
Volunteer Income Tax		Chief Teurseurs Candiana Branch	
Assistance Program	Ms. June Yamamoto	Chief, Taxpayer Services Branch Department of Taxation	
		P. O. Box 259	
		Honolulu, Hawaii 96809	548-3267
4-H Youth Education ar			
Development Program			
State	wide Dr. Richard Barker	State 4-H Leader	
		University of Hawaii 3050 Maile Way	
		Honolulu, Hawaii 96822	948-6736
Oahu	Mr. Steve Nagano	Coordinator	
Callu	n mit, dieve mayano	Cooperative Extension Service	
		45-260 Waikalua Road	A 2000 10 1 - 1
		Kaneohe, Hawaii 96744	247-0421

DEPARTMENT	ISLAND	CONTACT	ADDRESS	PHONE
L	<u></u>			
UNIVERSITY OF	HAWAII (co	ntinued)		
	Hawaii	Ms. Carol Ikeda	Coordinator Cooperative Extension Service P. O. Box 208 Kealakekua, Hawaii 96750	322-2718
	Kauai	Ms. Sarah Effken Purcell	Coordinator Cooperative Extension Service P. O. Box 3150 Lihue, Hawaii 96766	245-4471
	Maui	Ms. Mabel Sakuma	Coordinator Cooperative Extension Service 310 Kaahumanu Avenue Kahului, Hawaii 96732	244-3242
Family Communit	ty Leadersh	ip		
	Statewide	Ms. Jean Young	State Coordinator Family Community Leadership Progr Cooperative Extension Service University of Hawaii at Manoa 3050 Maile Way Honolulu, Hawaii 96822	am 948-7711
	Hawaii Cor	unty	·····	
		Dr. Tadashi Higaki	County Administrator Family Community Leadership Progra 875 Komohana Street Hilo, Hawaii 96720	am 959-9155
	Hawaii (Hilo)	Ms. Betty Jo Thompson	Cooperative Extension Service University of Hawaii at Hilo 875 Komohana Street Hilo, Hawaii 96720	959-9155
	Hawaii (Kamuela)	Ms. Paulita Mahilum	Family Community Leadership Progra P. O Box 237 Kamuela, Hawaii 96743	am 775-0369
	Hawaii (Kona)	Ms. Carol Ikeda	Family Community Leadership Progra P. O. Box 208	
			Kealakekua, Hawaii 96750	322-2718

Appendix D

 $\begin{array}{c} \mbox{Calendar No. 331} \\ \mbox{101st Congress} \\ \mbox{1st Session} \end{array} \end{array} \\ \begin{array}{c} \mbox{Senate} \\ \mbox{Senate} \\ \mbox{Senate} \\ \mbox{101-176} \end{array}$

NATIONAL AND COMMUNITY SERVICE ACT OF 1989

OCTOBER 27 (legislative day, SEPTEMBER 18), 1989.—Ordered to be printed

Mr. KENNEDY, from the Committee on Labor and Human Resources, submitted the following

REPORT

together with

ADDITIONAL AND MINORITY VIEWS

[To accompany S. 1430]

The Committee on Labor and Human Resources to which was referred the bill (S. 1430) to enhance national and community service, and for other purposes, having considered the same, reports favorably thereon with an amendment in the nature of a substitute and recommends that the bill as amended do pass.

CONTENTS

-	
1.	Background and need for legislation
11.	History of the legislation
ШІ,	Hearings
IV.	Committee views
٧.	Votes in Committee
VI.	Cost estimate
VII.	Regulatory impact statement
III.	Section-by-section analysis
IX.	Additional and minority views
Х.	Changes in existing law

Strike all after the enacting clause and insert the following:

SECTION 1. SHORT TITLE AND TABLE OF CONTENTS.

(a) SHORT TITLE.—This Act may be cited as the "National and Community Service Act of 1989".

23-311

**Star Print

sevoliol as a structure lo side of the renarmed to aller (d)

Sec. 2. Short title and table of contents. Sec. 3. Purposes. Sec. 3. Purposes.

Sec. 132. Work study programs.

Sec. 401. Definitions. Sec. 402. Beachimhment of Corporation: application of District of Columbia Nonprofit Corporation Act. Sec. 403. Board of directors.

See 301 Short Little See 301 Short Little See 300 Chambra of service See 300 Training See 300 Trai

JILTE HI-HVLIONT' SEXAICE DEM Sec. 213 Yoppetrepou q abidokrenejoor Sec. 214 Appetrepou q abidokrenejoor Sec. 215 Taipt Bijonavia Sec. 215 Taipt Bijonavia Sec. 215 Taipt Bijonavia Sec. 215 Taipt Bijonavia Sec. 215 Taiptensenjo Caletre jo eurojonavi Sec. 215 Taiptensenjo Caletre jo Sec. 205 Taiptensen o Logit Vene Sec. 205 Taiptensen o Logit Vene Sec. 205 Sec. 206 Jaun Boby Sec. 205 Bujisterna Sec. 205

SOLARISS TWOOLLYN ROM NOLLWROAROD- AL WILLI.

TITLE III-WATOWAL SERVICE DEMONSTRATION PROCEAM

PUTLE II-YOUTH SERVICE CORPS

NOTADUG MERGIN TO RECEIPTING AL REPUBLIC OF REACH THREE WAY AND THE REACH AND THE REAC

BRARDORT TOURD REA THARD EVERIDAL THROUTO FRATE-D TRAT

employed workers, and

It is the purpose of this Act to-

lorce in addressing national problems;

poverty in the United States;

--- Jadi abnil mergao. BONIGNIA 1 'DBR

Sec. 501. Short title. Sec. 502. Authorization of appropriations.

POI SPOR NOS

Sec. 601. Short fills. Sec. 602. Programs of national and local alguificance. Sec. 603. Authorization of appropriations.

See, MA Officers and employees See, 405, Officers and employees See, 405, Housing and education rouchers, living allow See, 405, Housing and Artestan See, 405, Housing and Entertain See, 413, Hondiagramma and Entertain See, 413, Hondiagramma and Entertain See, 413, Hondiagramma See, 414, Hondiagramma See, 41

404. Officers and employees. 405. Nonprofit and nonpolitical nature of the Corporation. 405. Housing and education rouchers; living ellowances. 417. Burnets.

programe that involve a substantial time commitment;

SEC 2 LINGDORES สกรวกระกร

for all citizens, particularly youth and older Americans. (6) involve volunteers in activities that would not otherwise be performed by for all citizers to retricularly could allocate the other there been created by high provide the other and local service the format framework of Pederal. State, and local (5) build on the exteribing organizational framework of Pederal. State, and local programs and agreement of expand full-time and pertine agreement of expanding the service of perturbine the perturbine for all citizers.

lanoitan of insmitmumos beniataus a susan of anaprenik grund sidans (4) full-time or part-time service to the Nation; (3) begin to call young people to serve in national volunteer programs; (1) romewithe ethics force responsibility in the United States; (2) some the ethics force responsibility in the United States; (2) ask citizens of the United States, regardless of age or income, to engage in

tunities should be made available to all citizens, particularly youth and older

(10) a range of full-time and part-time mational and community service oppor-(9) many potential volunteer acannot participate in a full-time volunteer service program; to program, but should have the option of part-time service in such a program;

5001 et sech provide 500,000 hours of service each year and are a vital (8) older Americane, through the Older American Voluncer Programs (as es-tablished by title II of the Domestic Voluncer Service Act of 1973 (42 U.S.C.

ing deter many young adults from volunteering for VISTA and other service (7) the cost of higher education, loan indebtedness, and the high price of hous-

1973 (42 U.S.C. 4951 et seq.), is one of the most cost effective means to fight (6) conservation and sugary interpret interpret interpret of the formation activity activity activity interpret in the service corps provide important benefits to (a) the Volunders in Service to America Frequent (b) the Volunders in Service to America Frequent (c) the Volunder in the Service to America Frequent (c) the Volunders in Service to America Frequent (c) the Frequent (c) the Frequent (c) the Present (c) the Monte (c) the Present (c) the Monte (c) the Present (c) the Present (c) the America Frequent (c) the Monte (c) the Present (c) the Monte (c) the Present (c) the Monte (c) the Monte (c) the Present (c) the Monte (

(3) serving others builds self-esteem and teaches teamwork, decision making, (1) set vice of community and use commercial an exponential of such a community of such commercial and such a commercial such a commercial such as a commerc

(1) service to the community and the Mation is a responsibility of all citizens

STATE VI-VATIONAL OLDER AMERICANS VOLUTTEER PROGRAMS

LILL'E A - EXLY/RION OF ADTUMLEERS IN SERVICE TO AMERICA

8

offer a powerful and largely untapped resource for community service; and problem-solving; (4) the 70,000,009 youth of the United States who are between ages 5 and 25

LILTE 1-BCHOOF BYSED COMMONILL SERVICE

Sec. 161. Loan forgive date. Sec. 162. Effective date.

49

- Sec 111, Authorization of appr Sec 101, Structure outpound Sec 100, Structure Sec 100, Event activition Sec 202, Event activition Sec 202, Event activition Sec 202, Event activition Sec 202, Program activition Sec 203, Broot structure Sec 203, Br

Sec. 151. Loss cancellation authorized. Sec. 152. Effective data:

- anoitaingongaa la noitaninodtuk. [[1]. 208

Sec 144 Dates on deformants and cancellations. Sec 144 Est competing for sourcements Sec 144 Est competing for an deformants and cancellations.

- Soc. 121. Innovative projects for community service.

(7) generate 1,000,000,000 additional volunteer service hours to help meet human, educational, environmental, and public safety needs, particularly those needs relating to poverty.

TITLE I-SCHOOL BASED COMMUNITY SERVICE

PART A SCHOOL BASED COMMUNITY SERVICE

SEC. 1at. SHORT TITLE.

This part may be cited as "Serve America, the Service to America Act of 1989". SEC. 182. PROGRAM AUTHORIZED.

(a) IN GENERAL -- The Secretary is authorized, in accordance with the provisions of this Act, to make grants to States or local applicants to create or expand service opportunities for students and out-of-school youth and for community members, particularly senior citizens, to volunteer in schools.

(b) TERM OF GRANT .- The term of the grants may be for a period of not longer than 3 years.

SEC. 101. DEFINITIONS.

For purposes of this part-

(1) COMMUNITY-BASED AGENCY. The term "community-based agency" means a private nonprofit organization that is representative of a community or a significant segment of a community and that is engaged in meeting human, educational, or environmental community needs. (2) EDUCATION INSTITUTION.—The term "education institution" means a local

educational agency, elementary or secondary school or a community-based

agency that provides educational services. (3) EDUCATION PARTNERSHIP PROGRAM.—The term "education partnership program" means a program in which school volunteers work in an educational institution in support of the school's objectives to enhance the education of students

(4) ELEMENTARY SCHOOL.—The term "elementary school" has the same mean-ing given that term in section 1471(8) of the Elementary and Secondary Educa-tion Act of 1965 (20 U.S.C. 2891(8)).

(5) INSTITUTION OF HIGHES EDUCATION. - The term "institution of higher education" has the same meaning given that term in section 1201(a) of the Higher Education Act of 1965 (20 U.S.C. 1141(a)).

(6) LOCAL EDUCATIONAL AGENCY. -- The term "local educational agency" has the same meaning given that term in section 1471(12) of the Elementary and Secondary Education Act of 1965 (20 U.S.C. 2891(12)).

(7) LOCAL GOVERNMENT AGENCY.-The term "local government agency" means a public agency that is engaged in meeting human, social, educational, or environmental needs.

(8) OUTO-PSCHOOL YOUTH.-The term "out-of-school youth" means an individ-ual who has not attained the age of 25, has not completed college or the equivalent, and is not enrolled in an elementary or secondary school or institution of higher education.

(9) PARTICIPANT .- The term "participant" means a student or out-of-school youth who provides services pursuant to a program funded under this title.

(10) PARTNERSHIP PROGRAM.—The term "partnership program" means a coop-erative effort to enhance the education of students among an education institu-tion and one or more of the following:

(A) the private sector;
(B) public and non-profit agencies;
(C) institutions of higher education; and

(D) community organizations.

(11) PLACEMENT.-The term "placement" means the matching of a participant or team with a specific project.

(12) PROJECT.—The term "project" means any activity that results in a specific identifiable service or product that otherwise would not be done with existing funds, and which shall not duplicate the routine services or functions of the employer to whom participants are assigned.

(13) SCHOOL VOLUNTEER.-The term "school volunteer" means a person beyond the age of compulsory schooling, including an older American, an employee of a private business, an employee of a public or nonprofit agency, or any

other individual working without financial remuneration and under the direc-tion of professional staff within a school or school district.

(4) SECONDARY SCHOL.—The term "secondary school" has the same meaning given that term in section 1471(21) of the Elementary and Secondary Education Act of 1965 (20 U.S.C. 2891(21)). (15) SECERTARY.—The term "Secretary" means the Secretary of Education.

(16) SERVICE OFFORTUNITY. -- The term "service opportunity" means a program or project enabling students or out-of-school youth to perform meaningful and or project ensuing students or out-or-achool youth to perform meaningful and constructive service in agencies, institutions, and situations where the applica-tion of human talent and dedication may help to meet human, educational, and environmental community needs, especially these relating to poverty. (17) STATE.—The term "State" means a State, the Commonwealth of Puerto Rico, Guam, the District of Columbia, American Samos, the Virgin Islands, the Commonwealth of the Northern Mariana Islands, the Republic of the Marshall label of Mariana State States of Mariana Palacian and States of the Marshall

Islands, the Federated States of Micronesia, or Palau.

(18) STATE AGENCY FOR HIGHER EDUCATION. "The term "State agency for higher education" means the State board of higher education or other agency or officer primarily responsible for the State supervision of higher education, or. if there is no such officer or agency, an officer or agency designated by the Governor or by State law.

(19) STATE EDUCATIONAL AGENCY .- The term "State educational agency" has the same meaning given that term in section 1471(23) of the Elementary and Secondary Education Act of 1965 (20 U.S.C. 2891(23)).

(20) STUDENT .-- The term "student" means an individual who is enrolled fulltime or part-time in an elementary or secondary school or institution of higher education.

(21) STUDENT COMMUNITY SERVICE PROGRAM.-The term "student community service program" means a program in which students or out-of-school youths are offered service opportunities in the community or an educational institution.

SEC 104 ALLOCATION OF FUNDS.

(B) GRANTS TO STATES.-

(1) IN GENERAL -- The Secretary shall use 85 percent of the funds appropriated under section 111 to make grants to States that have submitted applications under section 105.

sider the number of students enrolled in elementary and secondary schools in the State, the quality of the proposal, and evidence of State commitment to the program.

(B) GRANT FORMULA.—If more than \$50,000,000 is appropriated for this part, then the Secretary shall allocate 85 percent of the funds appropriated part, then the Secretary shall allocate of percent of the threat application under section 110 according to the chapter 1 basic grant formula described in section 1005 of chapter 1 of title 1 of the Elementary and Secondary Edu-cation Act of 1965 (20 U.S.C. 2711) to States that have applications approved under section 105.

(b) REMAINDER.-The Secretary shall use the remaining 15 percent of the funds appropriated under section 111 for program support, evaluation, training, technical assistance, and activities described in section 109

described in section 106 in evaluating such local applications.

SEC. 195. STATE APPLICATION.

Each State desiring to receive a grant under this part shall, through the State educational agency, submit an application to the Secretary at such time and in such manner as the Secretary may reasonably require. Each of such applications shall describe how-

(1) local applications will be ranked according to criteria described in section 106;

(2) service programs in the State will be coordinated;

(3) cooperative efforts among education institutions, local government agencies, community-based agencies, businesses, and State agencies to provide serv-ice opportunities, including those that involve the participation of urban, subur-ban, and rural youth working together will be encouraged;

(C) set forth the target numbers of students and out-of-school youth who will perticipate in the program assisted under this part and the target tor participation;

(A) describe an age-appropriate learning component for participants that (A) describe an age-appropriate learning component for participants to reflect on service superiences and appected learning outcomes; (B) describe whether or not the participants will receive academic credit for participation;

some and pericipants or achool volutieers: (1) describe the means by which exemplary service will be recognized; (1) describe the means by which exemplary service will be recognized; (2) describe the means by which exemplary service will be recognize and (3) describe the made, the program will comult with any local labor organize tion representing employees in the artes who are engaged in the same or (3) Strugger or the starter program will comult with any local labor organize a student representing employees in the artes who are engaged in the same or (3) Strugger of the program will comult with any local labor organize a student community eravice program, in addition to the above information, (3) Strugger and a service program, in addition to the above information, (3) Strugger and a service program, in addition to the above information, (3) Strugger and a service program, in addition to the above information, (3) Strugger and a service program, in addition to the above information, (3) Strugger and a service program, in addition to the above information, (4) Strugger and a service program, in addition to the above information, (4) Strugger and a service program, in addition to the above information, teach applicant aball-

sors and participants or school volunteers;

and the convolution of the provided;
 (1) describe how the participants or school volunteers will be recruited, including any special efforts to recruit out-of-achoot, and responsibilities of the program;
 (3) describe presents and operation of the program;
 (4) state the name, if available, qualifications, and responsibilities of the program;
 (4) state the name, if available, qualifications, and responsibilities of the program;
 (5) describe presents and operation of the program;
 (6) state the name, if available, qualifications, and responsibilities of the program;
 (6) state the name, if available, qualifications, and responsibilities of the program;
 (7) state the name, if available, qualifications, and responsibilities of the program;
 (7) state the name, if available, qualifications, and responsibilities of the program;
 (7) state the name, if available, qualifications, and responsibilities of the program;
 (7) state the name, if available, qualifications, and responsibilities of the program;
 (7) state the name, if available, qualifications, and responsibilities of the program;
 (7) state the name, if available, qualifications, and responsibilities of the program;
 (6) state the name, if available, qualifications, and responsibilities of the program;
 (7) state the name, if available, qualifications, and responsibilities of the program;

volunteers and the community.

bers, labor, and business; (C) describe the goals of the program, including goals that are quantifi-able, measurable, and demonstrate benefits to the participants or school able, measurable, and terministic

(B) specify the memberabip and role of an advisory committee consisting agencies, students, parents, teachers, administrators, school board mem-

(A) IN CENERAL—Any education institution, local government agency, community-based agency or consortia thereof desiring to receive a grant under this part shall form a pertnership consisting of 1 or more aducation institutions and 1 or more local community for any aducation

-,чинаялила (I)

-, GRAUGER NOTTADLITE (8)

SEC. IN LOCAL APPLICATION.

ы С

ment of a broad range of organizations, particularly community-based organiza-(9) disseminations of information and outreach will be used to ensure involve. pup innot loodoo lo-tuo bun

(8) non-Federal funds will be used to expand service opportunities for students sound add

tribal areas that exist in such States, (7) technical assistance and training will be provided to service programs in

usis with handloapping conditions, are assured service opportunities: (6) programs receiving funds under this title will be evaluated; (6) programs funded under this pair will serve urban sud turel ereas and any

(4) economically and educationally disadvantaged students, including individ-

be used by a local applicant to pay more than-(i) 80 percent of the costs of programs assisted under this title for the first year in which the applicant receives funds under this part;

margorg wen a statifit of to margorg guiteize as avorgam

prevention, or nutrition and health education.

'efin siun Lepur

---- MEARS LARE (8) SEC. 147, FEDERAL SRARE.

ineers entooni-wol

ture: Soud avp

---'ALGIORA (3) their teachers.

- TVAONAAV (Q) .elsinqonqqs sa

10100107

SELAICE DLOJECE

pue isquedion

(1) For the state of the state of the state of the costs of the state of the state

year period and are renewable for a second period of up to 3 years to expand or (d) Dusarrow.-Grants to local applicants under this part may be for up to a 3-

teer and meriding programs-(A) involving electroning for community; or private business in the community; or (A) which from on drug and alcohol abuse prevention, school drop-out (A) which from on drug and alcohol abuse prevention, school drop-out

(2) School voluments and schwarents recordent agency shall give priority to the rate and perturbation for the State educational agency shall give priority to

(W) programs which involve individuals of different ages, races, sexes, (W) programs which involve individuals of different ages, races, sexes, ethnic groups, and economic backgrounds serving together; and (D) in the case of applicants that are educational institutions, programs that are indegrated into the academic program. Thon, wor intrast awn sarwaxathe reconstance.

(B) programs in greatest need of assistance, such as programs targeting

(B) opportunities for school volunteers to work with st-risk children or

(A) preservice and inservice training for both supervisors and school volunceel and pertnerably programs, the State educational agency shall only ap-prove local applications describing programs that provide-(3) SCHOOL VOLUTIER AND PARTNERIN PROGRAMS.--In the case of school vol-

of to triantimumo beniateus a sham liw atnequiting tail somebive (O) will be provided; and involving representatives of the community where service opportunities

service experiences; service the supervice training for both supervisors and pervice the supervice and the supervice training of the supervise service training to the supervise service s plications describing programs that provide (A) an age appropriate learning component for participants to reflect on ty service programs, the State educational agency shall approve only local ap-

(1) STUDENT COMMUNITY EXENTS PROCEAME.-In the case of student communi-

options and their benefits, such as student losn deferment and forgiveness,

Service Demonstration programs funded under this Act, and other service

(G) provide assurances that participants will be provided with informa-tion on VISTA, the Feace Corps (as setablished by the Feace Corps Act (22 U.S.C. 2501 et ac.)), chapter 30 of title 38, United States Code, chapter 106 of title 10, United States Code, full-time Youth Service Orps and Vational Service Theorematic States Code, full-time Youth Service Orps and Vational

(F) include other relevant demographic information about expected per-

as a group; the entropertion of expected participents who are educationally or economically disedvantaged, including participents with handicapping

numbers of hours of service such participants will provide individually and

cational agency shall give priority to applications describing-(I) IN ORNERAL -In providing assistance pursuant to this part, the State edu-

pus the second year in which the applicant receives funds under this title for (iii) 60 percent of the costs of programs assisted under this title for the third year in which the applicant receives funds under this part; and

(ii) 70 percent of the costs of programs assisted under this title for

costs of programs assured under this part to be paid by a local appli-costs of programs assured under this part to be paid by private profit making organizations then subsection (a) shall be applied by substitut-

austed under this part that are to be paid by a local applicant from sources other than Federal funds may be paid in cash or in kind, fairly (i) NON-FEDERAL SOURCES -The portion of the costs of programs as-

(iv) 50 percent of the costs of programs assisted under this title for the fourth and each successding year in which the applicant receives

delivery. Such evaluations shall include, where appropriate, comparisons with aption to cost, the impact on related programs, and the structure and mechanisms for to determine program effectiveness in achieving stated goals in general and in relathat measure and evaluate the impact of programs authorized by this part, in order

extremp: Such evaluations shall include, where appropriate, comparisons with appropriate control groups composed of individuals who have not participated in such programs. Evaluations shall be conducted by individuals not directly involved in the stratistic of particular shall be conducted by individuals not directly involved in (b) Stanbaus-The Secretary shall develop and publish general standards for clude the opinions of programs under this part, the Secretary shall evaluate and endities where appropriates on programs of programs and the stratific and endities where set latents are dominantly involved in (c) invort.—In carrying the attentificant methods of the results of vorgrams (d) Publicanty.—The Secretary shall publish summeries of short programs (d) Publicanty...

tions of program impact and effectiveness no later than 60 days after the completion

noiseulave nous lo

United States. (e) Ownerstatter or PROPERTY.—All studies, evaluations, proposals, and data pro-duced or developed with assistance under this part shall become the property of the

SEC. III, AUTHORIZATION OF APPROPRIATIONS.

There are authorized to be appropriated to carry out this part, \$65,000,000 for

facal year 1991, and such sums as may be necessary for each of the 4 succeeding

PART B-HIGHER EDUCATION

semended to read as follows: Part C of Title X of the Higher Education Act of 1965 (20 U.S.C. 1135e et seq.) is SEC. 111 INNOVATIVE PROJECTS FOR COMMUNITY SERVICE.

PART C-INNOVATVE PROJECTS FOR COMMUNITY SERVICE

RATEMENT OF PURPOSE

"Sac. 1061. It is the purpose of this part to support innovative projects to determine the feasibility of encouraging student participation in community service scritties before, during, or after the completion of such student's higher education.

EDIABES ALINGHINGO ROA SLOEPOES HAILYAONNI

agencies and nonprofit organizations working in partnership with institutions of higher education. for purposes including, but not limited towith the provisions of this part, to make grants to, and contracts with, institutions of higher education (including combinations of such institutions), and other public "Sac. 1062. (a) Свижаль Аυтновиту.-Тhe Secretary is authorized, in accordance

"(1) encouraging students to participate in community service activities that will engender a sense of social responsibility and commitment to the

Anunuuo:

(2) Community, (2) Community, (2) Community service activity (2) Community service activity (2) Community service activity field in the course of completing postsecondary education; (3) reaction; (3) reaction; (3) reacting reaction; (3) reacting reacting activity service activity activit

(4) encouraging the integration of community service into academic curricupup impeces and

BNOISIAOBI MALLVELSINIMOV,

"Director") at such time, in such manner, and contained or accompanied by such tered into, under section 1062 unless an application is made to the Director of the fund to under section destry Education (hereinalter referred to as the "SEC. 1063. (a) APPLICATION -- No grant may be made, and no contract may be en-

"(b) Consistent with the provisions of section 1003(c), the National Board of the Programs, priorities, and the selection of projects developed under the Birector on section 1062. miormation as the Director may reasonably require

continuing evaluation of programs assisted under this part, including evaluations (a) EVALUATION .-- The Secretary shall provide, through grants or contracts, for the

NOLLVIPITAS '611 '33R

(с) Імиоуллук сливисиция матекімія.-Тіре бесгесагу із ацтогілее to make

ble to receive grants under paragraph (1) of this subsection: (3) Distantiation about national model youth service programs funded under para-information about national model youth service programs funded under para-rearch (1) of this subsection

(2) ELOURILITY —States, education institutions, local government agencies, community-based agencies, or consorts of the source or source shall be eligible to the source or source and the transmission and the source of the s

(3) FUNCTION OF CLEARINGHOUSES — Mational and regional clearinghouses (a) FUNCTION OF CLEARINGHOUSES — Mational and regional clearing the funded under this part. (Under under this part.)

agencies with extensive experience in student community service and school volunteer and partnership programs shall be eligible to receive lunds under paragraph (1) of this subsection.

pend, allowance, or other financial support to any participant except reimbureement to transportation, meaks, and other reasonable out-of-pocket expenses incident to participation in a program assisted under this part.

for supervision of participants, program administration, training, reasonable trans-for supervision costs, insurance, and other reasonable expenses (c) Strustum.--Funds provided under this part shall not be used to pay any sti-cond allocation or other function

nical assistance, curriculum development, and coordination activities. (1) ADMINISTRATION.—States shall use no more than 20 percent of funds allo-cated under section 104 for the costs of administration, including training, tech-

(II) "65 percent" for "60 percent", and (II) "75 percent" for "70 percent", and (III) "65 percent" for "60 percent",

(b) Local Application - Local applicants may use funds provided under this part

(2) SCHOOL VOLUNTEER AND PARTHERSHIP PROGRAMS.—States shall use no more than 10 percent of funds allocated under section 104 for school volunteer and

(2) PUBLIC AND PRIVATE NONPROFIT AGENCIES -- Public and private nonprofit (a) CLEARINGHOUSES ----(b) Lin OEXVEAL.---The Secretary, in consultation with the Mational Service Board authorized under title IV of this Act, is authorized to fund one or more national or regional clearinghouses on service. (2) Durator AND BRIVATE MOMBORT ACRUCES --- Public and private monprofit (3) Durator AND BRIVATE MOMBORT ACRUCES --- Public and private monprofit (3) Durator AND BRIVATE MOMBORT ACRUCES --- Public and private monprofit (3) Durator AND BRIVATE MOMBORT ACRUCES --- Public and private monprofit (3) Durator AND BRIVATE MOMBORT ACRUCES --- Public and private monprofit (3) Durator AND BRIVATE MOMBORT ACRUCES --- Public and private monprofit

thorized to make grants to fund national model youth service programs. (i) In CENERAL.-The Secretery, in consultation with the Board of Directors for the Corporation for Mational Service as established under section 403, is au-

-..etnaя() (d)

SEC 100 LEDERVE VCLIALLES

partnership programs.

Delenieve

Junds under this part.

--- 'SELVIC (8)

"80NAL 10 38A 301 TYS

graph (1) of this subsection.

52

Soverning appointments in the competitive service, technical employees to adminis-ter this tisk who may be paid without regard to the provisions of chapter 51, and subchapter III of chapter 53 of such title relating to classification and General Conduits number 10 and 10 an "(c) TECHNICAL ELEVENTER.-(1) The Secretary may appoint, for terms not to exceed 3 years, without regard to the provisions of title 5 of the United States Code

Schedule pay rates. (2) The Secretary may appoint no more than I technical employee for each (2) The Secretary may appoint no more than I technical employee for each (3) The Spectrated under section 1064. (6) The provisions of section 1064. (10) The Provisions of section 1064. (10) The Provisions of section 1064. (10) The Spectrated under section 1064.

2286 8143 (e) Nothing in this part shall be construed as requiring an institution of higher education to offer academic credit as a requirement of receiving maintaince under

"(ii) obtain partial cancellation of the student loan for service under the Presce Corps Act (as established by the Presce Corps Act of 1973 (42 \times 10.5.C. 4951 et seq.)" or under the Domestic Volunteer Service Act of 1973 (42 U.S.C. 4951 et seq.)"

of this title, or both, may "(i) obtain deferring for the repryment of the principal and interest for "(i) obtain deferring for our for the Poste Corps Act of 193 (42 U.S.C. 260) at seq.), or for comparable by the Poster Service Act of 193 (42 U.S.C. 495) at seq.), or for comparable full-time service as a volumeer for a tex-exempt organization of demonstrated effective-reast.

(a) Section SK(aX)) of the Higher Education Act of 1966 (20 U.S.C. 1092(a)(X1)) (a) Section SK(aX) of the Higher Education Act of 1966 (20 U.S.C. 1092(a)(X1)) (b) the settle referred to as the "Act") is amended— (1) by striking out the period at the end of subparagraph (J) (2) by striking out the period file following: (3) by striking out the period file solitowing: (4) by striking out the period file solitowing: (5) by striking or the period file solitowing: (5) by striking or the period file solitowing: (6) by striking or the period file solitowing: (7) the terms and conditions under which students receiving guaranteed student loans under period file or direct student loans under part E student loans under part B of this title or direct student loans under part E student loans under part B of the still of the solitowing:

PART D-PUBLICATION

(a) Work Strupt Protection 4430.05) of the Higher Education Act of 1965 (a) Work Strupt Protection 4430.05) with the statistic year 1990-1991 and 1090-1991 and 1990-1992; (a) 1000-1991 and inserting "70 percent for academic year 1995-1993".

(B) The Servetry shall make the allotment required under supparagraph (A) on (B) The Servetary shall make the allotment required under supparagraph (A) on the basis of the number of students participating in campus-based community work learning study jobs assisted under this subpart in each State as compared to the four number of students participating in such jobs in all States.

Bection 4158(a) of the Higher Education Act of 1966 (20 U.S.C. 1070c-1(a)) is amended by inserting the oppropriation for this subpart exceeds \$75,00,000, the provisions of section 416(X) in this screed, the appropriation for this provisions of section 418(X)NSXAA, allot 50 percent of such excess to the States for the provisions of section 418(X)NSXAA, allot 50 percent of such excess to the States for the provisions of section 418(X)NSXAA, allot 50 percent of such excess to the States for the provisions of section 418(X)NSXAA, allot 50 percent of such excess to the States for the provisions of section 418(X)NSXAA, allot 50 percent of such excess to the States for the provisions of section 418(X)NSXAA, allot 50 percent of such excess to the States for the provisions of section 418(X)NSXAAA, allot 50 percent of such excess to the States for the provisions of section 418(X)NSXAA, allot 50 percent of such excess to the States for the provisions of section 418(X)NSXAA, allot 50 percent of such excess to the States for the provisions of section 418(X)NSXAA, allot 50 percent of such excess for the provisions of section 418(X)NSXAA, allot 50 percent of section 418(X)NSXAA, allot 50 percent of section 418(X)NSXAA, allot 50 percent of such excess for the section 418(X)NSXAA, allot 50 percent of section 418(X)NSXAA, allot 50 percent of section 418(X)NSXAA, allot 50 percent of section 50 percent of section 50 percent for the section 50 percent of section 50 percent for the section 50 percent

SEC. IN: VODILIONAL RESERVATION FOR CAMPULATERABED COMMUNITY WORK LEADING STUDY SWARDORY PROGRAMS PART C-STATE STUDENT INCENTIVE GRANT AND WORK

BUNEGUES NOT NOTANDOWN IN THE CASE

SEC. HE WORK STUDY PROGRAMS.

11RCCXPX3XBT

BURGE AND ADDRESS APPLIE

റ്റ

"Sac. 1964. There are authorized to be appropriated to carry out this part, \$35,000,000 for facat year 1991, and such sums as may be necessary for each of the 4 succeeding ways." BNOFFALETORTTA TO NOFFAXLEOHTUA"

SEC. HT. EXIT COUNSELING FOR BORROWERS.

"noitose dous to (J)(I)(a)

where the set of the conditions and such that the structure of the set of th

(1) by striking "and" at the end of paragraph (1). (2) by striking the period at the end of paragraph (2) and inserting in lieu (2) by striking the period at the end of paragraph (2) and inserting in lieu (2) by striking the period at the end of paragraph (1).

the period a comma and the following. "particularly the requirements of subsection

(b) Section 487(aN1) of the Act (20 U.S.C. 1094(aX1)) is smended by inserting before π

Section 486(b) of the Act (20 U.S.C. 1092(b)) is amended-

SEC 147. DEPARTMENT INFORMATION ON DEFERMENTS AND CANCELLATIONS.

"(v) in the case of service described in subparagraph (F) of paragraph (2) at the rate of 15 percent for the first or second year of such service and 20 percent of the third or fourth year of such service."

(b) Section 456(x83X,A) of the Act (20 U.S.C. 1087ee(aX3XA)) is amended— (b) Section 456(x83X,A) of the Act (20 U.S.C. 1087ee(aX3XA)) is amended— (1) by striking out 'or' at the end of clause (ii), (2) by striking and the word 'or', and thereof a semicolon and the word 'or', and (3) by adding at the end thereof the following new clause: (3) by adding at the end thereof the following new clause: (3) by adding at the end thereof the following new clause: (3) by adding at the case of service described in subnersereph (3) of narearent (3) by adding at the case of service described in subnersereph (3) of narearent (3) by adding at the case of service described in subnersereph (3) by adding at the case of service described in subnersereph (3) by adding at the case of service described in subnersereph (3) by adding at the case of service described in subnersereph (3) by adding at the case of service described in subnersereph (3) by adding at the case of service described in subnersereph (3) by adding at the case of service described in subnersereph (3) by adding at the case of service described in subnersereph (3) by adding at the case of service described in subnersereph (3) by adding at the case of service described in subnersereph (3) by adding at the case of service described in subnersereph (3) by adding at the case of service described in subnersereph (3) by adding at the case of service described in subnersereph (3) by adding at the case of service described in subnersereph (3) by adding at the case of service described in subnersereph (3) by adding at the case of service described in subnersereph (3) by adding at the case of service described in service des

(b) for each of the product in the product of the (a) Section 465(aX2) of the Higher Education Act of 1965 (20 U.S.C.1087ee(aX2))

OF HIGHER EDUCATION PART E-DIRECT LOANS TO STUDENTS IN INSTITUTIONS

ble full-time service as a volunteer for a tax exempt organization of demonstrat-

Thereof a semicolon and "said" and the set of paragraph (4) and monthly for send, and "said" (5) by adding the following new paragraph after paragraph (4). "(5) the exact amount of 1973 (42 U.S.C. 2501 et aqc.), for every current for service meetic Volunteer Service Act of 1973 (42 U.S.C. 4501 et aqc.), and for compara-meetic Volunteer Service Act of 1973 (42 U.S.C. 4501 et aqc.), and for compara-meetic Volunteer Service Act of 1973 (42 U.S.C. 4501 et aqc.), and for compara-meetic Volunteer Service as a volunteer for a tax-exemul purgital-ple full-time service as a volunteer for a tax-exemul purgital-tion of demonstrati-

"(F) as a full-time volunteer in service comparable to service referred to in subparagraph (E) for an organization of demonstrated effectiveness which is exempt from taxation under section 501(cX3) of the Internal Reve-nue Code 1986.

Section 486(d) of the Act (S0 U.S.C. 1093(d)) is amended by inserting the following Section 486(d) of the Act (S0 U.S.C. 1093(d)) is amended by inserting the following before the last full sentence: "The Secretskry shall provide information on the specif-before the last full sentence: "The Secretskry shall provide information on the specif-before the last full sentence: "The Secretskry shall provide information on the specif-before the last full sentence: "The Secretskry shall provide information on the specif-defer repayment fol leans for service under the Peace Corps Act (se setablished by 1993 (42) U.S.C. 4961 as earylos under the Peace Corps Act (se setablished by 1993 (42) U.S.C. 4961 as earylos under the Peace Corps Act (se setablished by 1993 (42) U.S.C. 4961 as earylos under the Peace Corps Act (se setablished by 1993 (42) U.S.C. 4961 as earylos under the Peace Corps Act (se setablished by 1993 (42) U.S.C. 4961 as earylos under the Peace Corps Act (se setablished by the Peace Corps Act (S2 U.S.C. 5501 et seq.) as the organism as a volumeter the Peace Corps Act (S2 U.S.C. 5501 et seq.) as the organism as a volumeter the Peace Corps Act (S2 U.S.C. 5501 et seq.) as the organism as a volumeter the Peace Corps Act (S2 U.S.C. 5501 et seq.) as the organism as a volumeter the Peace Corps Act (S2 U.S.C. 5501 et seq.) as the organism as a volumeter the Peace Corps Act (S2 U.S.C. 5501 et seq.) as the organism as a volumeter the Peace Corps Act (S2 U.S.C. 5501 et seq.) as the organism as a volumeter the Peace Corps Act (S2 U.S.C. 5501 et seq.) as the organism as a volumeter the Peace Corps Act (S2 U.S.C. 5501 et seq.) as the organism as a volumeter the Peace Corps Act (S2 U.S.C. 5501 et seq.) as the organism as a volumeter the Peace Corps Act (S2 U.S.C. 5501 et seq.) as the organism as a volumeter the Peace Corps Act (S2 U.S.C. 5501 et seq.) as the organism as a volumeter the Peace Corps Act (S2 U.S.C. 5501 et seq.) as the organism as a volumeter the Peace Corps Act (S2 U.S.C. 5501 et seq.) as a volumeter the P cancellations or delerments when they serve as a paid employee of a tax exempt or-

SEC. HI. FORM CONCEPTIVION VOLHOBIZED.

38C 14F DVLV ON DELEMBRILS VID CVICETTVLLORS

Section 485B(a) of the Act (20 U.S.C. 1092b(a)) is amended-

(1) by striking "and" at the end of paragraph (3); (2) by striking the period at the ond of paragraph (4) and inserting in Heu

The amendments made by sections 2 and 3 of this part shall apply only to loans made to cover the costs of instruction for periods of enrollment beginning on or after 30 days after the date of enactment of this part to individuals who are new after an art of this date.

borrowers on that date.

EVEL E-FOVN FORGIVENESS

SEC. HI. LOAN FORCIVENESS.

couply with the agreement.

the following new subsection:

exempt organization of demonstrated effectiveness.

--- Jarli 910866

SEC: 107 EMECLIAR DVLE

be paid by the Secretary. (C) Nothing in this subsection shall be construed to suthorize refunding of any ".neol odi no insurvager

(%) If a portion of the loan is paid by the Secretary under this subsection for any vear, the entire amount of interest on such loan which accrues for such year shall be needed.

13

on that date.

The amendments made by section 161 of this part shall apply only to loans made to over the costs of instruction for periods of enrollment beginning on or after 30 days after the date of enactment of this Act to individuals who are new portowers day that the

TITLE II-YOUTH SERVICE CORPS

TISDASI WEID dividual who is responsible for supervising a crew of participants, including the

the supervision of other participants engaged in work projects pursuant to this (2) CREW LEADER .-- The term "crew leader" means a participant who assists in

(I) BOARD -- The term "Board" means the Mational Service Board established

a ni bellorne laubivibni na ansem "inaqisitraq" mrei edT--.ruAustraAq (T)

OF CFW with a specific project. (9) PROCRAM.-The term "program" means an activity carried out under this program funded under this litle. (8) PLACEMENT -1 he term "placement" means the matching of a participant

(10) PROGRAM AGENCY.—The term "program agency" means-(M) a folderal or State agency designated to manage a program in a State; (B) the gevening body of an Indian tribe; or (C) a local another administration a program hinder under this title.

(1) Provent event even of an interant error, or (C) a local applicant administering a program funded under this title. (1) Provect.—The term "project" means an activity that results in a specific identifiable services or product that otherwise would not be done with existing funds, and that does not duplicate the routine services or functions of the em-prover to whom participants are assimted.

Гилая, ала Unat осезя пот адпризает пра тошило вегугова от ципскола от цле ет-роучет бо м йоли ратборалћа вте авацитед (12) Рупяцс цамис—Тће term "public lands" means any lands от waters (or (12) Рупяцс цамис—Тће term "public lands" means any lands от waters (or interest therein) wored or a State or local government. (13) Sълтк—The term "State" means each of the several States, the District (13) Sълтк—The term "State" means each of the Northern Mariana lalands, or Palau American Samos, the Commonwealth of the Northern Mariana lalands, or Palau (14) Suntar reconsea, "The term "suntant program fands, or Palau (14) Suntar reconstant.—The term "suntant or the Marianal lalands, or Palau (14) Suntar reconstant.—The term "suntant or the Marianal lalands, or Palau (14) Suntar reconstant.—The term "suntant or the Marianal lalands, or Palau (14) Suntar reconstant.—The term "suntant program" means a program au-thorized under this title that is limited to the months of June, July, and Mugust.

gram" means a program, such as a conservation corps or human services corps, (15) YOUTH SERVICE CORPS PROGRAM.-The term "youth service corps pro-

AI offici appun

-aliti aidt ni beeu aA SNOLLINIARD TOT DESLIVITIONS.

"(3) The Secretary shall in each fiscal year pay to the holder of each loan for which the Secretary assumes responsibility under this subsection the amount specifical in paragraph (5).

with the agreement will assume the obligation of paying the amount of any such loan attributable to the period for which the student borrower failed to

(2) The agreement under paragraph (1) shall contain provisions designed to

U.S.C. 4961 et see,) or for comparable full-time service as a volunteer with a tex-(22 U.S.C. 2501 et seq) or under the Domestic Volunteer Service Act of 1973 (42

 $\langle 0 \rangle$ is the second second

comparable full-time service as a volunteer with a tax-exempt organization of demonstrated effectiveness for the payment by the United States of the percent of the amount of loans specified in section 432()."

(B) structing our trap period as the end of study structing our the period as the order of structing our the period as the end of the repeating new subparagraph. (C) adding at the end thereof the following new subparagraph: "(V) provides duplet to section 432(j)) in the case of any student borrower who, prior to the beginning of the repeatiment period, agrees in writing to serve who, prior to the beginning of the repeatiment period, agrees in writing to serve who, provides duplet to section 432(j) in the case of any student borrower "(V) provides duplet to section 432(j) in the case of any student borrower who, prior to the beginning of the repeatement period, agrees in writing to serve who, prior to the bounder the fease Corps Act (22 U.S.C, 4561 et seq.) or under the Domestic Volunteer Service Act of 1973 (42 U.S.C, 4561 et seq.) or under the Domestic Volunteer Service Act of 1973 (42 U.S.C, 4561 et seq.) or under the Domestic Volunteer Service Act of 1973 (42 U.S.C, 4561 et seq.) or under the Domestic Volunteer Service Act of 1973 (42 U.S.C, 4561 et seq.) or under the Domestic Volunteer Service Act of 1973 (42 U.S.C, 4561 et seq.) or under the Domestic Volunteer Service Act of 1973 (42 U.S.C, 4561 et seq.) or under the Domestic Volunteer Service Act of 1973 (42 U.S.C, 4561 et seq.) or under the Domestic Volunteer Service Act of 1973 (42 U.S.C, 4561 et seq.) or under the Domestic Volunteer Service Act of 1973 (42 U.S.C, 4561 et seq.) or under the Domestic Volunteer Service Act of 1973 (42 U.S.C, 4561 et seq.) or under the Domestic Volunteer Service Act of 1973 (42 U.S.C, 4561 et seq.) or under the Domestic Volunteer Service Act of 1973 (42 U.S.C, 4561 et seq.) or under the Domestic Volunteer Service Act of 1970 et seq.) or under the Domestic Volunteer Service Act of 1970 et seq.) or under the Domestic Volunteer Service Act of 1970 et seq.) or under the Domestic Volunteer Service Act of 1970 et seq.) or under the Domestic Volunteer Service Act of 1970 et seq.) or under the Domestic Volunteer Service Act of

Limberts volumests volumest are take (as 0.50, start et seq.) or for compara-ble full-time service as a volumest by the United States of the percent of the strative effectiveness for the part by the United States of the percent of the amount of four septibility and section 432(1), and". (N1) Section 428(N1)(D) of the Act (20 U.S.C. 1978(N1)(D)) is amended by insert-ing after "paragraph" the following, "and subject to subparagraph (V)". (S) Section 428(N1) of the Act (20 U.S.C. 1978(N1)(D)) is amended by insert-ing after "paragraph" the following, "and subject to subparagraph (V)". (S) Section 428(N1) of the Act (20 U.S.C. 1978(N1)) is amended by insert-(S) Section 428(N1) of the Act (20 U.S.C. 1978(N1)) is amended by --(S) Section 428(N1) and "at the end of subparagraph (V)". (B) straignest up the period at the end of subparagraph (U) (B) straignest and "and" at the end of subparagraph (U) (B) straignest the period at the end of subparagraph (U) (B) straignest the period at the end of subparagraph (U) (B) straignest the period at the end of subparagraph (U) (B) straignest the period at the end of subparagraph (U) (B) straignest the period at the end of subparagraph (U) (B) straignest the period at the end of subparagraph (U) (B) straignest therefore and "and"; and

who, prior to the beginning of the repayment period, agrees in writing to volum-teer to marrice under the Peece Corps Act (22 U.S.C. 2501 et seq.) or under the Domestic Volunteer Service Act of 1973 (42 U.S.C. 451 et seq.) or toor compara-tion and the service market and the service a

(B) redesignating subparagraph (H) as subparagraph (I); and (C) insering after subparagraph (G) the following new subparagraph (H): "(H) provide subject to section \$23(b) in the case of any student horrower horrower to the horrowing of the representation of any student horrower

(B) the student borrower who fails to volunteer for service in accordance "(A) the Secretary will assume the obligation of paying the percent of any loan made, insurated, or guaranteed under this part pursuant to the schedule de-scribed in persgraph (5); and sufficient forenease who fails to volustees for service in accordance

(4) The Serveiry paragraph (a). (4) The Serveiry paragraph (a). (4) The Serveiry below to which the United States is entitled under paragraph any, or any part of, the loan to which the United States is entitled under paragraph (SKA) The percent of a loan which shall be paid by the United States under para-"(SKA) The percent of a loan which shall be paid by the United States under para-"(SKA) The percent of a loan which shall be paid by the United States under para-"(SKA) The percent of a loan which shall be paid by the United States under para-"(SKA) The percent of a loan which shall be percent for the first or second year of service graph (SXA) of this subsection is 15 percent for the first or second year of service and 20 percent for the third or fourth year of service described in paragraph (1).

40

that gives participants a mix of work experience, basic and life skills, education, that offers full-time, productive work (to be financed through stipenda) with visible community benefits in a natural resource or human service setting and

The rectange exception comments of the exception of an action man action man action man action and the recently left (neter care; (4) a list of requirements to be imposed on such approaching organization of an individual serving in a program or project under this title, including a require-individual serving in a program or project under this title, including a require-individual serving in a program or project under this title, including the require-ing organization that does not make such an investinant; (5) a description of the manner of appointment of sufficient supervisory staff (5) a description of the manner of appointment of sufficient supervisory staff (including participants who have displayed exceptional leadership qualities), (including partities),

There receive a contraction of the approximation of the strong of the received from their time programs or work experience obtained under this strong and crew leaders necessary for the proposed projects, the length of time that the services of such participants and crew leaders will be required, the support services that will be required for such participants and crew leaders, and a plan for recruiting such participants, and the support services of such participants, and the proposed projects, the length of time that the services of such participants, and the support services of such participants, and the support services that will be required for such participants, and the such participants, and crew leaders, and a plan for recruiting such participants, and the support services of such participants, and the support services of such participants, and the support services that will be required for the support services of such participants, and the support services of such participants, and the support services that will be required for the support services of such participants, and the support services of such participants, and the support services of such participants, and the support services of such as the support services of se

(2) a plan for certification of the training skills acquired by participants and award of academic credit to participants for competencies developed from train-

(1) a comprehensive description of the objectives and performance goals for the program, a plan for manging and funding the program, and a description of the types of the types of projects to be carried out, including a description of the types and duration of training and work expensions to be provided by such and any contrastion for earlier and the training and

(1) any Youth Service Corps program proposed to be conducted directly by auch State with funds provided under section 319, and (3) any grant program to entities within auch State proposed to be conducted by such State with innde provided under section 319, (2) any section 319, and any section 319, any section 319, and any section 319, and any section 319, and any

(b) CONTENT OF APPLICATION FOR A STATE.-In such application, the State shall de-

(a) SUBMERSION OF AFFLICATION.---In order to receive a grant under section 219, a Sieke shall aubmit an application to the Board at such time and in such manner as

made available to a State under section 219 may be used for summer Youth

funds made available to a program agancy under section 213 may be used for

(2) ADMINISTRATIVE EXPENSES - Not more than 15 percent of the amount of

(b) Distance Distance. (c) Distance State does not apply for assistance under this title, the Board may make a grant directly to a local apply the criteria described in section 305 (2) Eventuarrow. The Board shall apply the criteria described in section 305

(a) Comparized applications under section 205.

(c) RENEWING CRENT.-The Board may renew such grant for an additional three-

(d) TERM OF CENTRATE form of such grant shall be for a period not greater

(3) SUMMAR PROGRAMS-Not more than 10 percent of the amount of funds

(c) Contraint or Application for a State of Local Application for the application for the pursuant to section 219 to directly conduct a Youth Service Corps program pursuant to section 204 (a) or (b), each applicant shall include in the application for such that the section for the sect

aditoe

iO

ñ

the Board may reasonably require

Service Corps programs.

sommetretive expenses.

BUNDA AO NOLLVOOTTV 'NE 'DBB

AICS COLDS DLOBLEUN

in evaluating such local applicant.

"MORE THE REAL PROPERTY OF THE

-'NOLLVLIMPT (3)

APPL return

เบชน เบเอง โดยเม

low-income housing, energy conservation (including solar energy tech-inquesi, removal of architectural barriera to access by handicapped individ-

cilities, neighborhood improvements, literacy training that benefits educa-cilities, neighborhood improvements, weatherization of and basic repairs to (it) activities that focus on the rehabilitation or improvement of public fa-(D) private nonprofit organizations that primarily focus on social service;

(K) reclamation and improvement of strip-mined land; and (K) reclamation and improvement of strip-mined land; and (L) forestry, nursery, and cultural operations; or (L) forestry, nursery, and cultural operations; or (B) nursing homes, proprices, senior centers, hospitals, local libraries, parks, recreational facilities, day corrects, programs service individuals parks, recreational facilities, day corrects, programs service individuals with bandicanned conditions and achools;

(J) energy conservation projects, renewable resource enhancement, and

(c) fish culture, habitat maintenance and improvement, and other fishery B) urban revitalization, and historical and cultural site preservation; (1) in the case of conservation corps programs, focus on-(A) conservation, reinabilitation, and improvement of wildlife habitat, rangelands, parks, and recreational areas; (R) under extrained and reinarical and utilities areas;

(i) the improvement of abandoned railroad beds and rights-of-way; (H) insect, disease, rodent, and fire prevention and control;

(a) In GRURAL-Programs tunded under this title may carry out projects that-

ensure that program agencies consult with each local labor organization repre-

the credit to each participant for competencies developed from training pro-

(5) the State will encourage cooperation among programs and the appropriate (5) the State will encourage conneil established under the Job Training continuing council established under the Job Training Act (29 U.S.C. 1501 et seq.); Pattership Act (29 U.S.C. 1501 et seq.);

focal applicants will be evaluated;
 acrive programs in the State will be coordinated;
 acrive programs will be recruited;
 focumically and educationally disadvantaged youth, including youth who have left focater, will be recruited;
 programs will be evaluated;

(d) Charar Processar -- Each State desiring to receive funds under this title to con-duct a grant program pursuant to subsection (bX2) shall describe in its application

(7) a description of the facilities, quarters, and board (in the case of reasitor, the assorbion of the facilities, quarters, and board (in the case of reasitor, its facilities, limited and emergency medical care, transportation from adminituatives, limites to work attes, and other applications, and escription of basic standards of work requirements, health, mutition, (9) a description of basic standards of work requirements, health, mutition, (9) a description of the pine standards of work requirements, health, mutition, (9) a description of the pine standards of work requirements, health, mutition, (9) a description of the pine standards and pool hours of the protection as the moment that with any local labor organization or presents as a reast of the protection as the assorbable and processing to the carried out by such program, and (11) the earne of the track who are engreded in the earne or similar work as each through on the program, and (11) the protection as the Board may presenting and (11) the protection as the Board may presenting and (11) the earne of the program, and (11) the protection as the Board may presenting to the earne or similar work as an entiting propesed to be carried or using a submittee and the program and (11) the track point of the program and (11) the track point of the program and (11) the track point of the program and (11) the track and (11) the track point of the program and (11) the track point of the program and (11) the track and (

(i) a description of the facilities, quarters, and board (in the case of residen-

crew structure and a youch development component,

bus sidesplayed of a plan to ensure the on-site presence of how plant a lo notidiroseb a (8)

who shall in turn provide for other central elements of a youth corps, such as

31

(I) before any placement of a volunteer under this Act is made, the State will

(6) such State will certify the training skills acquired by each participant and

(D) road and trail maintenance and improvement; (D) road and trail and storm damage assistance and controls; (E) sustoin, lake, waterfront harbor, and port improvement; (P) anitant serioriton and pollution control:

with handicapped conditions, and schools;

(0) wellands protection and pollution control;

the work that is proposed to be carried out by such program.

recovery of biomage;

.

SEC. M. LOCES OF PROCESSE.

running, and support services.

(a) IN CENERAL.---The Board is authorized, in accordance with this tille, to provide a grant to a State or local applicant to create or expand full-time or summer youth SEC 201 LIGGERY WATHORNED.

91

uals to public facilities, and conservation, maintenance, or restoration of natural resources on publicly held lands; and (F) any other nonpartisan civic activities and services that the Board de-termines to be of a substantial social board in the instant. Au-cational, or environmental modes (particularly needs related to poverty) or in the community where volunteer service is to be performed. (b) INELGERE SERVICE CATECORES.-In order to be eligible to receive assistance under this title, the service projects referred to in subsection (a) shall not be con-ducted by any--

business organized for profit;

(2) labor union
(3) partisan political organization;
(4) organization ergaged in religious activities, unless such project does not involve any religious functions; or
(5) domestic or personal service company or organization.
(c) Linkarrow on Servicz.—No participant shall perform services in any project for more than a 1-year period.

SEC 207. RELATED PROCRAMS.

An activity administered under the authority of the Secretary of Health and Muman Services, that is operated for the same purpose as a program eligible to be enried out under this title, is encouraged to use services available under this title.

SEC. 208. PUBLIC LANDS OR INDIAN LANDS.

(a) LIMITATION.—The Board shall only fund programs that involve projects on public lands or indian lands or that provide a public benefit. (b) REVIEW or APPLICATIONS.—The Board shall consult with the Department of the Interior in reviewing applications proposing programs or projects on public lands or Indian lands.

(c) CONSISTERNCY.---A project carried out under this title for conservation, rehabili-tation, or improvement of any public lands or Indian lands shall be consistent with

(1) the provisions of law and policies relating to the management and administration of each lands, and all other applicable provisions of law, and (2 all management, operational, and other plans and documents that govern (2 all management, operational, and other plans and documents that govern (3 REEXEVATION OF FUNES.—The Board shall reserve not more than 5 percent of funds appropriated under section 219 for Federal diasstar relief program.

SEC. 209. TRAINING AND EDUCATION SERVICES.

(a) ASSESSMENT OF SKILLS.—Each program agency shall assess the educational records of participants at the time of entranoo into the program, using any available records or simplified as assessment, means or methodology. (b) ENHANCEMENT OF SKILLS.—Each program agency shall, through programs and in the program agency shall, through programs and in the program. (c) PROVISION OF IN-SERVICE TRAINING AND EDUCATION.— (1) REQUIREMENT—A program agency ball use not less than 10 percent of the funds made svaliable to such agency to provide in-service training and edu-ctional materials and services for participants and individuals serving in a continuel materials.

project. (2) ADREMENTS FOR ACADEMIC STUDY.—A program agency that is receiving au-isistance under this Act may enter into arrangements with academic institutions or education providers, including— (A) local education agencies; (B) community colleges; (B)

ପରିଜ୍ଞ

4-year colleges: area vocational-technical schools; and community based organizations;

for academic study by a participant in order for such participant or individual to upgrade literary skills, to obtain a high school diploma or the equivalent of such diploma, to obtain a college degree, or to enhance employable skills. (3) Couvestavo.—Career and educational guidance and counseling shall be provided to a participant during a period of invervice training as described in this agneetion.

(4) PRIMERY FOR PARTICIPANTS WITHOUT HIGH SCHOOL DIFLOMAS.—A participant who has not obtained a high school diploma or the equivalent of such diploma shall have priority to recoive services under this subsection.

(d) Post-Service Education and Thaining Assistance...

(1) Consuments and a statistical statistical state states with states with states with states with states with states and local officials shall carefity that standards and proceedures with respect to and local officials shall carefity that standards and proceedures with the respect to the awarding of scademic credit and the certification of educational attainment in programs conducted under subsection (c) are considered with the requirements of applicable State and local subsection (c) are considered with the requirements of applicable State and local law and regulations. (2) Accurate standards and procedures shall specify that an individual serving in a program or project under this title. (A) who is not a high school graduate shall participate in an educational curriculum so that such individual can earn an high school diploma or the Guivalent of such individual can earn an angle of applicants of actual program or the Guivalent of such individual can earn an angle school diploma or the Guivalent of such individual can earn an angle school diploma or the Guivalent of such individual can earned in a successional curriculum so that such individual can earne and a high school diploma.

SEC. 219. AMOUNT OF AWARD.

In determining the amount of funds to be awarded to an applicant under this title, the Board shall consider— (1) the proportion of the unemployed youth population of the area to be served; and (2) the type of project or service proposed to be carried out with funds appropriated under this title.

SEC. 211. MATCHING REQUIREMENT.

(a) FEDERAL SHARE.—The Federal share of the cost of programs assisted under this tile shall not exceed 75 protent. (b) STATE on LOCAL APPLICANT.—The State or local applicant share of the costs of programs assisted under this title shall be at least 25 percent.

SEC. 212, PREFERENCE FOR CERTAIN PROJECTS.

In the approval of applications for programs and projects submitted under section 205, the Board shall give preference to programs and projects that----(1) will provide long-term benefits to the public; (2) will instill a work ethic and a sense of public; (3) will be labor intensive, and involve youth operating in crews; (4) can be planned and initiated promptly; and (5) will enhance akills development and educational level and opportunities for the participants.

SEC. 213. EFFECT OF ALLOWANCES ON ELIGIBILITY FOR OTHER FEDERAL ASSISTANCE.

Living allowances and benefits received under this title by an economically disad-variaged youth, as defined in section 480 of the lob Training Fartnership Act (28 USC: 1508(9), shall be disregarded in determining the eligibility of the family of the youth for, and the amount of, any Federal benefits based on need.

SEC. 215. POST-SERVICE BENEFITS.

(a) IN GENERAL.—The program genery shall provide post-service education and training benefits (such as scholarships and grants) for each participant in an amount not best start 550 per week nor more than \$100 per week.

56

(b) EXCLUSION FROM GROSS INCOME.-For purposes of section 61 of the Internal Revenue Code of 1986, in the case of an individual, gross income shall not include any amount received as assistance under this section.

SEC. 216. LEVING ALLOWANCE.

(a) IN GENERAL.—Each participant in a full-time youth service corps program shall receive a living allowance of not less than 100 percent of the poverty line for a single individual (as defined in section 673(2) of the Community Services Block Grant Act (42 U.S.C. 9902(2))) and not more than 100 percent of the amount such participant would have earned if such participant had been paid at a rate equal to the minimum wage for a 40-hour work week under section 6(a)(1) of the Fair Labor Standards Act of 1938 (29 U.S.C. 206(a)(1)) during the period of service of such par-ticipant.

(b) HEALTH INSURANCE.-In addition to the living allowance provided under subsection (a), each participant in a full-time youth service corps shall be provided with health insurance

(c) FACILITIES, SERVICES, AND SUPPLIES .-

(1) IN GENERAL .- The program agency may deduct, from amounts determined under subsections (a) and (b), a reasonable portion of the costs of the rates for room and board provided at residential facilities for a participat. (2) EVALUATION.—The program agency shall establish the deductions and rates under paragraph (1) after evaluating of costs of providing such room and

board.

(3) DUTIES OF PROGRAM AGENCY.—A program agency may provide facilities, quarters, and board and shall provide limited and emergency medical care, health insurance, transportation from administrative facilities to work sites, and other appropriate services, supplies, and equipment to each participant. (d) GUIDANCE AND PLACEMENT.-

(1) IN GENERAL —Each program agency shall provide such job and educational guidance and placement information and assistance for each participant as may be necessary.

(2) COORDINATION WITH OTHER ENTITITES .- Such assistance shall be provided in coordination with appropriate State, local, and private agencies and organizations.

SEC. 117. JOINT PROJECTS INVOLVING THE DEPARTMENT OF LABOR.

(a) DEVELOPMENT .--- The Board may develop, in cooperation with the Secretary of Labor, regulations designed to allow, where appropriate, joint projects in which ac-tivities supported by funds authorized under section 219 are coordinated with activities supported by funds authorized under employment and training statutes admin-istered by the Department of Labor (including the Job Training Partnership Act (29 U.S.C. 1501 et seq.)).

(b) STANDARDS.—Regulations promulgated under paragraph (1) shall provide standards for approval of joint projects that meet both the purposes of this title and the purposes of such employment and training statutes under which funds are available to support such projects.

SEC. 218. FEDERAL AND STATE EMPLOYEE STATUS.

(a) IN GENERAL -- Participants and crew leaders are considered as being responsible to, or the responsibility of, the program agency administering the project on which such participants, crew leaders, and volunteers work. (b) NON-FEDERAL EMPLOYEES -

(1) IN OENERAL -- Except as otherwise specifically provided in this subsection, a participant or crew leader in a project that receives assistance under this title shall not be considered a Federal employee and shall not be subject to the provisions of law relating to Federal employment.

(2) WORK-RELATED INJUKY. — For purposes of subchapter 1 of chapter 31 of title 5. United States Code, relating to the compensation of Federal employees for 5. United States Code, relating to the compensation of Federal employees for work injuries, a participant or crew leader serving in a program that receives assistance under this title shall be considered an employee of the United States within the meaning of the term "employee" as defined in section 8101 of title 5. United States Code, and the provision of that subchapter shall apply, except—(A) the term "employee of during" as used in such subchapter, shall apply, except—(A) the term "employee of during" as used in such subchapter, shall apply, except—(A) the term "employee of during" as used in such subchapter, shall apply, except—(A) the term "employee of during".

(A) the term "performance of duty", as used in such subchapter, shall apply, except-(A) the term "performance of duty", as used in such subchapter, shall not include an act of a participant or crew leader while absent from the as signed post of duty of such participant or crew leader, except while partici-pating in an activity authorized by or under the direction and supervision for the direction of the direction and supervision of a program agency (including an activity while on pass or during travel to or from such post of duty); and

(B) compensation for disability shall not begin to accrue until the day fol-lowing the date that the employment of the injured participant or crew leader is terminated.

(3) TORT CLAIMS PROCEDURE .- For purposes of chapter 171 of title 28, United States Code, relating to tort claims procedure, a participant or crew leaders assigned to a Youth Service Corps project shall be considered an employee of the United States within the meaning of the term "employee of the government" as defined in section 2671 of such title.

(4) ALLOWANCE FOR QUARTERS .- For purposes of section 5911 of title 5, United States Code, relating to allowances for quarters, a participant or crew leader shall be considered an employee of the United States within the meaning of the term "employee" as defined in paragraph (3) of subsection (a) of that section.

SEC. 219. AUTHORIZATION OF APPROPRIATIONS.

There are authorized to carry out this title \$100,000,000 for fiscal year 1991, and such sums as may be necessary for each of the fiscal years 1992, 1993, 1994, and 1995

TITLE III—NATIONAL SERVICE DEMONSTRATION PROGRAM

SEC. 301. SHORT TITLE

This title may be cited as the "National and Community Service Demonstration Act"

SEC. 302. DEFINITIONS

As used in this title-

(1) BOARD .- "The term "Board" means the National Service Board authorized under title IV

(2) ELIGIBLE ORGANIZATION.-The term "eligible organization" means a public or private nonprofit organization engaged in meeting human, educational, environmental, or public safety needs.

(3) INSTITUTION OF HIGHER EDUCATION .--- The term "institution of higher education" has the same meaning given that term in section 1201(a) of the Higher Education Act of 1965 (20 U.S.C. 1141(a)).

(4) PARTICIPANT .-- The term "participant" means an individual participating in a program under this title.

(5) PLACEMENT .-- The term "placement" means the matching of a participant or team with a specific project.

(6) PROGRAM .- "The term "program" means an activity carried out under this title

(7) SPECIAL SENIOR SERVICE MEMBER.-The term "special senior service member" means an individual who is age 60 or over and willing to work fulltime or part-time in conjunction with a full-time national service program.

(8) SPONSORING ORGANIZATION .-- The term "sponsoring organization" means an organization, eligible to receive assistance under this title, that has been se-(9) STATE -- The term "State" means a State, the Commonwealth of Puerto

Rico, Guam, the District of Columbia, American Samoa, the Virgin Islands, the Commonwealth of the Northern Mariana Islands, the Republic of the Marshall Islands, the Federated States of Micronesia, or Palau.

SEC. 303. CRANTS

(a) IN GENERAL-Not later than 1 year from the date of enactment of this title, the Board shall, in accordance with the provisions of this title, make grants to

States to create full-time and part-time national service demonstration programs. (b) TERM OF GRANT .- The term of such grant shall not extend beyond September 30, 1995.

(c) CRITERIA FOR RECEIVING APPLICATIONS .--- In awarding such grant to a State, the

Board shall consider

(1) the ability of the proposed program of such State to serve as an effective model for a large-scale national service program;

(2) the quality of the application of such State, including the plan of such State for training, recruitment, placement, and data collection;

(3) the extent that the proposed program builds on existing programs; and

DUB :#1# UON

fully cooperate with any evaluation undertaken by the Board pursuant to sec-(13) a plan for evaluating the program and assurances that such State will

(11) the State budget for the program; (12) whether the State desires to exercise the voucher alternative option au-(12) whether the State desires to exercise the voucher that such State will thorized inder section States the program and assurances that such State will

BUOU

(10) procedures for any program; (2) a plan for recruiting and selecting sponsoring organizations that will re-ceive participants under this title; (10) procedures for matching such participants with such sponsoring organiza-(10) procedures for matching such participants with such aponsoring organiza-

(i) assurances that, before such placement is made, such Btate will consult with employees at the proposed project site who are engaged in the same or (a) the anticipated number of full-time and part-time participants and special antice semiconstruction members in our bar.

:เมษาสักเส

with any local labor organization representing employees in the srea who are engaged in the same or similar work as that proposed to be carried out by such (6) assurances that, before such placement is made, such State will consult (6) assurances in the area who are

(5) the plan for placing such participants in teams or making individual place-

(4) the geographical areas within such State in which the program would be openated to provide the optimum match between the need for volunteer services and the anticipated supply of volunteers; min Soud

(2) the method that participante, including economically and educationally disadvantaged youth, college-bound youth, and samployed individuals, shall be (3) procedures for training, supervising, and organizing participants in such (3) procedures for training, supervising, and organizing participants in such movement.

suant to a grant or contract;

(h) SAATE ADRUCTION FOR URANT.—In order to receive a grant under this title, a (h) SAATE ADRUCATION FOR URANT.—In order to receive a grant under the manner as Use Board and the Board and the User and private mappication to the Board at such the Board at the the User the application application applications if (h) the State administrative plan for the program, including auch functions, if (h) the State administrative plan for the program, including auch functions, if any that will be actived out by public and private nonprofit organizations puramet to constract. sind State

(3) COPREATIVE AREANORMENT.—For purposes of this subsection, a State oper-eding a national service program involving a cooperative arrangement with a multi-State organization or with sites in more than one State shall be counted and a minute State. part-time program.

(2) Studik recordent. For purposes of this subsection, a State operating a single national service program with both full-time and part-time options and be counted as a State operating a full-time program and a State operating a neutrine marchine.

In each of the flacel years 1994 and 1995.

(1) no less than it States are authorized to operate full-time programs and no more than 35 States are authorized to operate pert-time programs in thesi year 1998; and

(C) no more than 10 States are suthorized to operate full-time programs and no more than 10 States are suthorized to operate part-time programs

(ii) no more than eight States are authorized to operate full-time pro-grams and no more than eight States are authorized to operate part-time programs in facal year 1992;

in fiscal year 1991;

58

standiniting and (f) Convocations or Processas --The Board shall ensure that at least 25 percent of programs under section 314 include full-time, part-time and special senior serv-

ized under section 307(dX3).

(d) Dryggerre.—The Board shall ensure that programs receiving such a grant are elegraphically diverse and include programs in both urban and rural States.
 (e) ALTREMATTE VOUCHER OFTION LIMITED.—The Board shall ensure that no more (b) ALTREMATTE VOUCHER PROGrams and include programs in both urban and and and and and the more that and an anticent of States are authorized to exercise the alternative voucher authorized and an anticenter the anticenter the authorized and an anticenter the authorized an

THUOTIN

(4) the expediency with which the State proposes to make the program oper-

50

.enuper vidences are the Board may reasonably require.

form national service to meet unmet educational, human, environmental, and public (a) IN CENERAL -A participant in a program funded under section 314 ahall per-

types of service, such as illeracy programs, Head Start (as setablished (1) educational service, such as illeracy programs, Head Start (as setablished assistance, and service in schools, illeraries, and adult education programs, tutorial (2) human service, such as-

(b) Trate or MATIONAL SERVICE.—Such national activice may include the following

SEC 194 LIVES OF MATIONAL SERVICE.

7990 10 FI 988 8i (I)

-isubivibri dous il margore sivise lancit

(B) an average of 9 hours per week.

the service obligation completed by such individual

.eonebien

- INIT-THAY (6)

ALTISION SOF EPICIEITLA POBLCI.

-JOI BALOS

(c) TERMS OF SERVICE.--

- MOLANNE 40 HLONNY (8) SEC WE LESHE OF SERVICE

(2) human service, such as-

(b) Full-time - An individual may serve in a full-time national service program if

(2) PRIORITY.-In selecting applicants for a part-time program, States shall

(I) ELIOIBILITY REQUIREMENTS -An individual may serve in a part-time na-

(3) SPECIAL SERVICE — A special senior service participant performing and one service single service single service single service single service service single service servi (2) FULLTIME - A perficipant performing full-time national service shall serve for not less than 40 hours per week.

pleting a torm of service in the program for compelling personal circumstances abown by such participant, the Board may provide such participant with a portion of the financial assistance apecified in section 301 corresponding to the quantity of the financial assistance apecified in section 301 corresponding to the quantity of

before the period of activity service in the second state activity for activity and activity service individual service participant from com-national service shall serve for a period of time as allowed by the Board (b) Partiat. Computing service to the State releases a participant from com-nectional service in the provem for compaling the period from the service participant from com-leting a form of service in the provem for compaling the period from the service participant from com-leting a form of service in the provem for compaling the period from the service participant from com-tained a service in the provem for compaling the service participant from com-tained and service in the provem for compaling the service participant from com-tained and service in the provem for compaling the service participant from com-tained and service in the provem for the service participant from com-tained and service in the provem for compaling the service participant from com-tained and service in the provem for compaling the service participant from com-tained and service in the provem for compaling the service participant from com-tained and service in the provem for the service participant from com-tained and service in the service service and service participant from com-tained and service in the service service and service service and service service and service service and service ser

(2) Furthur - An individual volunteering for full-time national service shall agree to perform community service for at least one year but not more than two under this Act shall egree to perform community service for at least 3 years but not more than 6 years, at the discretion of such individual.

solvise lanoitan smitting for part-time national service

administering a program, including mentoring, supervision, and other func-

(5) in the case of special senior service members, service to assist a State in (5) in the case of special senior service membrane and they func-(4) public selety service, including placement with police, fire departments,

provide recreational opportunities, and to encourage community betterment or (3) environmental service, such as service in programs to conserve, recycle, maintein, and restore matural resources in urban and rural environments, to

S) enrices in the projection community health centers, (A) human services into mean pervoce clinics, community health centers, (A) services in frogramizations, facilities serving individuals, programs serv-ing individuals with handicapping conditions, and child-care centers, and (B) service in programs to assist elderly, disabled, poor, and homeless in-(B) service in programs to assist elderly, disabled, poor, and homeless in-to the individuals with fractions, build, restore, and maintain housing for (B) service in programs to assist (B) service in programs to assist (B) errice and individuals are programs to conserve, recycle, poor of noneless individuals and self-help programs.

to ; we example a month and 2 weeks during the year; or

(A) is age 17 or over; (B) is a citizen of the United States or lawfully admitted for permanent

-Isubivibri dous

cial sentior service participant under partegraph (1), that has been prorated ac-cording to the number of hours such part-time participant serves in the pro-cording to the number of hours and hours considers necessary and appro-

(2) PARTTINE — Each part-time special senior service participant shall receive a living allowance equal to a share of such allowance offered to a full-time spe-

essary and appropriate for a special senior service participant to carry out the

service obligation of such participant.

C with reacted mean reactions. (c) Structure Bartons Deartors Participant and the receive a living ellowance equal to the iving ellowance period participant and pante under subsection (a) and such other assistance as the Board considers nec-pende and entroportiate for a special service service participant to carry out the pender and entroportiate for a special service service participant to carry out the pender and entroportiate for a special service service participant to pender and entroportiate for a special service service participant to pender and entroportiate for a special service service participant to pender and entroportiate for a special service service participant to pender and entroportiate for a special service service participant to pender and service participant to carry out the pender service participant to carry out the pender service service participant to pender and service participant t

sometiment dilased dirw be

MONVMOTTY INTAIN THE THE

SECT MIL AOOCHEERS

·aouap

member if such individual

ဂ္ပ်

(b) HEALTH INSURANCE.—In addition to the living allowance provided under sub-section (a), each participant in a full-time national service provide.

(2) Construction of a sectorization of this restriction of a first house; or (3) downpayment or closing costs for a first house; or (3) futions are minutive of the full-time participation in an apprendicating program approved by the appropriate State agency. (2) Full-THAL.—A voucher issued pursuant to aubsection (b) shall only be used program approved by the appropriate State agency.

(A) payment of a student loan from Federal or non-Federal sources; (b) Use volume and volume to subsection (s) and (i) and (i) are used

(c) SPECIAL SERVICE SERVICE PARTICIPART. A special senior service participant

(b) FULLTIME -The Board shall annually provide to each full-time participant a non-transferrable voucher that is equal in value to \$5,600 for each year of service

(a) PARTTIME The Board shall annually provide to each part-time participant a non-transferrable voucher that is equal in value to \$3,000 for each year of service

is age 60 or over; and
 meets the eligibility criteria for special senior service membership estab-(2) meets the Board.

(2) has received a high school diploms or the equivalent of such diploms, or agrees to achieve a high school diploms or the equivalent of such diploms while participating in the program; and (3) is a critizen of the United States or lawfully admitted for permanent residence.

(c) Spannak Sawios Sawios An individual may serve as a pecial service

22

shall be ineligible to receive a voucher under this section.

that such participant provides to the program.

(a) its Gawaka, -Bach participant in a full-time national service program shall receive a living allowance of not less than 100 percent of the poverty line for a single individual less defined in section 5(43) of the Community Services Block participant are following anount such the minimum wage for a 40-hour workweek under section 6(6)) of the Fair Labor the minimum wage of a 40-hour workweek under section 6(6)) of the Pair Labor the minimum wage of a 40-hour workweek under section 6(6)) of the Pair Labor the minimum wage for a 40-hour workweek under section 6(6)) of the Pair Labor the minimum wage of a 40-hour workweek under section 6(6)) of the Pair Labor the minimum wage of a 40-hour workweek under section 6(6)) of the Pair Labor the minimum wage of a 40-hour workweek under section 6(6)) of the Pair Labor the minimum wage of a 40-hour workweek under section 6(6)) of the Pair Labor the minimum wage of a 40-hour workweek under section 6(6)) of the Pair Labor the minimum wage of a 40-hour workweek under section 6(6)) of the Pair Labor the minimum wage of a 40-hour workweek under section 6(6)) of the Pair Labor the minimum wage of a 40-hour workweek under section 6(6)) of the Pair Labor the minimum wage of a 40-hour workweek under section 6(6) of the Pair Labor the minimum wage of a 40-hour workweek under section 6(6) of the Pair Labor the minimum and the form and the form of a 40-hour workweek under section 6(6) of the Pair Labor the minimum and the form and the form of a 40-hour workweek minimum and the form and the form of a 40-hour workweek minimum and the form of a 40-hour workweek minimum and the form of a 40-hour section 6(6) of the form of a 40-hour workweek minimum and the form of a 40-hour section 6(6) of the form of a 40-hour section 6(6)

(a) Tararyarary or EDUCATION AND HOUSING VOUCHER.—For purposes of determin-ing eligibility for programs under rike 12 voi the Higher Education Act of 1966 (20 U S.C. 1070 et eq.) (hereafter in this section referred to as the "Act"), vouchers re-ceived in section 428(s/X)(s) it like IV of the Act for S.C. 1078(s/X/S)), except fined in section 428(s/X/S/N) et another as a section and the Act in a considered as a section the Act in a considered as a set in a section and the act is a section a section and the Act is a considered as a section and a section and a section and a section a section and a section and a section and a section a section a section a section and a section a section and a section a

U.S.C., 1961 YAR).
 (b) The Arther or Structure or Article Exercises of expected family contri-ceived under this Act be considered in the determination of expected family contri-bution or independent teudent statemt teature under "The Act (30 U.S.C. 1070a et eq.);
 (c) Darry Remonstructure "The Act (20 U.S.C. 1070a, et eq.);
 (c) Darry Remonstructure The Act (20 U.S.C. 1070a, et eq.);
 (c) Darry Remonstructure "The Act (30 U.S.C. 1070a, et eq.);
 (c) Darry Remonstructure The Act (30 U.S.C. 1070a, et eq.);
 (c) Darry Remonstructure "The Act is emended".
 (c) Darry Remonstructure "The Act is emended".
 (c) Darry Remonstructure "The Act is emended".

(2) both soft in the section of the part is of the both as emooni lated (2) that in no case shall such a vouchet be considered as-(1) such a vouchet be norme as defined in section 411F(1) of subpart 1 of part A of title IV of the Act (20 U.S.L. 1076-6); or (2) such a voichet defined in a vouchet be voichet by or (2) such a voichet be voi

STEERINGNEN MARDORY NOTATIZNOMER SUPPORTAN ARENDMENTS.

amoldib four to fastavinpe

Each State statt receives funds under section 314 shall provide to each participant enrolled a full-time program funded under section 314 in-service educational serv-tos and materials to enable such participant to obtain a fugh school diploma or the environment of auch distance.

SEC. HT IN SERVICE EDUCATION BENELLES

U.S.C. 1087 VILLI).

section 314, including cost-sharing arrangements with sponsoring organizations. The Board shall consider and develop opportunities for cooperation between public and private sufficies in the futburg and execution of a program and sufficient and an and the sufficient and an and the sufficient of the sufficie

SEC. 111. PUBLIC: PRIVATE PARTNERSHIP.

A more provided pursuant to section ally the costs of the section ally the

(a) variants intervise edition betefits provided pursuant to section 312, griev-perticipants intervise editorian benefits provided pursuant to section 312, griev-perticipants intervise editorian benefits provided pursuant to section 312, griev-perticipants intervise editoriant pursuant provided pursuant to section 312, griev-perticipants intervise editoriant pursuant provided pursuant to section 312, griev-perticipants intervise editoriant pursuant provided pursuant pursuant provided pursuant purs bebivorg miningr to bebreqre ef light abnut investigation of the structure of (a)

'90ND4 40 280 '611 '335

เมษามักระ

-Ilana (I) Adars

'ONINEVAL 'and 'D'RG JUNGIODING YOUR 10

by the Board in cooperation with the State.

(c) Auguer on Oncountarion Training and the net training described in wheedtons (n) and (b), each participant shall receive training from the sponsoring subscription in skills relevant to the work to be conducted. #1U#d

that each participant will perform. (b) Apprivity Taximing for partici-de additional training for partici-

(ii) If feesible, specialized training for the type of national service

tesk of such participent by providing-

(2) CONTRATE OF TRAINING SESSION -- Each training session described in para-

(1) IN ORNERAL -- Each perticipant shall receive 3 weeks of training conducted

prists for a special senior service participant to carry out the service obligation 82

(C) train each participant to effectively perform the sasigned program (B) build an ethic community service; and (A) orient each participant to the nature, philosophy, and purpose of the

"(F) Annual adjusted family income does not include any stipend received by a participant in the National Service Demonstration Program established under the National and Community Service Act of 1989.

(2) in section 411F(12xBXvi) (20 U.S.C. 1070a-6(12xBXvi)), by striking "(including all sources of resources other than parents)" and inserting "(including all sources of resources other than parents and stipends received as a result of participation in the National Service Demonstration Program established under the National and Community Service Act of 1989."; (3) in section 480(f) (20 U.S.C. 1087vv(f), by-

(A) striking "and" at the end of paragraph (1); (B) by striking the period at the end of paragraph (2) and inserting a (D) by setting the period at the setting and (C) adding at the end thereof the following new paragraph:

"(3) any stipend received by a participant in the National Service Demonstration Program established under the National and Community Service Act of 1989."; and

(4) in section 480(d)(2)(F) (20 U.S.C. 1087vv(d)(2)(F)), by inserting after "other than parents" "and stipends received as a result of participation in the National Service Demonstration Program established under the National and Community Service Act of 1989)"

SEC. 314. AUTHORIZATION OF APPROPRIATIONS.

There are authorized to be appropriated to carry out this title \$100,000,000 in fiscal year 1991, \$125,000,000 in fiscal year 1992, \$150,000,000 in fiscal year 1993, \$300,000,000 in fiscal year 1994, and \$300,000,000 in fiscal year 1995,

TITLE IV—CORPORATION FOR NATIONAL SERVICE

SEC. 401. DEFINITIONS.

As used in this title:

(1) BOARD .- The term "Board" means the Board of Directors for the Corporation for National Service.

(2) Corporation.—The term "Corporation" means the Corporation for Nation-al Service, as established by section 402(a).

(3) ELECTION .- The term "election" has the same meaning, when referring to (3) LLECTION.—Intertim "election has the same meaning, when referring to an election for Federal office, as given such term in section 301(1) of the Federal Election Campaign Act of 1971 (2 U.S.C. 431(1)).
 (4) FEDERAL OFFICE.—The term "Federal office" has the same meaning as given that term in section 301(3) of the Federal Election Campaign Act of 1971 (2 U.S.C. 431(2)).

(2 U.S.C. 431(3)).

SEC. 401. ESTABLISHMENT OF CORPORATION; APPLICATION OF DISTRICT OF COLUMBIA NON-PROFIT CORPORATION ACT.

(a) ESTABLISHMENT.—There is established a nonprofit corporation, to be known as the "Corporation for National Service", that shall not be considered an agency or establishment of the United States Government.

(b) APPLICATION OF DISTRICT OF COLUMBIA NONPROFIT CORPORATION ACT .--- The Corporation shall be subject to this Act, and to the extent consistent with this Act, to the District of Columbia Nonprofit Corporation Act.

SEC. 401. BOARD OF DIRECTORS.

(a) APPOINTMENT .---

(1) IN GENERAL -The Corporation shall be directed by a National Service Board consisting of 11 members appointed by the President, by and with the advice and consent of the Senate.

(2) TIME PERIOD FOR APPOINTMENTS.-The President shall appoint members of the Board not later than 90 days after the date of enactment of this title.

(3) POLITICAL AFFILIATION .- Not more than six members of the Board shall be members of the same political party.

(4) NOMINATIONS.-Three of the members of the Board shall be appointed from individuals nominated by the Speaker of the House of Representatives, and three of such members shall be appointed from individuals nominated by the majority leader of the Senate.

(b) QUALIFICATIONS. - The President shall select the members of the Board-

(1) from among citizens who are eminent in such fields as community service, youth service, education, civic affairs, business, labor, or military service; and (2) so as to provide as nearly as practicable a broad representation of various regions of the United States, various professions and occupations, and a variety of talent and experience appropriate for the performance of the functions and

responsibilities of the Corporation. (c) INCORPORATION .- The members of the initial Board of Directors shall serve as

incorporators and shall take whatever actions are necessary to incorporate the Corporation under the District of Columbia Nonprofit Corporation Act.

(d) TERM OF OFFICE .- The term of office of each member of the Board shall be 7 years, except that-

(1) any member appointed to fill a vacancy within the Board occurring prior to the expiration of the term for which the predecessor of such member was appointed shall be appointed for the remainder of such term:

(2) initial appointments to the Board shall be for terms of 3, 5, or 7 years; and (3) no member of the Board shall be eligible to serve more than two consecutive terms.

(e) VACANCY .-- A vacancy in the Board shall not affect the power of the Board and shall be filled in the same manner as the original appointment (f) MEETINGS.

(1) REQUIREMENT.-- A member of the Board shall attend not less than 50 percent of all duly convened meetings of the Board in any calendar year. (2) PENALTY ---

(A) IN GENERAL-A member who fails to meet the requirement of paragraph (1) shall forfeit membership on the Board. (B) APPOINTMENT OF NEW MEMBER.—The President shall appoint a new

member to fill such vacancy created under subparagraph (A) (while meeting the requirements of subsection (e)), not later than 30 days after such vacancy is determined by the Chairperson of the Board, as elected in subsection (g)

(3) QUORUM.—Six members of the Board shall constitute a quorum. (3) QUORUM.—Six members of the Board shall constitute a quorum. (g) ELECTION OF CHAIRPERSON AND VICE CHAIRPERSON.—Members of the Board shall annually elect one such member to be Chairperson and elect one or more of such members as a Vice Chairperson.

(h) COMPENSATION OF BOARD MEMBERS .---

(1) NON-FEDERAL EMPLOYEE.—A member of the Board shall not, by reason of such membership, be considered to be an officer or employee of the United States

(2) COMPENSATION .- Except as provided in paragraphs (3) and (4), a member of the Board shall, while away from the home or regular place of business of such member, be allowed travel and actual, reasonable, and necessary expenses.

(3) FEDERAL EMPLOYEE .- A member of the Board who is a full-time officer or employee of the United States shall receive no additional pay, allowances, or benefits by reason of such membership,

SEC. 404. OFFICERS AND EMPLOYEES.

(a) IN GENERAL ---

(1) RATE OF BASIC PAY .- The Corporation shall have a President, and such other officers and employees as may be named and appointed by the Board for terms and at rates of compensation fixed by the Board, except that no officer or employee of the Corporation may receive compensation at an annual rate of pay

employee of the Corporation may receive compensation at an annual rate of pay that exceeds the rate of basic pay payable from time to time for level I of the Executive Schedule under section 5312 of title 5, United States Code. (2) ADDITIONAL COMPENSATION.—No officer or employee of the Corporation shall receive any salary or other compensation from any source other than the Corporation for services performed for the Corporation. (2) The service and the service of the Compension shall be compensioned and the service of the Compension shall be compensioned and the service of the compension shall be compensioned and the service of the compension shall be serviced and the service of the serv

(3) TERM OF OFFICE.-All officers and employees of the Corporation shall serve at the pleasure of the Board.

(b) EMPLOYMENT OF SPECIAL SENIOR SERVICE MEMBERS .- In selecting employees. the Board is encouraged to include members of the Special Senior Service as defined in section 302(8)

(c) DETAIL OF GOVERNMENT EMPLOYEES .- Any Federal Government employee may be detailed to the Corporation without payment of reimbursement to the detailing agency. A detail of a Federal employee under this subsection shall not result in the interruption or loss of civil service status or privilege of such employee.

SEC. 495. NONPROFIT AND NONPOLITICAL NATURE OF THE CORPORATION.

(a) LIMITATIONS ON POWERS .---

25

(I) Issuance or stock -The Corporation shall not issue any shares of stock or

(2) hoose on Assers or the convokation. - No part of the income or easels of (2) hoose on Assers or the benefit of any director, officer, employee, or declare or pay any dividends.

any other individual except as salary or reasonable compensation for services

to or otherwise support any political party or candidate for elective public contribute

(a) Housing AND EDUCATION VOUCHERS. - The Corporation shall leave housing and education vouchers pursuant to section 307. The Board shall consult with the SEC. IM. HOUSING AND EDUCATION VOUCHERS, LIVING ALLOWANCES.

Department of Education in jesting education vouchers. (b) Laving AlLowaviczs. The Corporation shall establish living allowances pursu-

ant to section 308, taking into account regional variations in the cost of living.

8280-J88 '40 '036

ō

(a) $\Delta TaxTR Reports - Each State receiving minde under part A of title I and titles (1) we are a solution of the Board on the status of national terms in the Board on the status of the Horn II and State (1) and the state of the the state o$

tainment of program goals. that of that are the state of the state of the state of an are state of national and community youth service programs in such State. (2) Locks GRANTES.—Each State may require local grantees receiving funds made part A of this is and itslead it and III to supply such information as is nec-meany to complete such report, including a comparison of actual seconplish-ments with the goals established for the program, the number of participants in meeting to complete such report, including a comparison of actual seconplish-meeting to complete such report, including a comparison of actual seconplish-problems, the number of service hours generated, and the axistence of any problems, and delays or adverse conditions that have affected or will affect the sti-tist meeting of program, the grade service hours generated, and the axistence of any interaction action and the service hours generated, and the axistence of any the program, the grade service hours generated, and the axistence of any the program, the service program is the service of a state and any service service hours generated, and the state server ac-tion of actions that the service of a state server action action and the service of actions that the service of any service of any the service service service hours generated, and the state server the service service service service hours generated and the service service service service hours generated and the service ser

(3) REPORT DEMONSTRATING COMPLEXING

(i) Avaltabilistic accessery to comply with persgraph (i) and (i) Avaltabilistic or accessery to comply with persgraph (i) to a supply of the public on the provident of REPORT-Such report shall be available to the public on the provident of the provident of the public of the provident of the provident of the public of the provident of the provident of the public of the provident of the provident of the public of the provident o (A) IN ORNEAL--Each State receiving funds pursuant to this title shall include information in such report that demonstrates compliance with the provisions of section 412.

request. (b) Haroar TO CONGRERS. (c) Haroar TO CONGRERS. (c) Harakve.-The Board aball, not later than 120 days after the end of the appropriate autorising and appro-priation committees in Congress a report on programs funded under part A of these I and thiss II and III. (c) CONTRAT.-Such report shall summarize information contained in State nea (c) CONTRAT.-Such report shall summarize information contained in State nea (c) CONTRAT.-Such report shall summarize information contained in State nea (c) CONTRAT.-Such report shall summarize information contained in State nea (c) CONTRAT.-Such report shall summarize information contained in State nea (c) CONTRAT.-Such report shall summarize information contained in State nea (c) CONTRAT.-Such report shall summarize information contained in State nea (c) CONTRAT.-Such report shall summarize information contained in State nea (c) CONTRAT.-Such report shall summarize information contained in State nea (c) CONTRAT.-Such report shall summarize information contained in State nea (c) CONTRAT.-Such report shall summarize information contained in State nea (c) CONTRAT.-Such report shall summarize information contained in State nea (c) CONTRAT.-Such report shall summarize information contained in State nea (c) CONTRAT.-Such report shall summarize information contained in State nea (c) CONTRAT.-Such report shall summarize information contained in State nea (c) Contained in Contained in State nea (c) Contained in Containe

(2) Contrart --Such report shall summarize information contained in State re-ports required under subsection (a) and reflect the findings and actions taken as a result of any evaluation conducted by the Board

SEC. 100. SUPPLIEMENTATION.

(a) In variant of the transformation of the solution of the second of th

We have the second state of other and other provides the second of services assets, educet this Act in the previous fixed year. (b) Accorstorate stream of the segregate expenditure in such program for the index year peritouist program in the previous fixed year, excluding Federal funds and funds used in such program in the previous fixed year, excluding Federal funds and funds used by the remainder of the costs of programs assisted under this title.

BONDA 40 380 NO NOLLIGIHORA 169 (338)

(a) IN ORNEAL—Funds provided under part A of titls I and titles II and III shall not be used by program participants and program staff to— (b) give religious instruction, conduct workip services, or engage in any form (1) give religious instruction.

(2) assist, promote, or deter union organizing, and of proselytization;

(3) finance, directly or indirectly, an activity designed to influence the out-come of an election to Federal office or the outcome of an election to a State or

eaffic attauc isso

'\$1UəUləə198 (b) Contracts on collective BARGAINING AGREMENTS — A project assisted under (b) Contracts on collective BARGAINING contracts for services or collective bargaining

60 days after the filing of such grievance if no decision has been reached, (A) IN GENERAL -On the occurrence of an adverse grievance decision, or

(B) Decration, —A decision on any grievance shall be made not later than 60 days after the filing of such grievance.

days of filing such grievance.

06 nithiw betauloo ed llada sonaveira yna no guined A--.oniavati (A) (3) DRADLAR FOR HEARING AND DECISION .----

- TVAINED NI (B)

(c) KULAS AND RECULATIONS (c)

NOLLYNININOBIONON WIF 'DIE

SEC. 417. HOLICE, REARING, AND CRIEVANCE PROCEDURES.

องนอมทาววง podojja

(b) Haranvez, $H_{\rm Restrings}$ or other meetings that may be necessary to fulfill the requirements of this section shall be held at locations convenient to such recipient. (c) Taxascust of this section shall be held at locations convenient to such recipient. (c) Taxascust of Reconnuc.—A transcript or recording shall be made of a hear-ing conducted under this section and shall be available for inspection by any indi-vidual. Insterial failure to comply with this Act or the applicable terms and conditions of any such grant or contract issued pursuant to this Act. (2) Processues to sustens sustance...The Secretary of Education in the program funded under part A of still by or the Board (in the case of a with applicable terms and conditions of this Act except, in emergency (B) assistance under this Act shall not be turminated for failure to comply with applicable terms and conditions of this Act except, in emergency with applicable terms and conditions of this Act except, in emergency with applicable terms and conditions of this Act except, in emergency with applicable terms and conditions of this Act except, in emergency with familiance inder this Act shall not be terminated for failure to comply with abplicable terms and conditions of this Act except, in emergency with applicable terms and conditions of the terminated is the applicable terms and conditions of the terminated with the applicable terms and conditions of the act and with the applicable terms and conditions of the Act with applicable terms and conditions of the act with a set and the act and and the act and the act act of auch with the applicable terms and conditions of the act with the act and the act and with the act and the act and and and and and with the act and and and and and and and and with the act and a the act and with the act and a the act and with the act and a the act and and with the act and a the act and a the act and with the act and a the act and a the act and with the act and a the act and with t Sur hearing.

viding assistance under this Act whenever the Secretary determines there is a material failure to comply with this Act or the applicable terms and conditions

to suspend payments or to terminate payments under a contract or grant prounder title II or III), is sufforized, in accordance with the provisions of this Act.

(1) Superstant of the Bocretary of Education (in the case of a function (in the case of a program function of the Board (in the case of a program function of the state of the

pension of assistance for not more than 30 days, on an emergency basis, until notice and an opportunity to be heard can be provided

3) MULES AND INCOLATIONS.— (1) BOUCATION in the case of programs funded under part A of title I, the Gorgenetic of this section that hall issue rules and regulations to provide for the en-formance of this section that hall include provisions for summary suspension of assistance for no more than 30 days, on an ensergency basis, until notice and (2) VOLUNTEEE structure.—In the case of programs funded under title II and (2) VOLUNTEEE structure.—In the case of programs funded under title II and (2) VOLUNTEEE structure.—In the case of programs funded under title II and (2) VOLUNTEEE structure.—In the case of programs for summary sus-title III of this Action for that shell include provisions for summary sus-mary and regulations of this section that than 1 molecular provide for the section that shell include provision for under title II of this Action for the section the then 1 molecular for under title II and regulations for this section the the section the provide for the section that and the section the the section the provide include beneficial and the section that and the section the provide include the antipoteneous of this section that the section the provide include the section of the section that and the section the section the provide include the section of the section that the section that section the secti

(b) Noverscanarior An individual with responsibility for the operation of a program funded under this part A of title 1 and titles II or III shall not discriminate in the selection of participants to such program because of race, religion, color, national origin, see, age, handicap, or political affiliation.

(a) IN GENERAL—Any finencial sesistance provided under this Act shall consti-tute Federal financial assistance for purposes of ities VI of the Civil Rights Act of 1964 (42 U.S.C. 2000d et sec), itils IX of the Education Amendments of 1978 (20 1965 (32 U.S.C. 2000d et sec), itils IX of the Education Amendments of 1978 (20 Discrimination Act of 1975 (42 U.S.C. 6101 et sec), and the regulations issued there and the regulation act of 1975 (42 U.S.C. 6101 et sec), and the regulations issued there and the regulation act of 1975 (42 U.S.C. 6101 et sec), and the regulations issued there are actively act of 1975 (42 U.S.C. 6101 et sec). and the regulations issued the regulation act of 1974 active discrimination act of 1975 (42 U.S.C. 6101 et sec). and the regulations issued incre-tional active the sec) is the regulation act of 1973 (20 U.S.C. 701 et sec). Increase active the sec) is the regulation act of 1975 (42 U.S.C. 6101 et sec).

ĹΖ

the party filing the grievance shall be permitted to aubmit such grievance to binding arbitration before a qualified arbitrator who is jointly selected and the party filing the grievance parties. (B) Decaute of the interested parties. An arbitration proceeding shall be held (B) Decautes not successing arbitration around the file date of such arbitration proceeding shall be made within 30 days after the date of such arbitration proceeding shall be made within 30 days after the date of such arbitration proceeding shall be date of such arbitration proceeding. (D) Costruction proceeding that the cost of such arbitration proceeding that the date of such arbitration proceeding.

(1) the effectiveness of such programs in achieving stated goals in general and (a) IN CRNRAR.—The Board shall provide, through grants or contracts, for the continuing evaluation of programs assisted under titles II and III, including evaluation of programs assisted under titles II and the indeaure and evaluate the impact of programs authorized by titles II and the indeaure and evaluate the impact of programs authorized by titles II and the indeaure and evaluate the impact of several states and the impact of programs are several states and the including evaluate the several states and the indeaure and evaluate the impact of several states are included by titles II and the indeaure and evaluate the impact of several states are and the included by titles II and the indeaure and evaluate the several states are assisted and the indeaure and the indeaure and the indeaure and evaluate the impact of several states are assisted and the indeaure and a several se

(4) developing programs, training methods, curriculum materials, and other materials and activities related to programs funded under this Act.

(1) coordinating service programs and related programs within the State;
 (2) disseminating information about service programs funded under this Act;
 (3) recruiting participants for programs funded under this Act; and

(c) DUTES OF BOARD.—The State Advisory Board for National and Community Service shall assist the State sgency administering a program funded under part A of title I or titles II or III in...

in relation to cost:

(2) the impact of such programs on related programs, including VISTA, the military, the Peace Corps (as setablished by the Peace Corps Act (22 U.S.C. 260)

et seq.)), and the Mational Guard; and

(b) Comparisons.—The Board shall include in such evaluations, where appropri-(b) Comparisons.—The Board shall include in such evaluations, where appropri-tion of the succession of the suc

ate, comparisons of participants in such programs with individuals who have not

ats, competisons of participate in such programs with individuals who have not participated in auch programs. (c) Competisons of participated in the administration of such programs by individuals who are not directly involved in the administration of such programs. (d) Program Oskernves.—The Board shall ensure that programs funded under title III are evaluated for effective adverte participants in auch programs, consist-(f) forculting and enrolling of diverse garticipants in auch programs, consist-ent with the requirements of section 306, based on economic background, race, (2) promoting of the educational achievement of and handlergy, ent with the requirements of section 306, based on economic background, race, (2) promoting of the educational achievement of and handlergy, ent with the requirements of section 306, based on economic background, race, ent with the requirements of section 306, pased on economic background, race, (3) promoting of the educational achievement of and handlergy, ent with the requirements and completion of increasingly higher levels of grams, based on earning a high action of increasingly higher levels of educations.

uomeonoe

Given the service service to engage in public and community service (3) encouraging each participant to engage in public and community service strest completion of the program based on career choices and service program estab-lished under the Domestic Volunteer Bervice Act of 1973 (42 U.S.C. 4950 et seq.), the Peace Corps (as established by the Peace Corps Act (S2 U.S.C. 2501 et seq.), the militant and marking polynetizer service.

(4) promoting of positive attitudes among each participant regarding the role the military, and part-time volunteer service;

(3) providing a metricutes animaly scaling main plantingpart in expansions of providents in solving community problems based on the view of such participant regarding the provident regarding the previous of such participant in solving community provident as a cluster and community by member, and other factors: (b) mapping such participant to finance a lesser portion of the higher educa-tion of such participant to finance a lesser portion of the higher educa-(b) mapping such participant to finance a lesser portion of the higher educa-tion of such participant to finance a lesser portion of the higher educa-tion of such participant to finance a lesser portion of the higher educa-tion of such participant to finance a lesser portion of the higher educa-tion of such participant to finance a lesser portion of the higher educa-tion of such participant to finance a lesser portion of the higher educa-tion of such participant to finance a lesser portion of the higher educa-tion of such participant such a solver assistance to community agencies, but not overleading such agencies and activities that beams;

DUR SIA (8) providing services and activities that could not otherwise be performed by employed workers and that will not supplant the hiring of, or result in the dis-placement of, employed workers or impair the existing contracts of such work-placement of.

(9) attracting a greater number of citizens to public service, including service in the active and reserve components of the Armed Forces, the National Guard, the Place Corps (as setablished by the Peace Corps Act (22 U.S.C. 250) et sec)), and active and active and active the Peace Corps (22 U.S.C. 250) et sec)).

grams offering alternative voucher options, and programs utilizing individual place-(a) CONFARSON ON PROGRAM MODELE, —The BOARD shall evaluate and compare the effectiveness of different program models in meeting the programs objointing differ-aubsection (d) including full-time and part-time programs, programs involving differ-ent types of national service, programs using different recruitment methods, pro-ent types of national service, programs using different recruitment methods, pro-ter types of national service, programs using different recruitment methods, pro-ent types of national service, programs using different recruitment methods, pro-ter types of national service, programs using different recruitment methods, pro-ter types of national service, programs using different recruitment methods, pro-ter types of national service, programs using different recruitment methods, pro-ter types of national service, programs using different recruitment methods, pro-ter types of national service, programs using different recruitment methods, pro-ter types of national service, programs using different recruitment methods, pro-ter types of national service, programs using different recruitment methods, pro-ter types of national service, programs using different recruitment methods, pro-ter types of national service, programs using different recruitment methods, pro-ter types of national service, programs using different recruitment methods, pro-ter types of national service, programs using different recruitment methods, pro-ter types of national service, programs and programs and the service of national service and the service of national service and the service of national service and the service and th ATSIV bns

smeat bns sinam

NOLLYNYNAE 'N 'DAS

(1) Obstituted Interaction. \sim (1) In constant, \sim In performing the evaluation required under subsection (d), (f) In constant, \sim In performing the evaluation and State or local applicant to the Board may require each program participant and State or local applicant to

82

100. (b) Архонитакит от Макакик -- The Governor of such State shall appoint mem-bers to such Advisory Board from annong-(1) representatives of State agroncies administering community service, youth (1) representatives of late agroncies, and dob training programs; and (2) representatives of late agroncies, agencies atudents, teachers, (2) representations auch as community service, agroncies, students, teachers, (2) representations and burder frograms as established under title 11 of the Domes (2) representations and the groncies agencies, students, teachers, (3) representations and burder frograms, agencies and out aerrorice community (4) representations and the groncies agencies and the Domes (4) representation and the service programs, higher education institu-tion of the American Partice community service programs, higher education institu-tion. (a) Poswarrow or Boasn.—Each State applying for funds under part A of title 1 or (internation of title 1 or the state Advisory Board for National and Community Serv-

SEC. 417 STATE ADVISORY BOARD.

auch employee.

established under this Act.

Jaubivibai bevolqma

SEC. HT NONDERFICUTION AND NONDISFLACEMENT.

between the parties.

(v) employee who is on strike or who is being locked out.

assigned to any-(i)) employee who is subject to a reduction in force; (ii)) employee who is subject to a reduction in force; (iii) employee who is on leave (terminal, temporary, vacation, emer-for sick), or servey, or sick), or (v) employee who is on strike or who is being locked out. (v)

(C) DUTIES FORMERLY PERFORMED BY ANOTHER EMPLOYER.—A participant (C) not perform services or duties that have been performed by or were (B) Superiorna or High C.— A participant in any program under this of shall not perform any services or duties or engage in activities that will support the hiring of employed workers.

(A) DURLIGATION OF SERVICES.—A participant in a program under this Act ablail not perform any services or duries or engage in activities that would otherwise be performed by an employee as part of the assigned duties of

the Act that will infringe in any manner on the promotional opportunity of an (2) BERVICE OFFORTURITIES -- A service opportunity shall not be created under

(1) in utwartent. An employer shall not displace an employee or position, in-cluding partial displacement such as reduction in hours, wages, or employment benefits as a result of the use by such employer of a participant in a program activitation under this for

provided to a private nonprofit entity to conduct activities that are the same or substantially equivalent to activities provided by a State or local government agency that such entity resides in, unless the requirements of subsection (b) are able in the locality of such program. (2) PRIVATE NONPROFIT EWITY--Funds available under this Act shall not be (2) PRIVATE NONPROFIT EWITY--Funds available under the are the same or

(a) Nonupricerrow.— (b) Nonupricerrow.— (c) Nonupricers, and is in addition to, an activity otherwise avail-gram that does not duplicate, and is in addition to, an activity otherwise avail-ble in the locality of such program.

Detween the partnes. Detween the partnes. (5) Froncess the partnes. The partness of the partness are a grievance is filed regarding a proposed place-ment of a participant in a program assisted under this subsection include-(6) Ramfours - Remedies for a grievance filed under this subsection include-(6) Ramfours - Remedies for a grievance filed under this subsection include-(8) Berninskion of such payments and (9) Berninskion of such payments and (9) Berninskion of such payments and (10) Prohibition of such pareness and (10) Prohibition of such pareness and (10) Prohibition of such placement described in paragraph (5).

programs, school-based community service programs, higher education institu-tions, local educationel agencies, volunteer public safety organizations, educa-tional partnetship programs, and other organizations working with volunteers.

provide such information as may be necessary to carry out the requirements of this section.

(2) CONFIDENTIALITY .- The Board shall keep such information confidential (g) DRADLINE-The Board shall complete the evaluation required under subsec-

tion (d) not later than 4 years after the date of enactment of this Act.

SEC 416 SUNDANC

Of funds appropriated for titles II and III of this Act, not less than \$5 million or more than \$25 million shall be made available to the Board for program support and activities referred to in sections 414 and 416.

SEC. (14. FUNCTIONS OF THE NATIONAL SERVICE BOARD.

The National Service Board shall-

(1) administer programs authorized under title II and title III of this Act: (2) provide, directly or through contract with public or private nonprofit organizations with extensive experience in service programs, training and technical assistance to States, full-time youth service corps, and national service demonstration programs;

(3) in consultation with the Department of Education, provide one or more clearinghouses for information on service (the Board may contract with public or private non-profit organizations with extensive experience in service to perform such clearinghouse function); (4) consult with appropriate Federal agencies in administering programs

funded under titles II and III: and

(5) arrange for the evaluation of programs authorized under titles II and III of this Act, in accordance with section 414.

SEC. 417. PRESIDENTIAL AWARDS FOR SERVICE.

(a) PRESIDENTIAL AWARDS .----

(1) IN GENERAL --- The President is authorized to make Presidential Awards for service to individuals demonstrating outstanding community service and to outstanding service programs.

(2) NUMBER OF AWARDS - The President is authorized to make one individual and one program award in each Congressional district, and one statewide program award in each State.

(3) CONSULTATION .- The President shall consult with the Governor of each State in the selection of individuals and programs for Presidential Awards.

(4) PARTICIPANTS IN PROGRAMS .- An individual receiving an award under this

section need not be a participant in a program assisted under this Act. (b) INFORMATION.—The President shall ensure that information concerning individuals and programs receiving awards under this section is widely disseminated.

SEC. 418. COMPREHENSIVE BERVICE STRATEGY.

The President shall design a comprehensive Federal service strategy that shall include-

(1) the review of existing programs to identify and expand opportunities for service, especially by students and out-of-school youth;

(2) the designation of a senior official in each Federal agency who will be responsible for developing youth service opportunities in existing programs nationwide:

(3) the establishment of service projects in each Federal agency;

(4) the encouragement of Federal employees to participate in service projects; (5) the designation of an executive branch official to coordinate the Federal service strategy; and

(6) the annual recognition of outstanding service programs by a Federal agency.

SEC. 410, ENGAGEMENT OF PARTICIPANTS.

A State shall not engage a participant to serve in any program funded under this Act unless and until Federal funds have been appropriated for vouchers and other necessary expenses and costs associated with such participant in such fiscal year in such State

TITLE V-EXPANSION OF VOLUNTEERS IN

SERVICE TO AMERICA

SEC. MI. SHORT TITLE.

This title may be cited as the "VISTA Expansion Act of 1989".

BEC. ML AUTHORIZATION OF APPROPRIATIONS.

(a) NATIONAL VOLUNTEER ANTIPOVERTY PROGRAMS .- Section 501 of the Domestic Volunteer Service Act of 1973 (42 U.S.C. 5081) is amended by striking out paragraph (1) of subsection (a) and inserting in lieu thereof the following new paragraph

"(1) There are authorized to be appropriated to carry out part A of title I (except section 109) \$30,600,000 for fiscal year 1990, \$39,909,000 for fiscal year 1991, \$47,800,900 for fiscal year 1992, and \$56,500,000 for fiscal year 1993.".

TITLE VI-NATIONAL OLDER AMERICANS VOLUNTEER PROGRAMS

SEC. 101. SHORT TITLE.

This title may be cited as the "National Older American Volunteer Programs Expansion Act of 1989"

SEC. 491. PROCEAUS OF NATIONAL AND LOCAL SIGNIFICANCE.

Part D of title II of the Domestic Volunteer Service Act of 1973 (42 U.S.C. 5021 et seq.) is amended by adding at the end thereof the following new section:

"SEC. 235. PROGRAMS OF NATIONAL AND LOCAL SIGNIFICANCE.

"(a) IN GENERAL .- The Director shall establish, within each program authorized under this title, a program for making grants to support programs that address national problems on a local level.

(b) Use or GRANTS.-The recipient of a grant under the program established under subsection (a) shall use such grant to provide creative solutions to urgent problems.

"(c) AWARDING OF GRANTS.-

"(1) ESTABLISHMENT OF PROGRAM .- There is established the 'Programs of National and Local Significance' program. Under the program, the Director shall award grants each year to programs administered under this title to respond to an identified community need.

"(2) AWARDO.---

"(A) IN OXNERAL -The grants authorized under paragraph (1) may be awarded to both existing and new projects. "(B) LIMITATION.—A grant under paragraph (1) may not exceed \$150,000

per year. "(3) CRITERIA POR AWARDING GRANTS.---

"(A) IN ORNERAL.—Under the program established under paragraph (1), the Director shall award grants based on a demonstration by an applicant that such grant will enable such applicant to uniquely and effectively respond to an identified community need.

"(d) Use or GRANTE.-- A program receiving a grant under subsection (a) shall dem-onstrate that assistance provided by such grants shall be used to increase--

"(1) the total number of volunteers supported by such projects; and

"(2) the number of volunteers in such projects engaged in responding to the identified community need referred to in subsection (g) for which such grant waa made.

"(e) DISSEMINATION OF INFORMATION .- "The Director shall disseminate information on the Programs of National and Local Significance established under this section to field personnel of the ACTION Agency and other community volunteer organizations that request such information.

"(f) PRIORTY.—Priority for grants under this section shall be given to the follow-ing programs of national significance— "(1) programs that assist individuals with chronic and debilitating illnesses

such as immune deficiency syndrome; "(2) programs designed to decrease drug and alcohol abuse; "(3) programs that work with teenage parents;

SEC. 633. AUTHORIZATION OF APPROPRIATIONS.